Studies on regulation of versican gene expression by hypoxia in primary human <u>macrophages</u>

Thesis submitted for the degree of Doctor of Philosophy

by

Fattah Sotoodehnejadnematalahi

Supervised by Dr. Bernard Burke

Department of Infection, Immunity and Inflammation

University of Leicester, United Kingdom

July 2011

Acknowledgments

Firstly I wish to thank my supervisor Dr. Bernard Burke who provided the opportunity to do my PhD and has given great supervision, advice and wisdom over the years. His ability to see a bright side, broad knowledge, helpful suggestions, guidance and being available when needed played a major part in the course of my PhD.

My heartfelt thanks go to my parents and my sister for their constant love, supporting a large part of my education and moral encouragement throughout. Their support and faith gave me the initial encouragement and confidence to undertake a PhD. I also gratefully acknowledge my father, Mohammad Sotoodehnejad for providing the funding for this research.

I would like to thank Dr. Rojer James and Dr. Cordula Stover, members of my PhD committee meeting, for looking over this research and being available when needed. Thanks for critically discussing my data and your suggestions for my thesis.

I would also like to thank all the members of the Department of Infection, Immunity and Inflammation, who have given up their time and blood and for generosity in sharing equipments and chemicals especially the members of renal group (lab 117 and 118). Many thanks also to Lory, Abdulkareem and particularly Helen Pearson for great experimental advice. I couldn't have done it without all of you! I extend equally important thanks to Elvina Chrysanthou for her friendship and especially producing valuable data during her MSc project under my supervision.

Last but not least I wish to express my gratitude to my wife, Shayesteh Mashayekhi for all of her love and great encouragement. Her faith in me and the daily conversation kept me going through the final stages. Thanks for letting me go.

Abstract

Author: Fattah Sotoodehnejadnematalahi

Title: Studies on regulation of versican gene expression by hypoxia in primary human macrophages

Hypoxia is a hallmark of many pathological tissues. Macrophages accumulate in hypoxic sites and up-regulate a number of hypoxia-inducible genes. The extracellular matrix glycoprotein versican has recently been identified as one such gene, but the mechanisms responsible for hypoxic induction are not well characterised. Here, hypoxic up-regulation of versican was investigated in primary human monocytederived macrophages. Flow cytometry of isolated peripheral blood mononuclear cells demonstrated a three-fold increase in versican protein in macrophages after 5 days incubation in hypoxia. Subset analysis showed that macrophages, and not lymphocytes, are the main peripheral blood mononuclear cells which express, and show hypoxic up-regulation of, versican protein and mRNA. This study showed that versican mRNA is up-regulated 34-fold after exposure of primary human macrophages to hypoxia for 18hrs. Further investigation showed that versican mRNA decay rates are not affected by hypoxia, indicating that hypoxic induction of versican mRNA is mediated by increased promoter activity rather than increased mRNA stability. Extensive deletion and transfection analysis of proximal versican promoter luciferase reporter constructs identified two regions which are required for high level activity of the promoter in hypoxic primary human macrophages. A recent publication suggested that hypoxic induction of versican mRNA in macrophages is mediated by the hypoxia inducible transcription factor HIF-1 α . Here, bacterial lipopolysaccharide and the hypoxia mimetic agents desferrioxamine and cobalt chloride, three stimuli which are known to induce HIF-1 α , were used to investigate the role of HIF-1 in the upregulation of versican mRNA. Neither LPS nor cobalt chloride caused up-regulation of versican mRNA, although control HIF-1 regulated genes were up-regulated, suggesting that high-level transcription of the versican promoter in hypoxia occurs via a HIF-1 independent mechanism. Lastly, two specific inhibitors of PI3-kinase, LY294002 and Wortmannin, were shown to down-regulate hypoxic induction of versican mRNA, suggesting a possible role for PI3-kinase.

Abbreviations

α- Alpha

ALDA- Aldolase A

Amp- Ampicillin

AP-1- Activator Protein-1

APCs- Antigen Presenting Cells

APS- Ammonium Persulphate

AREs- Adenylate Uridylate Rich Elements

ARNT- Aryl hydrocarbon Receptor Nuclear Translocator

ASMCs- Arterial Smooth Muscle Cells

ATF- Activating Transcription Factor

ATP- Adenosine Triphosphate

β- Beta

 β -2M- β -2 microglobulin

bFGF- basic Fibroblast Growth Factor

bp- base pair

BSA- Bovine Serum Albumin

°C- Celsius

CBP- CREB Binding Protein

CBP/P300- CREB Binding Protein p300

CD- Cluster of Differentiation

cDNA- Complementary Deoxyribonucleic acid

ChIP- Chromatin Immunoprecipitation

CNS- Central Nervous System

CO2- Carbon dioxide

COX-2- Cyclooxygenase- 2

CREB- CAMP-Response Element Binding protein

CS- Chondroitin Sulphate

CSF- Colony Stimulating Factor

CSPGs- Chondroitin Sulphate Proteoglycans

- DEPC- Diethylpyrocarbonate
- DFO- Desferrioxamine
- dH₂O- Deionised Water
- DMSO- Dimethyl Sulfoxide
- DNA- Deoxyribonucleic acid
- dNTPs- Deoxynucleotide Triphosphates
- D-PBS- Degassed Phosphate Buffered Saline
- DTT- Dithiothreitol
- E.coli- Escherichia coli
- ECM- Extracellular Matrix
- EDTA- Ethylenediaminetetraacetic acid
- eEF2- eukaryotic Elongation Factor-2
- EGF- Epidermal Growth Factor
- ELISA- Enzyme-Linked Immunosorbent Assay
- EMSA- Electrophoretic Mobility Shift Assay
- EPO- Erythropoietin
- ER- Endoplasmic Reticulum
- ERK- Extracellular Regulated Kinase
- ET-1- Endothelin-1
- EtBr- Ethidium Bromide
- FACS- Fluorescence-Activated Cell Sorting
- FCS- Feta Calf Serum
- FGF2- Fibroblast Growth Factor 2
- FIH- Factor Inhibiting HIF
- FITC- Fluorescein Isothiocyanate
- FSC- Forward Scatter
- γ- Gamma
- g- Gramme or gravitational acceleration
- G3PDH- Glyceraldehyde 3-Phosphate Dehydrogenase
- GAG- Glycosaminoglycans
- GC- Golgi complex
- GLUT-1- Glucose transporter 1

GM-CSF- Granulocyte/Macrophage-Colony Stimulating Factor

GSK- Glycogen Synthase Kinase

H₂O- Water

HBSS- Hank's Balanced Salt Solution

HCl- Hydrochloride acid

HIF-1- Hypoxia-Inducible Factor 1

HMDM- Human Monocyte-Derived Macrophages

HRE- Hypoxia Response Elements

HRP- Horse Redish peroxide

IAP-2- Inhibitors of Apoptosis Protein 2

IFN- Interferon

Ig- Immunoglobulin

IL- Interleukin

iNOS- inducible Nitric Oxide Synthase

Kb- Kilo-base

KCL- Potassium Chloride

kDa- Kilo-Dalton

LAR- Luciferase Assay Reagent

LB-Luria Bertani Medium

LDHA- Lactate Dehydrogenase A

LDL- Low Density Lipoprotein

LPS- Lipopolysaccharide

MACS- Magnetic Cell Sorting System

MAPK- Mitogen Activated Protein Kinase

MCP-1- Monocyte Chemoattractant Protein 1

M-CSF- Monocyte Colony-Stimulating Factor

MCT2- Monocarboxylate Transporter 2

MIF- Migration Inhibitory Factor

MMP-7- Matrix Metalloproteinase-7

MR- Mannose Receptors

mRNA- Messenger Ribonucleic Acid

N₂- Nitrogen

- NaCl-Sodium Chloride
- NF-κB- Nuclear Factor-Kappa Binding
- NO- Nitric Oxide
- O₂- Oxygen
- ODD- Oxygen Dependent Degradation domain
- PAI-1- Plasminogen Activator Inhibitor-1
- PAMP- Pathogen Associated Molecular Patterns
- PBMC- Peripheral Blood Mononuclear Cells
- PBS- Phosphate Buffered Saline
- PCR- Polymerase Chain Reaction
- PDGF- Platelet Derived Growth Factor
- PEI- Polyethylenimine
- PGK- Phosphoglycerate kinase
- PHD- Prolyl Hydroxylase Enzymes
- PI3K-Phosphoinositide 3-kinase
- PKB- Protein Kinase B
- PKC- Protein Kinase C
- PRPs- Pattern Recognition Receptors
- **RLU- Relative Light Units**
- RNA- Ribonucleic Acid
- RNAi- Ribonucleic Acid interference
- RNase- Ribonuclease
- rpm- Revolutions Per Minute
- RT- Room Temperature or Reverse transcriptase
- RT-PCR- Real Time Polymerase Chain Reaction
- SDS- Sodium Dodecyl Sulphate
- SDS-PAGE- Sodium Dodecyl Sulphate Poly Acrylamide Gel Electrophoresis
- siRNA- small interfering Ribonucleic Acid
- SLC- Secondary Lymphoid tissue Chemokine
- SNP- Single Nucleotide Polymorphisms
- SP-1- Specific Protein- 1
- SR- Scavenger Receptors

SSC- Side Scatter

- **TAD-** Transactivation Domains
- TAMs- Tumour Associated Macrophages
- TBE- Tris Borate EDTA buffer
- TBS- Tris Buffered Saline
- TCF/LEF- Transcription Factor Lymphoid Enhancer binding Factor
- Tess- Transcription Element Search System
- TGF- Transforming Growth Factor
- Th-T helper
- TLR- Toll Like Receptor
- **TNF-** Tumour Necrosis Factor
- ULA- Ultra Low Attachment
- UTR- Untranslated Region
- UV- Ultraviolet light
- VEGF- Vascular Endothelial Growth Factor
- VHL- Vo Hippel Lindau protein
- V-Volts
- **XRE-Xenobiotic Responsive Element**

Acknowledgmentsi
Abstractii
Abbreviationsiii
List of figuresxiv
List of tablesxviii
Chapter 1: Introduction1
1.1 Macrophages1
1.1.1 Origin of macrophages1
1.1.2 Role of macrophages in the body3
1.1.3 The inflammatory macrophages6
1.2 Hypoxia
1.2.1 Macrophages in hypoxia10
1.2.2 Hypoxia responsive transcription factors11
1.2.3 Hypoxia-Inducible Factors (HIFs)12
1.2.4 Hypoxia Responsive Elements (HREs)15
1.2.5 Role of HIF-1 in macrophages15
1.2.6 Non-hypoxic up-regulation of HIF-117
1.2.6.1 Lipopolysaccharide (LPS)17
1.2.6.2 Phosphoinositide (PI) 3-kinase signaling18
1.2.6.3 Cobalt (CoCl ₂) stabilisation of HIF-1 α 19
1.2.6.4 Desferrioxamine (DFO) stabilisation of HIF-1 α
1.3 Extracellular matrix

1.3.1 Versican structure	22
1.3.2 Versican isoforms	23
1.3.3 Versican promoter	25
1.3.4 Versican function	27
1.3.4.1 Cell adhesion	
1.3.4.2 Cell Proliferation	29
1.3.4.3 Cell migration	
1.3.4.4 Extracellular matrix assembly	
1.3.5 Versican regulation	
1.3.5.1 Signal transduction pathways	
1.3.5.2 Transcription factors	
1.4 Aims of the project	
Chapter 2: Materials	
Chapter 2: Materials	38
Chapter 2: Materials 2.1 Reagents 2.1.1 Cell culture	
Chapter 2: Materials 2.1 Reagents 2.1.1 Cell culture 2.1.2 Reverse transcription and Real-Time PCR	 38
Chapter 2: Materials 2.1 Reagents 2.1.1 Cell culture 2.1.2 Reverse transcription and Real-Time PCR 2.1.3 Molecular cloning	 38
Chapter 2: Materials 2.1 Reagents 2.1.1 Cell culture 2.1.2 Reverse transcription and Real-Time PCR 2.1.3 Molecular cloning 2.1.4 Nuclear extraction	 38
Chapter 2: Materials	 38
Chapter 2: Materials	38 38 38 39 40 41 42 42
Chapter 2: Materials	 38
Chapter 2: Materials	38 38 38 38 39 40 40 41 42 42 42 42 42 43 43
Chapter 2: Materials	38 38 38 38 39 40 41 42 42 42 43 44 45

2.3.2 Media for bacteria	46
2.4 Primers	47
2.4.1 Primers for Real- Time PCR	47
2.4.2 Primers for PCR cloning	48
2.5 Recipies	49
2.6 Antibodies	54
2.7 Reporter plasmids	54
2.7.1 pGL4.10 [luc2] Luciferase reporter vector (#E6651, Promega)	54
2.7.2 pRL-TK vector (#E2241, Promega, Southampton, UK)	57
2.8 Equipments	58
Chapter 3: Methods	59
3.1 Cell culture	59
3.1.1 Isolation of peripheral blood mononuclear cells (PBMCs)	59
3.1.2 Determination of cell concentration	60
3.1.3. Generation of adherence-purified human monocyte- derived macrophag (HMDM)	es 61
3.1.4 Magnetic cell sorting system (MACS)	61
3.1.5 Culture of U937	62
3.2 Cell treatment	63
3.2.1 Normoxic & Hypoxic Incubation	63
3.2.2 Lipopolysaccharide (LPS) stimulation of adherence-purified HMDM	64
3.2.3 Cobalt chloride (CoCl ₂) stimulation of adherence-purified HMDM	64
3.2.4 Desferrioxamine (DFO) stimulation of adherence-purified HMDM	64
3.2.5 Ferric citrate stimulation of adherence-purified HMDM	65
3.2.6 Actinomycin D treatment of adherence-purified HMDM	65

3.2.7 PI3-Kinase inhibitor treatment of adherence-purified HMDM	65
3.3 General molecular cloning	66
3.3.1 Amplification of the versican promoter by polymerase Chain Reaction	66
3.3.2 Agarose gel electrophoresis	69
3.3.3 Restriction digestion	72
3.3.4 QIAquick DNA Agarose Gel Extraction	74
3.3.5 Ligation of DNA into plasmid vector	74
3.3.6 Transformation of plasmid-DNA into DH5- α competent cells	78
3.3.7 Purification of plasmid-DNA	79
3.3.8 Quantification of nucleic acids	81
3.4 Quantification of mRNA	82
3.4.1 Isolation of total RNA	82
3.4.2 Reverse transcription	83
3.4.3 Real-Time PCR	84
3.4.4 Quantification of mRNA	87
3.5 Transfection	88
3.5.1 JetPEI [™] DNA transfection reagent	88
3.5.2 Reporter gene assay	90
3.5.3 Luciferase assay system	91
3.5.4 Measuring protein concentration	93
3.6 Nuclear Extraction	94
3.7 Western blotting	96
3.7.1 Immunodetection of β -catenin	97
3.7.2 Re-probing the membranes	97

3.8 Flow cytometric analysis of versican protein expression	
3.9 Database analysis	101
3.10 Statistic	101
Chapter 4: Results-Hypoxic regulation of versican transcription in primary he macrophages	uman 102
4.1 Introduction	102
4.2 Short term hypoxia induces versican mRNA in adherence-purified HMDM	103
4.3 Comparison of versican mRNA hypoxic up-regulation in macrophages lymphocytes	and
4.4 Hypoxic up-regulation of versican mRNA occurs at the transcriptional level	112
4.5 Summary	114
Chapter 5: Results-Hypoxic regulation of versican protein synthesis in primary hemacrophages	uman 116
5.1 Introduction	116
5.2 Long term hypoxia induces versican protein expression in human pri macrophages	imary 117
5.3 Short term hypoxia did not induce versican protein expression in human pri macrophages	imary 122
5.4 Hypoxia-induced increases of versican protein expression are confine monocyte/macrophage cells	ed to 124
5.5 Summary	127
Chapter 6: Results-Versican promoter analysis	128
6.1 Introduction	128
6.2 PCR amplification of different regions of versican promoter	130
6.3 Cloning and screening of different fragments of versican promoter	134
6.4 Transfection optimisation using JetPEI TM in HMDM	143
6.5 Effect of 24 hours hypoxia on versican promoter reporter constructs expre in HMDM	ession 146

6.6 Effect of 5 days hypoxia on versican promoter reporter constructs expression in HMDM
6.7 Expression of versican promoter reporter constructs in hypoxia
6.8 Summary 155
Chapter 7: Results-Role of HIF-1 in regulation of versican gene expression in human monocyte-derived-macrophages
7.1 Introduction
7.2 HIF-1 α over-expression does not affect the expression of versican promoter reporter constructs in normoxic HMDM
7.3 LPS slightly induce versican mRNA expression in normoxic HMDM163
7.4 Desferrioxamine induces versican mRNA in HMDM but not cobalt chloride 165
7.5 Desferroxamine induces versican mRNA due to iron chelation
7.6 PI3 kinase inhibitors reduce hypoxic induction of versican mRNA in adherence- purified HMDM
7.7 Summary 173
Chapter 8: Discussion 175
Conclusion and future work 197
References
Appendix233

List of figures

Figure 1.1: Diagram illustrating haematopoietic stem cell derived from the bone marrow which gives rise to all types of blood cell2
Figure 1.2: Schematic diagram of macrophage structure and mechanisms of phagocytosis4
Figure 1.3: Regulation of HIF-1 in normoxia and hypoxia14
Figure 1.4: Schematic model of versican structure23
Figure 1.5: Cartoon of versican isoforms generated by alternative splicing of the mRNA transcript
Figure 1.6: Sequence analysis of versican gene promoter and exon 1
Figure 1.7: Schematic model of versican promoter regulation via PI3/PKB signalling and β -catenin TCF transcription factor complex suggested by Rahmani et al 200533
Figure 1.8: Effect of hypoxia on versican mRNA expression in PBMC
Figure 2.1: Generic pGL4.10 [luc2] Vector map56
Figure 2.2: The multiple cloning region of the pGL4 Vectors56
Figure 4.1: LightCycler PCR amplification of β -2M cDNA in normoxic (N, 20.9% O ₂) and hypoxic (H, 0.2% O ₂) adherence-purified HMDM105
Figure 4.2: LightCycler melting curve of amplified β-2M cDNA in the samples described in Fig 4.1106
Figure 4.3: LightCycler PCR amplification of versican cDNA in normoxic (N, 20.9% O_2) and hypoxic (H, 0.2% O_2) adherence-purified HMDM107
Figure 4.4: LightCycler melting curve of amplified versican cDNA in samples detailed in Fig 4.3
Figure 4.5: Effect of short term (18 hours) hypoxia on versican mRNA expression in adherence-purified HMDM109
Figure 4.6: Effect of short term (18 hours) hypoxia on versican mRNA expression in human primary macrophages and Lymphocytes

Figure 4.7: Effect of hypoxia on versican mRNA decay in adherence-purified HMDM113
Figure 4.8: Effect of hypoxia on GLUT-1 mRNA decay in adherence-purified HMDM114
Figure 5.1: Effect of long term (5 days) hypoxia on versican protein expression in brefeldin A-untreated PBMC-derived macrophages119
Figure 5.2: Effect of long term (5 days) hypoxia on versican protein expression in brefeldin A treated PBMC-derived macrophages
Figure 5.3: Effect of short term (18 hours) hypoxia on versican protein expression in PBMC derived macrophages treated with brefeldin A123
Figure 5.4: Intracellular flow cytometry analysis of hypoxic regulation of versican protein in human PBMC126
Figure 6.1: 0.6% agarose gel electrophoresis of PCR amplification of the 4.140-Kb versican promoter sequence from human genomic DNA130
Figure 6.2: 1.5% agarose gel electrophoresis of PCR amplification of the 240-bp, 480- bp and 830-bp versican promoter sequences from human genomic DNA131
Figure 6.3: 2% agarose gel electrophoresis of PCR amplification of the 110-bp, 160-bp, 210-bp, 80-bp, 130-bp, 180-bp and 210-bp versican promoter sequences from human genomic DNA
Figure 6.4: 2% agarose gel electrophoresis of the annealed 29-bp random nucleotide sequence
Figure 6.5: 0.6% agarose gel electrophoresis of restriction digested plasmid preps of 4.140-Kb of versican promoter pGL4.10 constructs135
Figure 6.6: 1.5% agarose gel electrophoresis of restriction digested plasmid preps of 480-bp of versican promoter pGL4.10 constructs136
Figure 6.7: 1.5% agarose gel electrophoresis of restriction digested plasmid preps of 240-bp versican promoter pGL4.10 constructs136
Figure 6.8: 1.5% agarose gel of restriction digested plasmid preps of 830-bp versican promoter pGL4.10 constructs137
Figure 6.9: 2% agarose gel electrophoresis of restriction digested plasmid preps of 110- bp, 160-bp and 210-bp (-56bp/+154bp) versican promoter pGL4.10 constructs138

Figure 6.10: 2% agarose gel electrophoresis of restriction digested plasmid preps of 80- bp versican pGL4.10 construct
Figure 6.11: 2% agarose gel electrophoresis of restriction digested plasmid preps of 130-bp and 180-bp versican pGL4.10 construct140
Figure 6.12: 2% agarose gel electrophoresis of restriction digested plasmid preps of 210-bp (-26bp/+184bp) versican pGL4.10 construct141
Figure 6.13: 2% agarose gel electrophoresis of restriction digested plasmid preps of random pGL4.10 construct142
Figure 6.14: Effect of 24 hours hypoxia on versican promoter reporter construct activity in HMDM147
Figure 6.15: Effect of 5 days hypoxia on versican promoter reporter construct activity in HMDM149
Figure 6.16: Effect of 5 days normoxia and hypoxia on 240-bp (-56bp/+184bp) and deleted versican promoter reporter construct in HMDM
Figure 6.17: Putative transcription factor binding sites within the 240-bp (- 56bp/+184bp) versican promoter sequence154
Figure 7.1: Effect of over-expression of HIF-1α on the 2.5-Kb (-2.316Kb+184bp) versican promoter construct in normoxic HMDM161
Figure 7.2: Effect of over-expression of HIF-1 α on the 240-bp (-56bp+184bp) versican promoter construct in HMDM162
Figure 7.3: Effect of LPS on VEGF mRNA expression in comparison with hypoxia164
Figure 7.4: Effect of LPS on versican mRNA expression in comparison with hypoxia164
Figure 7.5: Effect of 18 hours hypoxia on versican, VEGF and GLUT-1 mRNA expression in adherence-purified HMDM166
Figure 7.6: Effect of desferrioxamine on versican, VEGF and GLUT-1 mRNA expression in adherence-purified HMDM167
Figure 7.7: Effect of cobalt chloride on versican, VEGF and GLUT-1 mRNA expression in adherence-purified HMDM167
Figure 7.8: Effect of iron on versican mRNA induction by desferrioxamine in adherence-purified HMDM169

Figure 7.9: Effect of PI3-kinase inhibitors on versican gene expression in adherence- purified HMDM
Figure 7.10: Effect of hypoxia on β-catenin protein expression in U937 cell nuclear extracts
Figure 8.1: The versican 3'-untranslated region (UTR) from base pairs 2188-2428 is shown

List of tables

Table 2.1: list of reagents used in the cell culture
Table 2.2: list of reagents used in RNA isolation, reverse transcription and RT-PCR39
Table 2.3: list of reagents used in general molecular cloning40
Table 2.4: list of reagents used for nuclear protein extraction41
Table 2.5: list of reagents used in western blotting42
Table 2.6: list of reagents used in flow cytometry42
Table 2.7: list of reagent kits used in the project43
Table 2.8: list of materials used in the project44
Table 2.9: Media used for cell culture45
Table 2.10: Media used for bacterial cultures46
Table 2.11: List of primers used for Real Time-PCR47
Table 2.12: list of primers used for PCR cloning48
Table 2.13: list of recipes for all buffers and solutions49
Table 2.14: List of antibodies used for immunoblotting and flow cytometryexperiments
Table 2.15: List of equipments58
Table 3.1: PCR reaction mix for 4.140-Kb (-3.956Kb/+184bp), 830-bp (-646bp/+184bp), 480-bp (-296bp/+184bp) and 240-bp (-56bp/+184bp) of versican promoter construct
Table 3.2: PCR reaction mix for210-bp (-56bp/+154bp), 160-bp (-56bp/+104bp) and 110-bp (-56bp/+54bp), 210-bp (-26bp/+184bp), 180-bp (-26bp/+154bp), 130-bp (-26bp/+104bp), 80-bp (-26bp/+54p) constructs of versican promoter
Table 3.3: PCR set-up for amplification of 4.140-Kb (-3.956Kb/+184bp), 830-bp (- 626bp/+184bp), 480-bp (-298bp/+184bp) and 240-bp (-56bp/+184bp) of versican Promoter

Table 3.4: PCR set-up for amplification of 210-bp (-56bp/+154bp), 160-bp (-56bp/+104bp) and 110-bp (-56bp/+54bp), 210-bp (-26bp/+184bp), 180-bp (-26bp/+154bp), 130-bp (-26bp/+104bp), 80-bp (-26bp/+54) constructs of versican promoter
Table 3.5: Electrophoretic separation of PCR amplified fragments of the versicanPromoter
Table 3.6: Sfil digestion reaction mixture for PCR amplified fragments of versican promoter
Table 3.7: <i>Sfi</i> l digestion reaction mixture for pGL4.10 luciferase reporter plasmid73
Table 3.8: Different ratios of vector DNA (pGL4.10 vector): insert DNA (digestedversican promoter product)
Table 3.9: Reaction mixture for ligation of the digested versican promoter sequencesinto the pGL4.10 vector
Table 3.10: Reaction mix for reverse transcription
Table 3.11: Reaction master mix for Real-Time PCR
Table 3.12: PCR cycle used for amplification and detection mRNA of β 2-M, Versican, VEGF and GLUT-1 genes
Table 3.13: Instrument settings for the acquisition of human PMBCs100

Table 6.1: Transfection optimization in HMDM using different amounts of 2.5-Kb versican promoter pGL3-basic construct and JetPEI DNA transfection reagent.......145

1. Introduction

Chapter 1: Introduction

1.1 Macrophages

1.1.1 Origin of macrophages

Leukocytes are a diverse group of cells which regulate the body's immune response against infectious disease and foreign materials (Crowell et al., 1977; Stock et al., 2000). They circulate in the blood and/or lymphatic system so they can be recruited to areas where tissue damage or an infection has developed (van Furth and Cohn, 1968). Leukocytes consist of different subsets including lymphocytes, neutrophils, eosinophils, basophils, mast cells and monocytes (which differentiate into macrophages). These subsets are distinguished by functional and physical characteristics but all are derived from cells in the bone marrow known as hematopoietic stem cells (Fig 1.1) (Leder et al., 1967; Fogg et al., 2006). The large phagocytic mononuclear leukocytes represent a population of bone marrow-derived (myeloid) cells which are known as monocytes (Leder et al., 1967; Beekhuizen et al., 1993). Monocytes constitute ~5-10% of leukocytes in the peripheral blood, where they circulate for several days before populating tissues as macrophages, in the steady state or during inflammation (Issekutz et al., 1981; Gordon and Taylor, 2005). At sites of injury or microbial invasion, monocytes express chemokine receptors such as CCR2 and chemoattractants such as MCP-1 (Monocyte Chemoattractant Protein 1, also called CCL2, the ligand for CCR2) which elicits increased recruitment of monocytes to peripheral sites where they differentiate into macrophages and contribute to host defence, tissue remodelling and repair (Kuziel et al., 1997; Mutsaers et al., 1997; Lu et al., 1998; Smith et al., 2005).

The appearance of macrophages differs according to the tissues in which they reside, where they are classified and named differently. For instance, macrophages are known as Kupffer cells in the liver, red pulp macrophages in the spleen, microglia in the brain, reticular cells in the sinuses of spleen and lymph nodes, Langerhans cells in the skin and alveolar macrophages in the lungs (Dijkstra et al., 1985; Gordon and Taylor, 2005). Macrophages are "professional" phagocytic cells which act as an early line of defence in the immune system by recognising and engulfing pathogens such as bacteria and viruses (Serbina et al., 2008).



Figure 1.1: Diagram illustrating hematopoietic stem cell derived from the bone marrow which gives rise to all types of blood cells. Lymphocytes come from lymphoid progenitor cells and include B-cells, T-cells, and natural killer (NK) cells. Myeloid progenitor cells produce erythrocytes (red blood cells), megakaryocytes and platelets, granulocytes (including neutrophils, eosinophils and basophils) and monocytes, which differentiate into macrophages.

http://stemcells.nih.gov/StaticResources/info/scireport/images/figure51.jpg

1. Introduction

1.1.2 Role of macrophages in the body

Phagocytosis is the uptake of particles and it is an important function of macrophages, which are key players in the immune response and remove microorganisms and damaged or apoptotic cells of the host (Dentan et al., 1996; Kwiatkowska et al., 1999). An initial event, which is required for phagocytosis, is adhesion of particles by macrophages (Cougoule et al., 2004). It has been shown that adhesion structures (podosomes) and integrins which mediate cell adhesion to the extracellular matrix are involved in adhesion of macrophages to particles (Berton and Lowell, 1999; Calle et al., 2006). Phagocytosis is believed to be involved in macrophage activation and results in the release of cytokines such as IL-1 (Interleukin-1), IL-6 and TNF (Tumour Necrosis Factor) which promote inflammation (Nathan, 1987; Uchimura et al., 1997; Patel et al., 2003; Tripathi and Sodhi, 2008). After phagocytosis, membrane remodelling of macrophages leads to the complete enveloping of the particles in endosomes known as phagosomes. Phagosomes are membrane-bound organelles released into the cytoplasm where lysosomes fuse with them and release their contents into the phagosomes to kill, digest and degrade the contents (Fig 1.2) (Jutras et al., 2005; Stuart and Ezekowitz, 2005).



Figure 1.2: Schematic diagram of macrophage structure and mechanisms of phagocytosis. Phagocytosis begins with the macrophage flowing around the pathogen and engulfing it. Pathogens are enclosed in a phagosome (phagocytic vesicle). The next step is the fusion of lysosomes with the phagosome. The result is called a phagolysosome. Lysosomes are derived from the Golgi apparatus and their contents are focused on destroying microorganisms.

http://www.britannica.com/EBchecked/topicart/283636/138573/Macrophages-theprincipal-phagocytic-components-of-the-immune-system-ingest

Macrophage activation is closely related to defence against microbial infection, causing inflammation and other reactions which act against foreign body or particle invasion (van Rooijen et al., 1997; Smythies et al., 2005). Macrophages can act as antigen presenting cells (APC), which participate in the initial capture and processing of antigens (innate immunity) and then in the activation of specific T and B lymphocyte mechanisms (adaptive immunity). The activated T lymphocytes in turn cooperate with macrophages and enhance destruction of intra- and extra-cellular pathogens through phagocytosis (Grey et al., 1982; Unanue, 1984; Gordon and Taylor, 2005).

1. Introduction

The first step in the macrophage response to foreign organisms is to recognize them via pattern recognition receptors (PRRs) which are activated and affected immediately by conserved molecular patterns (Pathogen Associated Molecular Patterns, or PAMPs) carried by microbes, and are able to recognize many different classes of macromolecules (Gordon, 2002; Janeway and Medzhitov, 2002).

Macrophages express PRRs such as toll like receptors (TLRs) which are transmembrane proteins implicated in differential recognition of microbial and foreign molecular targets (Akira et al., 2001). For example, TLR4 is known to be involved in the recognition of lipopolysaccharide, a major cell wall component of Gram-negative bacteria (Triantafilou, M and Triantafilou, K, 2002). Recognition of LPS also requires other molecules such as CD14, a co-receptor expressed by monocytes and macrophages, in addition to TLR4. Activation of the CD14-TLR4 complex in monocyte/macrophages results in the induction of the adaptive immune response and triggers expression of multiple genes involved in immune responses (Akira et al., 2003; Miyake et al., 2004; Takeda and Akira, 2005).

Many other PRRs are expressed by macrophages, including scavenger receptors (SR) and mannose receptors (MR) (Krieger et al., 2001; Taylor et al., 2005). Scavenger receptors are primarily involved in the recognition, binding, phagocytosis and clearance of cellular components, such as modified low-density lipoprotein (LDL) and apoptotic cells (Peiser et al., 2002), whereas the mannose receptor is involved in innate and adaptive immune responses (East and Isacke, 2002). The mannose receptor is an endocytic and phagocytic receptor which binds to carbohydrate structures on the

surface of many pathogens, including bacteria, fungi, parasites and viruses, and leads to phagocytosis of these particles by macrophages (Llorca et al., 2008). It has been clearly demonstrated that activation of PRRs such as TLRs, SRs and MRs lead to increased phagocytosis of microorganisms by macrophages (Pearson, 1996; Fraser et al., 1998; Underhill et al., 1999).

1.1.3 The inflammatory macrophages

Inflammation is a response of a tissue to injury which could be a simple wound or a complex autoimmune inflammation such as rheumatoid arthritis (Schmid-Schonbein, 2006). It has been shown that macrophages are major players in the inflammatory response and secrete pro-inflammatory and antimicrobial mediators (Gordon, 1998; Ma et al., 2003). For example it has long been known that macrophages activated in vitro by interferon-y (IFN-y) followed by a microbial trigger can increase production of pro-inflammatory cytokines such as tumour necrosis factor and interleukins including IL-1 and IL-6 (Gordon, 2003). Also innate activation of macrophages by ligation of TLRs such as TLR-4 with LPS is associated with microbicidal activity and production of other pro-inflammatory cytokines such as IFN- α and IFN- β (Gao et al., 1998; Fujihara et al., 2003). Evidence to date suggests that macrophages-derived cytokines such as transforming growth factor- β (TGF- β), basic fibroblast growth factor and plateletderived growth factor are important in tissue repair and remodelling (Nathan et al., 1987; Ricardo et al., 2008). In addition, it has also been shown that deactivation of macrophages, which is induced by presence of cytokines such as IL-10 or TGF- β , is associated with increased production of IL-4, an anti-inflammatory cytokine (Goerdt and Orfanos, 1999; Gordon, 2003).

Several studies have suggested that macrophages can be classified into two major groups, M1 and M2 (Edwards et al., 2006; Mantovani et al., 2006; Gordon and Martinez, 2010). M1 macrophages are activated by IFN- γ , TNF or pathogen-associated molecular patterns such as LPS and can effectively destroy invading pathogens, tumour cells and foreign materials (Gordon, 2003; Mantovani et al., 2004). They act as antigen presenting cells and release pro-inflammatory cytokines such as TNF, IL-6, IL-1 and IL-12 and participate as inducer and effector cells in T helper 1 (Th1) responses (Gordon, 2003; Mantovani et al., 2004; Van Ginderachter et al., 2006). Accumulating evidence suggests that M2 macrophages, which result from culture in IL-4, IL-13, IL-10 or TGF- β , can release anti-inflammatory cytokines, growth factors and mediators which are involved in wound repair and tissue remodelling, and contribute as inducers in T helper 2 (Th2) responses (Gordon, 2003; Mantovani et al., 2004; Benoit et al., 2008; Laskin, 2009).

Overall there are many stimuli which can push macrophages toward the activation phenotype. The hypoxia which often occurs in tumours and sites of infection can therefore activate macrophage expression of a broad range of genes including proinflammatory cytokines such as TNF- α , IL-1 and IL-6 (Ghezzi et al., 1991; Scannell et al., 1993; Albina et al., 1995).

1. Introduction

1.2 Hypoxia

Molecular oxygen is essential for aerobic metabolism to maintain intracellular bioenergetics and to serve as an electron acceptor in many reactions (Freeman, 2000). Ambient air is 21% O₂ (150 mm Hg) at sea level; however, most mammalian tissues have O₂ levels of 24 to 66 mm Hg (2%–9% O₂) (Vaupel et al., 1989). Hypoxia is a term which describes low oxygen concentrations (Semenza, 2010), which can affect and regulate many physiological and pathophysiological processes, including embryonic development (Simon and Keith, 2008) and wound healing (Lee et al., 2009). In biological systems, hypoxia usually occurs in pathological tissues including tumours, ischemic tissues, chronic obstructive pulmonary disease, atherosclerotic plaques (Bjornheden et al., 1999; Brahimi-Horn and Pouyssegur, 2006), and arthritic joints (Stevens et al., 1991). It is known that a major obstacle to cell survival is reduction in oxygen availability, which is often confronted by cancer cells (Sutherland, 1998; Vaupel et al., 2004). In general, rapid growth and abnormal angiogenesis at the site of the tumour leads to insufficient blood supply and consequent depletion of oxygen. This eventually results in the formation of necrotic and hypoxic regions in the inner parts of the tumour (Griffiths et al., 2000; Vaupel et al., 2004). Work by Vaupel and Meyer showed that O₂ concentrations within cancers are reduced compared to surrounding normal tissue, with severe hypoxia correlating with invasion, metastasis and patient death (Vaupel and Mayer, 2007). The oxygen concentration in these pathological tissues ranged from 0 to 15mmHg (Hockel and Vaupel, 2001). Hypoxia is also found in healthy tissues such as the spleen (oxygen levels as low as 0.5% or 3 mmHg) (Caldwell et al., 2001) and it is also a condition encountered in embryogenesis, in which hypoxic

signalling is considered necessary for normal development (Brahimi-Horn and Pouyssegur, 2006).

A role for a hypoxic microenvironment in the pathogenesis and progression of human cancer was first proposed by Gray et al, when intratumoral hypoxia was correlated with reduced efficiency of radiation therapy (Gray et al., 1953; Brizel et al., 1999; Overgaard, 2007). Hypoxia has also been shown to be linked to increased mutation rates (Yuan and Glazer, 1998), tumour invasion (Pennacchietti et al., 2003) and metastasis (Subarsky and Hill, 2003).

Genomic tools, including DNA microarrays, have enabled study of the global gene expression of many different cells and tissues under hypoxic stress (Burke et al., 2003; Greijer et al., 2005; Weinmann et al., 2005) and more than 100 genes have been shown to be up-regulated by hypoxia. For example, hypoxia induces erythropoietin (EPO) (Wang and Semenza, 1993_a), and angiogenic cytokines such as vascular endothelial growth factor (VEGF) (Liu et al., 1995) and basic fibroblast growth factor (bFGF), which are required for the adaption of the whole organism to general hypoxia by enhancement of blood oxygen-carrying capacity and oxygen delivery (Egger et al., 2007). Also hypoxic up-regulation of glucose transporter-1 (GLUT-1), which facilitates the transport of glucose across the plasma membranes of mammalian cells, has been detected in a variety of malignant tissues (Airley et al., 2001; Macheda et al., 2005).

It is well known that a variety of signalling pathways are activated by hypoxia (Haddad, 2004; Benizri et al., 2008). Among these, the activation of the transcription factor hypoxia-inducible factor 1 (HIF-1) is a key element responsible for embryogenesis and

up-regulation of numerous hypoxia inducible genes (Kallio et al., 1998; Sutter et al., 2000). HIF-1-mediated gene expression allows an organism to respond to hypoxia by increasing oxygen delivery or adapting to decreased oxygen availability (Wenger, 2002). Such targets of HIF-1 play critical roles in glycolysis, oxygen homeostasis, tissue remodelling, fat metabolism, angiogenesis, erythropoiesis, and proliferation (Semenza, 2001_b; Hirota, 2002).

1.2.1 Macrophages in hypoxia

It has been known for some time that macrophages are recruited and retained in poorly vascularised, hypoxic and necrotic sites including in breast (Leek et al., 1999; Bingle et al., 2002) and ovarian carcinomas (Negus et al., 1997), wounds (Crowther et al., 2001), atherosclerotic plaques (Husain et al., 1999) and arthritic joints (Kinne et al., 2000). In addition, It has been reported that chemoattractants such as colony stimulating factor 1 (CSF-1), MCP-1, VEGF and endothelin 1 recruit peripheral monocytes to tumour regions which are characterised by extremely low levels of oxygen and trigger differentiation into tumour associated macrophages (TAMs) (Murdoch et al., 2004). Several studies have shown that these TAMs release a variety of enzymes and cytokines which promote tumour invasion, angiogenesis and metastasis, such as epidermal growth factor (EGF) and VEGF (Lewis et al., 2000; Goswami et al., 2005; Bingle et al., 2006; Lewis and Pollard, 2006; Mantovani et al., 2006).

A study by Burke et al., (2003) showed that certain genes are up-regulated by macrophages under hypoxic conditions. They used cDNA array hybridization to

determine the effect of hypoxia on mRNA of 1185 genes in primary human monocytederived macrophages (HMDM). This study showed hypoxia induced mRNA upregulation of the enzyme matrix metalloproteinase-7 (MMP-7), neuromedin B receptor and DNA-binding protein inhibitor (Id2) as well as known hypoxia inducible genes such as VEGF and GLUT-1 (Burke et al., 2003). Other cDNA array work by White et al., (2004) also revealed more than 30 mRNA pro-angiogenic genes which were upregulated by hypoxia in primary macrophages. The best characterized of these, apart from VEGF, were fibroblast growth factor 2 (FGF2), IL-8, macrophage migration inhibitory factor (MIF) and cyclooxygenase- 2 (COX-2) (White et al., 2004). In addition, It has been demonstrated that hypoxic macrophages up-regulate a number of transcription factors, such as HIF-1, which in turn up-regulate a broad array of genes including VEGF and GLUT-1 whose products promote tumour growth and angiogenesis (Burke et al., 2002 & 2003; Leek et al., 2002; Elbarghati et al., 2008).

1.2.2 Hypoxia-responsive transcription factors

Hypoxia activates a diverse array of transcription factors such as activator protein-1 (AP-1) (Bandyopadhyay et al., 1995; Rupec and Baeuerle, 1995; Fantozzi et al., 2003), cAMP-response element binding protein (CREB) (Beitner-Johnson and Millhorn, 1998; Leonard et al., 2008), specific protein 1 (SP-1) (Xu et al., 2000; Lee et al., 2004; Sanchez-Elsner et al., 2004) and most importantly HIF-1 (Semenza et al., 1994), which in turn activates a broad array of mitogenic, proinvasive, proangiogenic, and prometastatic genes (Semenza, 1999 & 2000_b). Since the discovery of HIF-1 by the Semenza lab in the early 1990s, it has been recognised as being of central importance

and has been described as the "master regulator" of the transcriptional response to hypoxia (Semenza, 1999).

1.2.3 Hypoxia-Inducible Factors (HIFs)

There are two main types of HIF, HIF-1 and HIF-2 (Wang and Semenza, 1995; Ema et al. 1997) which are the predominant transcription factors mediating the effects of hypoxia on gene expression (Semenza, 2001_a; Bertout et al., 2008). HIF-1 is the most ubiquitously expressed and best characterised of the family and is recognised as a master regulator of hypoxic signalling whose activation has been shown to regulate expression of over 70 genes at the transcriptional level (Semenza, 2004).

Both HIF-1 and 2 are heterodimeric molecules consisting of α and β subunits which belong to a family of basic helix-loop-helix proteins (Dery et al., 2005). The HIF- β subunit, also known ARNT (Aryl hydrocarbon Receptor Nuclear Translocator) is found in the nucleus in both normoxia and hypoxia (Wang and Semenza, 1995), whereas the α subunit is also constitutively produced but is subject to rapid degradation in the presence of oxygen (half-life less than five minutes), only becoming stable in the absence of oxygen (i.e. hypoxia) (Semenza, 2004). The HIF-1 α subunit contains an oxygen-dependent degradation domain (ODD) and two transactivation domains (TAD) which are required for the transcriptional activation activity of HIF-1, being capable of binding to two transcriptional co-activators, CREB binding protein (CBP) and p300 (Kallio et al., 1998; Ema et al., 1999; Jiang et al., 1997; Pugh et al., 1997 & 2003). In normoxia, specific proline residues at positions 402 and/or 564 in the ODD of the HIF- α subunit are hydroxylated by prolyl hydroxylase enzymes (PHD) (Semenza, 2000_a;

Jaakkola et al., 2001). PHD is a family composed of prolyl 4-hydroxylases (PHD1-4) which require iron (Fe (II)), 2-oxoglutarate, O_2 and ascorbate as substrates; their activity is reduced in hypoxia due to the limitation of O_2 (Epstein et al., 2001; Ivan et al., 2001). Hydroxylation of HIF-1 α acts as a signal for recognition by the tumour suppressor VHL (von Hippel–Lindau protein), leading to ubiquitination and proteasomal degradation (Fig 1.3) (Maxwell et al., 1999; Kamura et al., 2000; Tanimoto et al., 2000).

In addition to prolyl hydroxylation by PHDs, another oxygen-dependent modification occurs in the transactivation domains of HIF- α subunit. It is dependent on the presence of an asparagine hydroxylase enzyme known as factor inhibiting HIF (FIH) (Mahon et al., 2001; Cockman et al., 2009). In this oxygen-dependent regulatory mechanism, FIH blocks the interaction between HIF- α with p300 and CBP by hydroxylating an asparagine residue at position 803, thus inhibiting the activity of the HIF-1 α transactivation domain (Fig 1.3) (Hewitson et al., 2002; McNeill et al., 2002).

In hypoxia, PHD activity decreases which enables rapid accumulation of HIF- α in the nucleus where it dimerises with the HIF- β subunit and binds to hypoxia response elements (HREs) in the promoters of various genes (Sutter et al., 2000; Jewell et al., 2001). The decrease in oxygen availability also impairs FIH which results in a decrease in HIF- α subunit asparagine hydroxylation, allowing increased recruitment of transcriptional co-activators (p300/CBP) which eventually leads to the enhanced transcriptional activation of HIF target genes (Fig 1.3) (Wenger et al., 2005; Lando et al., 2002; Mahon et al., 2001) which are implicated in many different aspects of oxygen

delivery and metabolism including vasodilatation (nitric oxide synthases), iron metabolism (transferrin) (Ratcliffe et al., 1998), glucose transporters (GLUT-1), angiogenesis (VEGF), enhanced blood oxygenation (erythropoietin) (Gleadle et al., 1997) and glycolysis (phosphoglycerate kinase)(Semenza et al., 1994).



Figure 1.3: Regulation of HIF-1 in normoxia and hypoxia. Further details are given in the text.

1. Introduction

1.2.4 Hypoxia Responsive Elements (HREs)

Previous studies showed that HIF-1 binds to hypoxia responsive elements, a consensus sequence in the promoter of about 200 HIF target genes (among which around 100 genes have been confirmed) and initiates transcription by recruiting transcriptional co-activators such as p300/CBP (Semenza, 2000_b ; Wenger et al., 2005).

The minimal cis-regulatory element (CGTG) required for hypoxic induction of gene transcription was first identified by Semenza who also determined that this core HRE consensus sequence is required but not sufficient for effective gene activation in response to hypoxia (Semenza et al, 1994 & 1996). Analysis of 107 HIF-1 responsive genes has shown that neighbouring nucleotides occur with non-random frequency, especially in the 5' flanking bases, demonstrating that a fully functional HRE requires neighbouring DNA binding sites for additional transcription factors or co-activators, which may act to amplify the hypoxia response (Wenger et al., 2005).

1.2.5 Role of HIF-1 in macrophages

As previupsly mentioned, macrophages are associated with a number of inflammatory sites such as atherosclerotic plaques (Bjornheden et al., 1999), myocardial infarcts (Jurgensen et al., 2004), rheumatoid arthritis (Hollander et al 2001), healing wounds (Stevens et al., 1991), sites of bacterial infection, and malignant tumours (Blouin et al., 2004; Murdoch et al., 2005; Pouyssegur et al., 2006) in which hypoxia is present. In hypoxia, macrophages rely heavily on HIFs for energy generation and activity, and express HIF-1α protein abundantly and increase transcriptional activation of HIF target genes (Burke et al., 2002 & 2003). Unusually, however, macrophages are also

significantly dependent on HIF-1 regulated genes for energy generation in normoxia (Cramer et al. 2003).

Some early studies using a rat alveolar macrophage-derived cell line and the human monocytic cell line THP-1 reported that short term hypoxia did not increase HIF-1 α mRNA, suggesting instead that HIF-1 α is regulated by hypoxia by decreased protein stability (Blouin et al., 2004; Frede et al., 2006). However, a very recent study by us (Staples et al., 2011; see Appendix) showed increases in HIF-1 α mRNA levels after long term hypoxia (5 days) in human primary macrophages and reported that this up-regulation is mediated by increased transcription rather than increased mRNA stability. Similar increases in HIF-1 mRNA in hypoxia have been reported by other groups in non-macrophage cell types (discussed in Staples et al., 2011) but the subject is still somewhat controversial.

An increased level of HIF-1 α protein in activated macrophages was first demonstrated by Hollander et al., (2001) in inflamed joints of patients suffering from rheumatoid arthritis and later by Talks et al., (2000) in tumour sections and Burke et al., (2002) in isolated hypoxic human primary macrophages *in vitro*. Also other studies showed increased levels of HIF-1 α in the inflammatory cells of healing wounds and it was suggested that this could be due to a release of inflammatory cytokines such as TNF- α which can strongly increase HIF-1 α protein levels in cells after injury, leading to increased expression of HIF-1 responsive genes such as VEGF which regulate the process of tissue repair (Albina et al., 2001). Other groups have also investigated whether HIF-1 activity is induced during differentiation from monocytes to
macrophages (Knowles et al., 2006). It was shown by Oda et al., (2006) that both HIF-1 α and HIF-1 β protein levels markedly increase during the differentiation of monocytes to macrophages in the monocytic cell line THP-1 and in monocytes from human peripheral blood. They suggested that activation of protein kinase C (PKC) and mitogen-activated protein kinase (MAPK)-signalling pathways are responsible for this increase in HIF-1 gene expression (Oda et al., 2006).

1.2.6 Non-hypoxic up-regulation of HIF-1

Despite the name, numerous studies have now shown that HIF-1 α can be induced by a variety of stimuli in addition to hypoxia. The key studies in this area are reviewed below.

1.2.6.1 Lipopolysaccharide (LPS)

LPS is a component of the cell wall of gram-negative bacteria (Hajjar et al., 2002). It binds to the CD14 and TLR4 cell surface receptors of monocyte/macrophages (Hoshino et al., 1999; Moynagh, 2003) leading to the activation of a number of genes often associated with hypoxia, many of which are believed to be up-regulated independently of HIF-1 (Pocock et al., 2003; Covert et al., 2005; Karin and Greten, 2005; Karin, 2006). Several studies have shown that LPS treated macrophages up-regulate genes such as VEGF, GLUT-1 and iNOS which are known to be regulated by HIF-1 (Blouin et al., 2004; Frede et al., 2006). In contrast to hypoxia, which is generally considered not to upregulate HIF-1 α mRNA, LPS has been shown to stimulate HIF-1 α expression at the transcriptional level under normoxia in alveolar-derived rat macrophages and human primary macrophages through a NF- κ B site in the promoter of the HIF-1 α gene (Blouin et al., 2004; Frede et al., 2006). Blouin et al., (2004) showed that LPS increases HIF-1 α protein expression in a time and dose-dependent manner which in turn modulates hypoxic gene activation. Frede et al., (2006) reported induced HIF-1 α mRNA and protein expression in differentiated THP-1 cells treated with LPS under normoxia. This study, using RNAi against MAPK and also a specific inhibitor of this pathway, showed down-regulation of the LPS-induced HIF-1 α mRNA and protein in THP-1 cells suggesting a role for the MAPK pathway in LPS-dependent HIF-1 α induction (Frede et al., 2006).

1.2.6.2 Phosphoinositide (PI) 3-kinase signalling

PI3-Kinase activities have been found in all eukaryotic cell types and are linked to a diverse set of key cellular functions, including cell growth, proliferation, motility, differentiation and survival (Rameh and Cantley, 1999; Fry, 2001; Katso et al., 2001). PI3-kinase phosphorylates the 3 position hydroxyl group of the inositol ring of phosphatidylinositol (Zvelebil et al., 1996). PI3-kinase has been the focus of intense study as increasing evidence suggests a key role for the PI3-kinase pathway in many human diseases including allergy, inflammation, heart disease and cancer (Stein, 2001; Liu et al., 2009). An interesting mechanism was proposed by which the activation of PI3-kinase could increase the rate of HIF-1 α translation in normoxia (Page et al., 2002). It has been previously reported that activation of PI3-kinase by growth factors and hormones leads to the recruitment and activation of a downstream effector of PI3-kinase, known as the mammalian target of rapamycin (mTOR) (Treins et al., 2002; Alam et al., 2004). mTOR activation results in increased phosphorylation and inactivation of 4E-binding protein 1 (4E-BP), the eukaryotic translation initiation factor,

and activation of p70-S6 kinase 1 which leads to increased protein synthesis (Zhong et al., 2000; Treins et al., 2002). Inactivation of 4E-BP and activation of p70S6K has been shown to increase translation of HIF-1 α mRNA through the 5['] untranslated region (5[']UTR) (Richard et al., 2000; Page et al., 2002). This is believed to be the main effector mechanism responsible for HIF-1 α induction through the PI3-kinase dependent pathway, resulting in increased VEGF expression in vascular smooth muscle cells and human tumour cell lines (Jiang et al., 2001; Laughner et al., 2001).

1.2.6.3 Cobalt (CoCl₂) stabilisation of HIF-1α

It has been demonstrated that $CoCl_2$ induces hypoxia-regulated genes by stabilising HIF-1 α in normoxia (Maxwell et al., 1999). As outlined in section 1.2.3, hydroxylation of the proline residues, which reside in the oxygen-dependent degradation domain of HIF-1 α , by prolyl hydroxylase is one of the key mechanisms that mediate the binding of VHL with HIF-1 α which eventually leads to proteasomal degradation of HIF-1 α (Ivan et al., 2001; Jaakkola et al., 2001). A study by Epstein et al., (2001) suggested that iron is a critical factor for the activity of PHD as these enzymes have an iron-binding centre. In addition, this study suggested that $CoCl_2$ may act as a competitor for iron, inactivating PHD by binding and engaging an iron-binding site in the proline hydroxylase. Due to this enzymatic inhibition, HIF- α is not targeted for proteasomal degradation (Epstein et al., 2001). Beside the inactivation of PHD by $CoCl_2$, an investigation by Yuan et al., (2003) also suggested another mechanism in which HIF- α could be stabilized by cobalt. In this mechanism, cobalt stabilizes HIF-1 α protein by direct binding to the ODD in HIF-1 α , thereby preventing the interaction between HIF-

 1α and VHL protein and subsequently inhibiting proteasomal degradation which results in HIF-1 α stabilisation (Yuan et al., 2003).

1.2.6.4 Desferrioxamine (DFO) stabilisation of HIF-1α

Since the introduction of DFO in the 1960s, it has been widely used as a chelating agent to bind free iron in the bloodstream and removing excess iron from the body (Breuer et al., 2001). Several studies have demonstrated that normoxic cells treated with DFO induced HIF-1 target genes such as EPO (Wang and Semenza, 1993_b), VEGF (Agani and Semenza, 1998) and GLUT-1 (Dongiovanni et al., 2008) by inducing the accumulation of HIF-1 protein. An early study by Maxwell et al., (1999) demonstrated that DFO disrupts pVHL–HIF- α complex formation which is required for ubiquitination and proteasomal degradation of HIF-1 α in normoxia. It has been demonstrated that DFO inhibits hydroxylation of HIF-1 α by chelating the iron required for activity of the PHD enzyme (Yuan et al., 2003). Therefore due to inhibition of HIF- α stabilisation which results in induction of HIF-1 target genes (Mole et al., 2001; Pham et al., 2002; Woo et al., 2006).

1.3 Extracellular matrix

The extracellular matrix (ECM) provides structural support for organs and tissues (Badylak, 2002), cell layers in the form of basement membranes (Daley et al., 2008) and individual cells as substrates for migration (Werb, 1997). ECM is composed of collagens and elastic fibres which are embedded in a viscoelastic gel that comprises proteoglycans (e.g. versican and hyaluronan), glycoproteins and water (Berrier and Yamada, 2007; Campbell et al., 2010). ECM forms a complex, three-dimensional network among the cells of different tissues in an organ-specific manner (Adams and Watt, 1993) and plays vital roles in the differentiation, proliferation and survival of cells (Boudreau and Bissell, 1998; Streuli, 1999; van Horssen et al., 2007).

Proteoglycans are the main components of ECM, and are characterised by a protein portion (core protein) and one or more unbranched, long and negatively charged polysaccharide chains called glycosaminoglycans (GAG) which are covalently attached to the core protein (Hardingham and Fosang, 1992; lozzo and Murdoch, 1996). Depending upon the nature of the GAG chains, proteoglycans can be categorised as heparan sulphate proteoglycans, chondroitin sulphate proteoglycans (CSPGs) and dermatan sulphate proteoglycans, or keratan sulphate proteoglycans (lozzo, 1998; Kresse and Schonherr, 2001). Of these, the CSPGs such as versican are the most abundant type of proteoglycan in the ECM of mammalian tissues (Zimmermann and Ruoslahti, 1989; Carulli et al., 2005).

1.3.1 Versican structure

Versican is a large chondroitin sulphate proteoglycan which is a major component of the ECM (Naso et al., 1994). Versican is transcribed from a single gene which is localized on chromosome 5q 12-14 in the human genome and extends over 90 kb (lozzo et al., 1992) which is divided into 15 exons which range in size from 76 to 5262 bp (Naso et al., 1994). The alternative mRNA-splicing of these exons gives rise to four different versican isoforms which are distinguished by different core-middle regions (Dours-Zimmermann and Zimmermann, 1994). Versican is comprised of three domains. The amino terminal G1 domain interacts with a glycosaminoglycan (GAG) called hyaluronan present in the extracellular matrix (Zimmermann and Ruoslahti, 1989). The carboxyl terminal domain of versican is called the G3 domain and it contains a C-type lectin binding domain, two epidermal growth factor repeats and a complement regulatory region. The versican core protein contains the GAG attachment region and the chondroitin sulphate chains extend from this region of the protein (Fig 1.4) (Zimmermann and Ruoslahti, 1989; Zako et al., 1995; Yang et al., 2000).



Figure 1.4: **Schematic model of versican structure.** Versican contains globular domains at the amino terminus (G1) and carboxyl terminus (G3). The G1 domain is composed of an immunoglobulin-like motif, followed by two proteoglycan tandem repeats which bind hyaluronan (HABR; hyaluronan binding region). The G3 domain contains two epidermal growth factor-like repeats, a carbohydrate recognition domain (a lectin-binding domain) and complement binding protein (ELC). The core-middle region of versican contains GAG attachment regions that are encoded by exons 7 and 8 which give rise to four different versican isoforms. GAG chondroitin sulphate chains extend from the core protein (Wu et al., 2005)

1.3.2 Versican isoforms

The central area of versican is encoded by two exons that specify chondroitin sulphate attachment regions (Zimmermann and Ruoslahti, 1989). RNA splicing of these two exons generates four isoforms of versican named V0, V1, V2 and V3 core protein molecular weight of about 370 kDa, 263 kDa, 180 kDa, and 74 kDa, respectively (Dours-Zimmermann and Zimmermann, 1994). V0, the largest versican isoform, is encoded by exons 7 and 8 and contains the GAG- α and β domains. The V1 isoform contains GAG- β attachment domain which is encoded by exon 8 (lacking exon 7) whereas the V2 isoform contains a GAG- α domain which is encoded by exon 7 (lacking exon 8) (Lemire et al., 1999). V3 does not include either exon 7 or 8 and consequently

has no GAG attachment sites (Fig 1.5) (Zako et al., 1995; Lemire et al., 1999). An interesting study on human adult tissues showed that all four versican isoforms are transcribed in more than 50% of tissues including the brain and skin, although, intriguingly, only the V1 isoform is expressed in liver and spleen (Cattaruzza et al., 2002). Another study suggested that V1 versican enhances cell proliferation and protects mouse fibroblast cell lines from apoptosis whereas the V2 isoform exhibits opposite activities by inhibiting cell proliferation (Sheng et al., 2005). In addition, it has been reported that V0 and V1 are the predominantly expressed isoforms in tumours suggesting that these isoforms are mainly involved in tumour development (Touab et al., 2002; Arslan et al., 2007). In contrast, over-expression of V3 (the smallest versican isoform, which consists of only the G1 and G3 domains) in melanoma cancer cells markedly reduced cell growth, suggesting a role for V3 versican as an inhibitor of tumour growth (Serra et al., 2005; Hernandez et al., 2010).



Figure 1.5: Cartoon of versican isoforms generated by alternative splicing of the mRNA transcript. Different colours show specific domains in the gene. G1 and G3 are shown at the amino and carboxyl termini respectively. Purple = hyaluronan binding region (HABR); yellow = α GAG exon; red = β GAG exon and green = epidermal growth factor repeats (E), a lectin binding domain (L) and a complement regulatory region (C). The glycosaminoglycan chondroitin sulphate chains are shown as (/)

1.3.3 Versican promoter

The transcription start site for the versican gene was first identified by Naso et al., (1994). This study reported that the versican promoter contains a typical TATA box located approximately 16 base pairs upstream of the transcription start site (Fig 1.6). Transient transfections of a reporter construct driven by an 876-bp (-632/+240) fragment of the versican promoter showed significant expression in HeLa cells and IMR-90 embryonic lung fibroblasts. Furthermore, deletion constructs of the 876-bp confirmed that the human versican 5'-flanking sequence contains promoter, enhancer and repressor elements which are able to drive the expression of the versican reporter gene in different cells (Naso et al., 1994). In addition, sequence analysis has revealed potential binding sites for several transcription factors in the 876-bp versican promoter region including CREB (Naso et al., 1994), T-cell factor/ lymphoid enhancer-binding factor (TCF/LEF) (Rahmani et al., 2005), AP-1 and SP1 (Domenzain et al., 2009) (Fig 1.6).



Figure 1.6: Sequence analysis of versican gene promoter and exon 1. Naso et al., (1994) revealed a typical TATA box located approximately 16 bp upstream of the transcription start site and binding sites for a number of transcription regulatory factors including CREB at position -36bp, SP-1 at position -370bp, XRE at position - 509bp and AP-2 at positions -426bp, -355bp, -218bp and -190bp respectively. Rahmani et al., (2004) showed two binding sites for TCF/LEF at positions – 546bp and -492bp respectively. Also further investigation by Domenzain et al., (2009) revealed one binding site for SP-1 at position -166bp and one binding site for AP-1 at position -36bp. Positions of all transcription factors are relative to the transcription start site. The sequence of exon 1 is shaded. CREB; CAMP response element-binding, AP-1& 2; (activator protein 1&2), XRE; xenobiotic responsive element, TCF/LEF; T-cell factor/lymphoid enhancer factor.

1.3.4 Versican function

Versican is a main component of the ECM where its hygroscopic properties create a loose and hydrated matrix which is necessary to support key events in development (Lemire et al., 2002; Hinek et al., 2004). Increased expression of versican is observed in sites of tissue injury (Landolt et al., 1995) and in cancers including breast (Kischel et al., 2010), ovarian (Voutilainen et al., 2003), gastrointestinal tract (Theocharis, 2002), prostate (Sakko et al., 2003), brain (Schwartz and Domowicz, 2004), cervical (Kodama et al., 2007) and melanoma (Domenzain et al., 2003). Several reports have also highlighted the role of versican in wound healing (Cattaruzza et al., 2002), angiogenesis, tumour growth (Zheng et al., 2004) and in vascular diseases, especially atherosclerosis (Talusan et al., 2005; Kenagy et al., 2006). It has been demonstrated that versican binds low-density lipoprotein (LDL) particles and it is believed that accumulation of versican in blood vessels promotes extracellular lipoprotein retention, suggesting roles in lipid accumulation, inflammation, and thrombosis (Olin et al., 1999; Wight and Merrilees, 2004). Due to versican's structural composition and its widespread expression in the body it is able to regulate cell adhesion and survival, cell proliferation, cell migration and ECM assembly (Ricciardelli et al., 2009) that the key studies in these areas are reviewed below.

1.3.4.1 Cell adhesion

Early studies reported that most chondroitin sulphate proteoglycans may be considered as anti-adhesion molecules for the regulation of cell adhesion to the substratum, which is essential for various cell and tissue functions (Yamagata et al., 1989 & 1993). Different studies presented evidence suggesting that this inhibitory activity could be to due to the G1 domain of versican (Yamagata et al., 1989; Sugiura et al., 1993). They showed that selective exclusion of versican from podosomes of cultured human osteosarcoma cells suppresses the malignant cell-adhesive phenotype, suggesting that versican can act as an anti-adhesive molecule (Yamagata et al., 1994). However there is evidence that the carboxy-terminal domain of versican interacts with the β 1 integrin of brain tumour cells leading to the activation of focal adhesion kinase (FAK), promoting cell adhesion and protecting the cell from apoptosis (Wu et al, 2002 & 2004).

Interaction of versican with selectins and chemokines has been studied. It has been shown that versican binds to L-selectin, adhesion molecules on the surface of activated endothelial cells, through its chondroitin sulphate chains (Kawashima et al., 2002). In addition, versican has been shown to bind secondary lymphoid tissue chemokine (SLC) through chondroitin sulphate chains and this binding tends to downregulate chemokine function for recruitment of lymphocytes (Hirose et al., 2001). Taken together the data suggest that versican, which is induced in inflammatory conditions such as arthritis (Wight and Merrilees, 2004), asthma and lung disease (Johnson, 2001; Merrilees et al., 2004), may regulate inflammation by regulating interaction with selectins and chemokines (Hirose et al., 2001).

1.3.4.2 Cell Proliferation

Abundant expression of versican in fast growing tissues and cells suggested a key role for versican in cell proliferation (Bode-Lesniewska et al., 1996; Touab et al., 2002). For example high expression of versican is detected in the loose connective tissue of various organs including the central and peripheral nervous system (Bode-Lesniewska et al., 1996), blood vessels (Zheng et al., 2006), dermis and in the proliferative zone of the epidermis (Zimmermann et al., 1994).

Versican is also involved in the proliferation of smooth muscle cells (SMC) (Schonherr et al., 1997; Cardoso et al., 2010). Several studies have reported proteins such as platelet-derived growth factor (PDGF) and transforming growth factor- β 1 (TFG- β 1) increase versican synthesis in arterial smooth muscle cells (ASMCs) (Schonherr et al., 1991 & 1997). It was demonstrated that increases in versican and the associated protein hyaluronan in response to PDGF and TFG- β 1 cause increases in the pericellular matrix of the cells and expansion of the ECM that is required for the proliferation and migration of these cells (Evanko et al., 1999). In addition it was shown that proliferation of ASMCs treated with PDGF is blocked by inhibition of the formation of versican-hyaluronan complex which serves as an important mechanism for controlling cell shape and cell division (Evanko et al., 2001).

Other studies have suggested a role for versican in cell proliferation through its two epidermal growth factor sequences in the G3 domain of the molecule (Zhang et al., 1998; Wu et al., 2001). These studies showed that expression of the G3 domain of versican promotes proliferation in NIH3T3 fibroblasts cells whereas this effect can be inhibited by removing EGF motifs in the versican G3 domain (Zhang et al., 1998). These results suggested that the EGF-like motifs in the versican G3 domain may promote cell proliferation through direct or indirect interaction with the EGF receptor (EGFR) (Zhang et al., 1999).

1.3.4.3 Cell migration

Controversial studies have demonstrated that versican is widely expressed at both mRNA and protein level in neural crest pathways and influences neural cell migration (Perissinotto et al., 2000) whereas a number of other studies have shown that versican prevents migration of these cells (Landolt et al., 1995; Perris et al., 1996). These contradictory findings are believed to be due to different versican isoforms which differ in the core-middle region (Dutt et al., 2006). Some studies have investigated the role of versican in the nervous system and axonal outgrowth (Fidler et al., 1999; Jones et al., 2003; Tang et al., 2003). These studies showed that chondroitin sulphate (CS) chains of versican isotype V2 are involved in inhibiting axonal outgrowth and migration of the mature nervous system. The role of versican in axonal migration was investigated by Asher et al., (2002) who showed up-regulation of versican following central nervous system (CNS) injury and suggested these changes in versican regulation are associated with the failure of nerves to regenerate.

As mentioned earlier, the G3 domain of versican can interact with integrin β1 which is able to form clusters with EGF receptors (Yamada and Even-Ram, 2002). Growing evidence indicates that interaction of integrins with EGF receptors induces downstream signal to extracellular regulated kinase (ERK) which is crucial in regulating a range of cell activities, such as migration (Adelsman et al., 1999; Cabodi et al., 2004;

Wu et al., 2004). Also, the role of versican in the migration of embryonic cells in the development of the heart has been studied (Henderson and Copp, 1998; Kern et al., 2006). It has been reported that versican mRNA and protein is strongly expressed during the development of mouse heart suggesting a key role for versican in cardiac development (Henderson and Copp, 1998).

Furthermore other studies have reported that the G3 domain of versican directly interacts with fibronectin, another extracellular matrix glycoprotein (Yamagata et al., 1986) which showed that formation of a complex of versican G3 domain and fibronectin with VEGF can enhance endothelial cell migration which this process was reversed by removal of the complex (Zheng et al., 2004). This study and other investigation by Wijelath et al., (2002) indicated that expression of versican G3 enhanced brain tumour growth, suggesting the role of versican G3 fragment on promoting angiogenesis and tumour growth that it suggests targeting versican G3 fragments may help to develop a new approach for anticancer and antiangiogenic therapies.

1.3.4.4 Extracellular matrix assembly

Versican interacts with different ECM molecules and has been reported to have an important role during ECM assembly (Wu et al., 2005). Possibly the best known is a specific interaction between the G1 domain of versican and hyaluronan (LeBaron et al., 1992). Hyaluronan is a large polysaccharide in the ECM and is able to create a lattice structure which may regulate cell adhesion and migration (Lee and Spicer,

2000). Versican binds hyaluronan and this binding requires the double tandem repeat present in the G1 domain of versican (Mukaratirwa et al., 2004; Suwiwat et al., 2004).

1.3.5 Versican regulation

1.3.5.1 Signal transduction pathways

The signalling pathways which modulate versican synthesis are not fully understood although studies on the intracellular pathways have reported that PDGF-stimulated versican expression in arterial smooth muscle cells (SMC) is regulated by endogenous tyrosine kinase activity of the PDGF receptor which up-regulates versican synthesis at both mRNA and protein levels in vascular smooth muscle cells (Schonherr et al., 1997; Syrokou et al., 1999). In addition, another study has suggested the role of protein kinase C (PKC) and ERK in the PDGF stimulated versican gene expression in non-human ASMC (Cardoso et al., 2010).

An interesting study by Rahmani et al., (2005) demonstrated the role of the PI3-K/ PKB (Protein kinase B) pathway in versican expression in SMC. They suggested that phosphorylation and inactivation of glycogen synthase kinase-3 β (GSK-3 β), a downstream effector of PI3K/PKB (Harwood, 2001), leads to activation and nuclear accumulation of β -catenin which binds to the TCF/LEF transcription factors in the versican promoter and then increases versican transcription in SMC (Fig 1.7) (Rahmani et al., 2005). Further investigation by this group using a specific inhibitor of the PI3-Kinase pathway, LY294002, which inhibited the activation of downstream PKB, showed significant reduction of versican promoter reporter activity and also versican mRNA expression in SMC. A similar mechanism has been observed for other genes such as

MMP-7 (Deguchi et al., 2009) and VEGF (Zhang et al., 2001), which have also been shown to be targeted by PI3-K/PKB and up-regulated by formation of a β -catenin TCF/LEF complex.



<u>Figure 1.7</u>: Schematic model of versican promoter regulation via PI3/PKB signalling and β -catenin TCF transcription factor complex suggested by Rahmani et al., 2005. Activation of PI3K signalling by growth factors leads to phosphorylation and inactivation of GSK-3 β which results in β -catenin cytoplasmic accumulation and subsequent translocation to the nucleus. Nuclear accumulation of β -catenin leads to complex formation with TCF/LEF transcription factors and transactivation of TCF/LEF target genes. Specific inhibitors of the PI3K signalling pathway such as LY294002 and wortmannin lead to activation of GSK-3 β and subsequently to β -catenin degradation.

1.3.5.2 Transcription factors

Analysis of the versican promoter by Rahmani et al., (2005) revealed two putative binding sites for TCF/LEF transcriptions factors at positions -546 and -492 bp relative to the transcription start site (Fig 1.6). They showed that site-directed mutagenesis of the TCF sites in the versican promoter markedly diminished reporter luciferase activity in SMC. Furthermore, electrophoretic mobility shift assay (EMSA) and supershift assays revealed that the β -catenin / TCF transcription factor complex is essential for versican expression in SMC (Rahmani et al., 2005).

A recent study by Domenzain et al., (2009) identified several other transcriptional regulatory elements including AP1, SP1 and AP2 on a 620-bp (-618/+2 relative to the transcriptional start site) in a proximal versican promoter reporter construct (Domenzain et al., 2009). This study demonstrated that mutagenesis of the AP-1 site at position -36bp completely abolished versican promoter activity in human melanoma cell lines. Also further investigation by EMSA confirmed the importance of the AP-1 binding site for versican promoter transcription in these cell lines. In addition, versican promoter activity in a TCF/LEF mutated construct was reduced by half, suggesting that versican expression is also up-regulated via the β -catenin/TCF pathway in human melanoma cell lines (Domenzain et al., 2009).

Yoon et al., (2002) investigated the role of the transcription factor p53 in versican gene expression in a broad range of human carcinoma cell lines. p53 is a transcription factor involved in important cellular processes such as cell cycle checkpoint regulation, DNA damage and apoptosis (Harris, 1996; Bossi and Sacchi, 2007). Oligonucleotide-array gene expression analysis of human carcinoma cell lines by Yoon et al., (2002) demonstrated high expression of the versican gene in wild type p53 (p53 +/+) cells but lower expression in p53 -/- cells, suggesting that versican is a direct target of p53. Further investigation using wild type p53 over-expression in p53-null cells transfected with versican promoter reporter constructs showed 200-fold increases in luciferase activity in comparison with the control plasmid. EMSA and super-shift assays confirmed the interaction of p53 protein and the versican p53 binding site (Yoon et al., 2002).

1.4 Aims of the project

As previously outlined, several studies have highlighted the role of versican in angiogenesis and tumour growth (Zheng et al., 2004) where macrophages and hypoxia are present. In addition, another study has shown macrophage activation by versican over-expression through pattern recognition receptor engagement leading to the release of pro-inflammatory cytokines (Kim et al., 2009). Therefore, taking into account the above findings linking disease outcomes with macrophages, hypoxia, and versican expression, I aimed in this study to:

 Identify the principle mononuclear cells which show hypoxic up-regulation of versican mRNA by analysing adherence-purified HMDM, CD14⁺ purified monocyte/macrophages and lymphocytes.

Investigate the effect of hypoxia on versican protein expression in PBMC after either
24 hours or 5 days hypoxia.

• Study whether the observed up-regulation of versican mRNA by hypoxia is due to increased transcription or increased versican mRNA stability.

• Elucidate in detail the mechanisms responsible for the up-regulation of versican mRNA by hypoxia in primary HMDM using promoter reporter constructs and transcription factor over-expression.

In addition, unpublished data by Staples and Burke et al., (Bernard Burke, personal communication) using Real-Time RT-PCR showed that versican is up-regulated over 600-fold at the mRNA level in 5 day old hypoxic human peripheral blood mononuclear cells (PBMC) (Fig 1.8). This preliminarily investigation set me out to establish how the versican expression is regulated by hypoxia in human primary macrophages.



<u>Figure 1.8</u>: Effect of hypoxia (0.2% O₂) on versican mRNA expression in peripheral blood mononuclear cells (PBMC). PBMCs were quantified by Real-Time RT-PCR to analysis the effect of 24 hours and 5 days hypoxia on versican mRNA expression (See section 3.3 for RNA isolation and Real-Time RT-PCR experiment). 2 x10⁶ Cells per well were exposed to normoxia for 5 days (5 d N) or hypoxia for 5 days (5 d H) or 4 days in normoxia followed by 1 day in hypoxia (4 d N, 1 d H; See section 3.2.1). Data were normalised to β2M mRNA levels. Data from 8 independent experiments using different donors are expressed as means ± SEM. The normoxic value in each experiment was assigned an arbitrary value of 1. Data were further analysed for significant induction using two-tailed, paired t-tests. ** p < 0.01,* = p <0.05. Staples, Burke et al., unpublished data.

Chapter 2: Materials

2.1 Reagents

2.1.1 Cell culture

Reagent	Catalogue number	Company
Actinomycin D	A-9415	Sigma-Aldrich, Dorset, UK
Cobalt chloride (CoCl ₂)	C-8661	Sigma-Aldrich, Dorset, UK
Desferrioxamine (DFO)	D-9533	Sigma-Aldrich, Dorset, UK
Dimethyl sulphoxide (DMSO)	D-5879	Sigma-Aldrich, Dorset, UK
Ferric citrate	F-3388	Sigma-Aldrich, Dorset, UK
Foetal Bovine Serum (FBS)	S0115	BiochRom AG, Berlin, Germany
Ficoll-Paque [™] Plus	17-1440-03	GE healthcare, Buckinghamshire, UK
Hank's Balanced Salt Solution	H-6648	Sigma-Aldrich, Dorset, UK
Heparin (5000 U/ml)	Kindly provided by Glenfield Hospital, Leicester, UK	Leo laboratories limited, UK
Human AB serum	S4190-100	BioSera, Ringmer, UK
Iscove's Modified Dulbecco's medium	13390	Sigma-Aldrich, Dorset, UK
L-glutamine 200 mM	G-7513	Sigma-Aldrich, Dorset, UK
LY294002 (PI3 kinase inhibitor)	9901	New England Biolabs, Hitchin, UK
Non-essential amino acids (NEAA) 100X	M-7145	Sigma-Aldrich, Dorset, UK
Penicillin-Streptomycin 100 mM	P-0781	Sigma-Aldrich, Dorset, UK
Salmonella abortus equii lipopolysaccharide (SAE LPS)	581-009-L002	Alexis, San Diego, USA
Salmonella minnesota lipopolysaccharide (SMI LPS)	L-6261	Sigma-Aldrich, Dorset, UK

Trypan blue	6146	Sigma-Aldrich, Dorset, UK
VLE RPMI 1640 medium	F1415	BiochRom AG, Berlin,
(Very Low Endotoxin)		Germany
Wortmannin	681675	EMD Biosciences,
		Darmstadt, Germany

Table 2.1: List of reagents used in the cell culture

2.1.2 Reverse transcription and Real-Time PCR

Reagent	Catalogue number	Company
AMV Reverse Transcriptase	M510A	Promega Corporation
Chloroform	C-2432	Sigma-Aldrich, Dorset, UK
Deoxyadenosine triphosphate (dATP)	U120D	Promega Corporation, Southampton, UK
Deoxycytidine triphosphate (dCTP)	U122D	Promega Corporation Southampton, UK
Deoxyguanosine triphosphate (dGTP)	U121D	Promega Corporation Southampton, UK
Diethyl pyrocarbonate (DEPC)	D-5758	Sigma-Aldrich, Dorset, UK
Ethanol	N/A	<i>Chemistry department,</i> Leicester University <i>, UK</i>
Random Hexanucleotides	N/A	Eurofins MWG Operon, Ebersberg, Germany
Isopropanol	P/7500/15	Fisher, Loughborough, UK
RNasin [®] Ribonuclease Inhibitor	N251B	Promega Corporation Southampton, UK
SYBR® Green JumpStart™ Taq ReadyMix™	S-1816	Sigma-Aldrich, Dorset, UK
Thymidine triphosphate (dTTP)	U123D	Promega Corporation Southampton, UK
TRI reagent	T-9424	Sigma-Aldrich, Dorset, UK

Table 2.2: List of reagents used in RNA isolation, reverse transcription and RT-PCR

2.1.3 Molecular cloning

Reagent	Catalogue number	Company
Agarose	A5093	Sigma-Aldrich, Dorset, UK
Ampicillin	A9518	Sigma-Aldrich, Dorset, UK
Boric acid	100160	Merck, Darmstadt, Germany
Bovine Serum Albumin (BSA)	A-2153	Sigma-Aldrich, Dorset, UK
Bovine Serum Albumin (BSA) 100X	R396D	Promega Corporation Southampton, UK
DNA molecular weight marker (100 bp ladder)	N0467S	New England Biolabs, Hitchin, UK
DNA molecular weight marker (1 Kb ladder)	N3232S	New England Biolabs, Hitchin, UK
DNA molecular weight marker (HyperLadder™ V)	33031	Bioline, London, UK
Ethidium Bromide (EtBr)	46065	Sigma-Aldrich, Dorset, UK
Pfu DNA polymerase	EP0571	Fermentas, York, UK
REDAccuTaq [®] LA DNA Polymerase	D-1313	Sigma-Aldrich, Dorset, UK
<i>Sfil</i> Restriction Enzyme	R0123S	New England BioLabs, Hitchin, UK
T4 DNA ligase	600011	Stratagene, Stockport, UK
Xylene cyanol	X4126	Sigma-Aldrich, Dorset, UK

Table 2.3: List of reagents used in general molecular cloning

2.1.4 Nuclear extraction

Reagent	Catalogue number	Company
Aprotinin	A-1153	Sigma Aldrich, Dorset, UK
Disodium hydrogen	1.06580.5000	Merck, Darmstadt,
phosphate (Na ₂ HPO ₄)		Germany
Dithiothreitol (DTT)	D-9163	Sigma-Aldrich, Dorset, UK
Ethylenediaminetetra-acetic acid (EDTA) 0.5 M, pH 8.0	E-7889	Sigma-Aldrich, Dorset, UK
Ethylene glycol tetra-acetic acid (EGTA)	E-8145	Sigma-Aldrich, Dorset, UK
Glycerol	G-7757	Sigma-Aldrich, Dorset, UK
Leupeptin	L-2023	Sigma-Aldrich, Dorset, UK
N-(2- Hydroxyethyl)piperazine-N'- (2-ethanesulfonic acid) (HEPES)	H-4034	Sigma-Aldrich, Dorset, UK
Nonidet NP40 (Igepal)	I-3021	Sigma-Aldrich, Dorset, UK
Pepstatin A	P-4265	Sigma-Aldrich, Dorset, UK
Phenylmethylsulfonyl fluoride (PMSF)	P-7626	Sigma-Aldrich, Dorset, UK
Potassium chloride (KCl)	P/4280/60	Fisher, Loughborough, UK
Potassium dihydrogen phosphate (KH ₂ PO ₄)	A323473	Merck, Darmstadt, Germany
Sodium chloride (NaCl)	S/3160/63	Fisher, Loughborough, UK
Sodium vanadate	590088	Sigma-Aldrich, Dorset, UK

Table 2.4: List of reagents used for nuclear protein extraction

2.1.5 Western blotting

Reagent	Catalogue number	Company
Acrylamide/bisacrylamide	161-0158	Bio-Rad Laboratories,
		Hertfordshire, UK
Ammonium persulphate	A-3678	Sigma-Aldrich, Dorset, UK
(APS)		
Bromophenol blue	8122	Merck, Darmstadt,
		Germany
ECL advance [™] western	RPN2135	GE healthcare,
blotting detection kit		Buckinghamshire, UK
Glycine	G-8898	Sigma-Aldrich, Dorset, UK
Methanol	M/4000/17	Fisher, Loughborough, UK
Pre-Stained High Molecular	LC5699	Invitrogen Life Technology,
Weight Protein Standard		Paisley, UK
Sodium Dodecyl Sulphate	L-5750	Sigma-Aldrich, Dorset, UK
(SDS)		
Tetramethylethylenediamine	T-9281	Sigma-Aldrich, Dorset, UK
(TEMED)		
Trizma [™] (Tris base)	T-1503	Sigma-Aldrich, Dorset, UK
Tween-20	P-5927	Sigma-Aldrich, Dorset, UK

Table 2.5: List of reagents used in western blotting

2.1.6 Flow cytometry

Reagent	Catalogue number	Company
Brefeldin A	B-6542	Sigma-Aldrich, Dorset, UK
Paraformaldehyde (formalin)	F-8775	Sigma-Aldrich, Dorset, UK
Saponin	S-7900	Sigma-Aldrich, Dorset, UK

Table 2.6: List of reagents used in flow cytometry (for antibodies used see table 2.14)

2.1.7 Reagent Kits

Reagent Kit	Catalogue number	Company
Dual luciferase [®] Reporter Assay System [*]	E1910	Promega Corporation Southampton, UK
EndoFree [™] Plasmid Maxi Kit ^{**}	12362	QIAGEN, West Sussex, UK
JetPEI [™] DNA transfection reagent	101-40N	Polyplus-transfection Inc, New York, USA
Library Efficiency [®] DH5α TM competent cells ^{***}	18263-012	Invitrogen Life Technology, Paisley, UK
Luciferase assay system ****	E1501	Promega Corporation Southampton, UK
Pierce 660 nm Protein Assay	22660	Thermo Scientific, Rockford,USA
QIAquick Gel Extraction kit	28704	QIAGEN, West Sussex, UK

Table 2.7: List of reagent kits used in the project

* Dual luciferase® Reporter Assay System kit contains 5x Lysis buffer, Luciferase Assay

Reagent II and Stop & Glo reagent

** EndoFree[™] Plasmid Maxi Kit contains TE buffer (10mM Tris-HCl [pH7.5] and 1mM

EDTA)

 *** Library Efficiency® DH5 α^{TM} competent cells Kits contains control pUC19 DNA and

S.O.C medium

**** Luciferase assay system kit contains 5x Lysis buffer and Luciferase Assay Reagent

2.2 Consumables

Material	Catalogue number	Company
0.5 ml Centrifuge tube	311-02-051	Axygen Bioscience, California , USA
1.5 ml Centrifuge tube	311-08-051	Axygen Bioscience, California , USA
15 ml Centrifuge tube	430791	Corning Incorporated, Tampaulipas, Mexico
50 ml Centrifuge tube	21008-178	VWR International, West Chester, USA
5 ml Falcon tube (polystyrene)	352008	Becton Dickinson, New Jersey, USA
14 ml Falcon tube	352059	Becton Dickinson, New Jersey, USA
Hybond-P PVDF membrane	RPN303F	GE healthcare, Buckinghamshire, UK
LightCycler [®] Capillaries (20 µl)	04929292001	Roche Diagnostics, Burgess Hill, UK
MACS MS column	130-042-201	Miltenyi Biotec, Bergisch Gladbach, Germany
Multidish 6-Well Nunclon plates (adherent plates)	S3127XC	NUNC A/S, Roskilde, Denmark
20 ml Syringe	SZR-205-110L	Fisher, Loughborough, UK
50 ml Syringe	SZR-205-160T	Fisher, Loughborough, UK
75 cm ² Tissue culture flasks	658170	Greiner Bio-One, Stonehouse, UK
75 cm ² Tissue culture flasks filter cap	658175	Greiner Bio-One, Stonehouse, UK
VarioMACS separator	130-090	Miltenyi Biotec, Bergisch Gladbach, Germany
6-Well Costar ultra low attachment plates	CC227	Appleton Woods, Birmingham, UK
96-well cell culture cluster	3596	Corning Incorporated, New York, USA

Table 2.8: List of materials used in the project

2.3 Media

2.3.1 Media for cell culture

Medium	Reagents	Recipe
COMPLETE Iscove's Dulbecco's Modified Medium	Iscove's Dulbecco's Modified Medium 2 mM L-glutamine 200 U/ml Penicillin 200 µg/ml Streptomycin 2.5 % Human AB serum	500 ml Iscove's Dulbecco's Modified Medium 5 ml 200 mM L-glutamine 10 ml 10000 U/ml Penicillin / 10000 μg/ml Streptomycin 12.5 ml Human AB serum store at 4°C
COMPLETE RPMI 1640 medium	VLE RPMI1640 medium 2 mM L-glutamine 200 U/ml Penicillin 200 µg/ml Streptomycin 10x Non-essential amino acids 10 % FBS	500 ml VLE RPMI1640 medium 5 ml 200 mM L-glutamine 10 ml 10000 U/ml Penicillin / 10000 μg/ml Streptomycin 5 ml 100x NEAA the medium was passed through a Gambro U-2000 ultrafiltration column to eliminate potential contaminants such as endotoxin and other bacterial components then add 50 ml FBS and store at 4°C

Table 2.9: Media used for cell culture

2.3.2 Media for bacteria

Medium	Reagent	Recipe
LB-medium with	10 g/l NaCl	10 g Luria broth powder
Ampicillin	5 g/l Yeast extract	400 ml dH ₂ O
	10 g/l Tryptone	autoclave and let cool to
	100 μg/ml Ampicillin	55°C
		add 400 μl 100 mg/ml
		Ampicillin
		store at 4°C
LB-Agar medium with	10 g/l NaCl	10 g Luria broth
Ampicillin [*]	5 g/l Yeast extract	6 g Agar
	10 g/l Tryptone	400 ml dH ₂ O
	15 g/l Agar	autoclave and let cool to
	100 μg/ml Ampicillin	55°C
		add 400 µl 100 mg/ml
		Ampicillin
		pour into plates
		store at 4°C

Table 2.10: Media used for bacterial cultures

^{*}LB-medium and LB-agar were prepared by the medium kitchen, Department of Infection, Immunity and Inflammation, University of Leicester

2.4 Primers

2.4.1	Primers	for	Real-	Time	PCR
-------	----------------	-----	-------	------	-----

β₂Μ	5' Forward	5'-GGCTATCCAGCGTACTCCAAAG-3'
β₂M	3' reverse	5'-CAACTTCAATGTCGGATGGATG-3'
Versican	5' Forward	5'-ACAAGCATCCTGTCTCACG-3'
Versican	3' reverse	5'-TGAAACCATCTTTGCAGTGG-3'
VEGF	5' Forward	5'-CAGCGCAGCTACTGCCATCCAATCGAGA-3'
VEGF	3' reverse	5'-GCTTGTCACATCTGCAAGTACGTTCGTTTA-3'
GLUT-1	5' Forward	5'-CAACTGGACCTCAAATTTCATTGTGGG-3'
GLUT-1	3' reverse	5'-CGGGTGTCTTATCACTTTGGCTGG-3'

<u>**Table 2.11:**</u> List of primers used for Real Time-PCR (β_2 M: Beta-2 Microglobulin, VEGF: - Vascular Endothelial Growth Factor, GLUT-1: Glucose Transporter -1)

Note: Versican primer sequences were derived from NCBI (National Centre for Biotechnology Information) nucleotide bank accession no. NM_004385. Forward primer is from nucleotides 9879 - 9897; reverse primer is from nucleotides 10254 – 10273. β 2M, VEGF and GLUT-1 primer sequences were obtained from published work (Burke et al., 2003; Gao et al., 2005). All primers (Eurofins MWG Operon, Ebersberg, Germany) were diluted with autoclaved dH₂O to make 100 μ M laboratory stocks which were then further diluted 1:20 with autoclaved dH₂O to produce 5 μ M working stocks.

2.4.2 Primers for PCR cloning

Primers	Sequences
-3.956kb versican 5'	5'-aattGGCCTAACTGGCCATGCTCTGCTCCAGGCAGCCTC-3'
-646bp versican 5'	5'-aattGGCCTAACTGGCCGTGCAGTGGGGGAATTCTTAC-3'
-296bp versican 5'	5'-aattGGCCTAACTGGCCCAGAGCGATGTTACTGAGTG-3'
-56bp versican 5'	5'-aattGGCCTAACTGGCCCTGAGTCTCCTCCGGCTC-3'
-26bp versican 5'	5 ⁻ -aatt <mark>GGCCTAACTGGCC</mark> GTGATATAATGATGATGGGTGT-3'
+54bp versican 3'	5'-aattGGCCGCCGAGGCCGAGTTCTTCCCCAGAAAGGC-3'
+104bp versican 3'	5'-aattGGCCGCCGAGGCCGCTCCTGTCCACAACACCTA-3'
+154bp versican 3'	5'-aattGGCCGCCGAGGCCGTGCTGGAAATGCGGGAGG-3'
+184bp versican 3'	5'-aattGGCCGCCGAGGCCCCCGGGGAAGGATGCGGAGG-3'
Random 5'	5'- TGGCCCTAGCTAGCTAGCTAGCTAGCTAGCGGCCTCGG-3'
Random 3'	5'-AGGCCGCTAGCTAGCTAGCTAGCTAGCTAGGGCCAGTT-3'

Table 2.12: List of primers used for PCR cloning

Note: Versican promoter sequences were generated by PCR from human genomic DNA (See section 3.3) with the appropriate sets of primers designed using Primer3 web-based software (available at: <u>http://frodo.wi.mit.edu/primer3/input.htm</u>) based on the published versican promoter sequence (Naso et al., 1994, accession number U15963). The shaded sequence is the *Sfi*l restriction site. The sequence "aatt" is a random sequence added to both ends of the clones in order to facilitate restriction endonuclease activity.

2.5 Recipes

All solutions and buffers were prepared using nanopure deionised water (dH $_2$ O).

Where stated, buffers and solutions were autoclaved (20 minutes, 121°C).

Reagents	Concentration	Recipe
X % Agarose gel	X % (w/v) Agarose 1x TBE 0.25 μg/ml EtBr	X g Agarose 100 ml 1x TBE 2.5 μl 10 mg/ml EtBr bring to the boil in a microwave oven, let cool and pour into the gel tray
Ampicillin stock solution	100 mg/ml Ampicillin	1 g Ampicillin add to 10 ml autoclaved dH ₂ O, filter sterilise and store aliquots at -20°C
2 mg/ml Aprotinin	2 mg/ml Aprotinin	5 mg in 2.5 ml dH ₂ O Store in -20°C
10% APS	10% (w/v) Ammonium persulphate	10 g APS in 100 ml dH ₂ O
1 mg/ml BSA	1 mg/ml BSA 1x luciferase lysis buffer	10 mg BSA 10 ml 1x luciferase lysis buffer
50 μg/ml BSA	50 μg/ml BSA 1x luciferase lysis buffer	100 μl 1mg/ml BSA 1.9 ml 1x luciferase lysis buffer
100 μg/ml BSA	100 μg/ml BSA 1x luciferase lysis buffer	100 μl 1mg/ml BSA 900 μl 1x luciferase lysis buffer
250 μg/ml BSA	250 μg/ml BSA 1x luciferase lysis buffer	100 μl 1mg/ml BSA 300 μl 1x luciferase lysis buffer
500 μg/ml BSA	500 μg/ml BSA 1x luciferase lysis buffer	100 μl 1mg/ml BSA 100 μl 1x luciferase lysis buffer
750 μg/ml BSA	750 μg/ml BSA 1x luciferase lysis buffer	100 μl 1mg/ml BSA 30 μl 1x luciferase lysis buffer
Degassed PBE buffer (D- PBE)	1x PBS 1% FBS 2 mM EDTA	50 ml 1x filtered PBS 500 μl FBS 200 μl 0.5 M EDTA place under vacuum for 45 min for degassing

DEPC-dH ₂ O	0.1 % Diethyl pyrocarbonate	1 litre dH ₂ O add 1 ml DEPC and incubate at RT overnight, then
75% DEPC-Ethanol	75% (v/v) Ethanol	37.5 ml 100% Ethanol 12.5 ml 0.1% DEPC-dH ₂ O store at -20°C
10 mM dNTP mix	10 mM dATP 10 mM dTTP 10 mM dCTP 10 mM dGTP	100 μl 100 mM dATP 100 μl 100 mM dTTP 100 μl 100 mM dCTP 100 μl 100 mM dGTP 600 μl DEPC-dH ₂ O store aliquots at -20°C
1 M DTT	1 M Dithiothreitol	0.54 g DTT in 3.5 ml dH ₂ O filter sterilise and store at - 20°C
DNA 6x gel loading buffer (10 ml)	0.25% Bromophenol blue 0.25% Xylene cyanol 30% Glycerol	7.0 ml dH ₂ O 250 mg Bromophenol blue 250 mg Xylene cyanol 3.0 ml Glycerol store aliquot at 4°C
10 mg/ml EtBr	10 mg/ml EtBr	g EtBr adjust to 10 ml with dH ₂ O and store in the dark at RT
0.1M EGTA pH 8.0	380 g/l EGTA	1.9 g Adjust PH to 8.0 adjust to 50 ml with dH ₂ O and autoclave
1% Formalin	1x PBS/2% FBS 1% Formalin	100 μl formalin 9.9 ml 1x PBS/2% FBS
1M Hepes pH7.5	238.3 g/l Hepes	11.915 g adjust pH to 7.5 adjust to 50 ml with dH ₂ O and autoclave
1M KCl	74.56 g/l KCl	3.72 g in 50 ml dH ₂ O and autoclave
2 mg/ml Leupeptin	2 mg/ml Leupeptin	5 mg in 2.5 ml dH ₂ O Store in -20°C
1 M NaCl	58.44 g/l NaCl	2.92 g in 50 ml dH ₂ O and autoclave
Nuclear extract buffer A (50 ml)	10 mM Hepes pH7.5 0.1 mM EDTA pH 8.0 0.1 mM EGTA pH 8.0 10 mM KCl	48.94 ml dH ₂ O 500 μl 1 M Hepes pH7.5 10 μl 0.5 M EDTA pH 8.0 50 μl 0.1 M EGTA pH 8.0 500 μl KCl

AL 1		
Nuclear extract buffer B	20 mM Hepes pH 7.5	28.4 ml dH ₂ O
(50 ml)	1 mM EDTA pH 8.0	1 ml 1 M Hepes pH7.5
	1 mM EGTA pH 8.0	100 μl 0.5 M EDTA PH 8.0
	0.4 M NaCl	500 μl 0.1 M EGTA
		20 ml 1 M NaCl
1x PBS (Phosphate	137 mM NaCl	40 g NaCl
buffered saline)*	2.7 mM KCl	1 g KCl
	10 mM Na₂HPO₄	7.2 g Na₂HPO₄
	2 mM KH ₂ PO ₄	1.2 g KH ₂ PO ₄
		adjust to 5 litre with dH_2O
		autoclave
		passed through a Gambro
		11 2000 ultrafiltration
		column to oliminato
		potential contaminants
		such as endotoxin and other
		bacterial components
		store at 4°C
1x PBS/2% FBS	1x PBS	98 ml 1x PBS
	2% FBS	2 ml FBS
1 mg/ml Pepstatin A	1 mg/ml Pepstatin A	5 mg in 5 ml dH ₂ O
		Store in -20°C
0.2 M PMSF	174.19 g/l PMSF	0.35 g in 10 ml 100%
		Ethanol
		Store in 4°C
4% polyacrylamide gel	0 13 M Tris/HCl 6 8	3 ml dH ₂ O
(stacking gel)	0.25% SDS	340 ul ml 1 5 M Tris/HCl 6 8
	1%	
	Acrylamido /bisacrylamido	500 µl 20%
		Acrulamida /bicacrulamida
	0.125% AP3	
	0.25% TEIVIED	50 μι 10% APS
		10 µI TEMED
		After addition of APS and
		TEMED the gel was poured
		immediately and left at
		least 30 min at RT to
		polymerise
8% Polyacrylamide gel	0.4 M Tris/HCl 8.3	5.8 ml dH ₂ O
(resolving gel)	0.1% SDS	1.27 ml 3 M Tris/HCl 8.3
	8%	100 μl 10% SDS
	Acrylamide/bisacrylamide	2.7 ml 30%
	0.1% APS	Acrylamide/bisacrylamide
	0.3% TEMFD	100 μl 10% APS
		30 µl TEMED
		After addition of APS and
		IEMED the gel was poured
		immediately and left at least
		30 min at RT to polymerise

0.1% Saponin	1x PBS/2% FBS	0.1 g Saponin in 100 ml 1x
	1% Saponin	PBS/2% FBS
10% SDS	10% (w/v) sodium dodecyl sulfate	10 g SDS in 100 ml dH ₂ O
0.1 M Sodium vanadate	121.93 g/l Sodium vanadate	0.12 g in 10 ml dH ₂ O Store in 4°C
Solution A (fresh	Nuclear extract buffer A	493.25 μl Nuclear extract
prepared nuclear buffer	1 mM DTT	buffer A
A) (500 μl)	0.5 mM PMSF	0.5 μl 1 M DTT
	1mM sodium vanadate	1.25 μl 0.2 M PMSF
		5 μl 0.1 M Sodium vanadate
Solution B (fresh	Nuclear extract buffer B	482 μl Nuclear extract
prepared nuclear buffer	1 mM DTT	buffer B
A) (500 μl)	1 mM PMSF	0.5 μl 1 M DTT
	10 μg / ml Leupeptin	2.5 μl 0.2 M PMSF
	10 μg/ml Pepstatin	2.5 μl 2 mg/ml Leupeptin
	10 μg/ml Aprotinin	5 μl 1 mg/ml Pepstatin
	1 mM sodium vanadate	2.5 μl 2 mg/ml Aprotinin
		5 μl 0.1 M Sodium vanadate
10x TBE (Tris-borate-	0.9 M Tris base	108 g Tris base
EDTA)	0.9 M Boric acid	55 g Boric acid
	2 mM EDTA PH 8.0	40 ml 0.5 M EDTA
		adjust to 1 litre with dH ₂ O
		autoclave
1x TBE with EtBr	90 mM Tris-borate	100 ml 10x TBE
	0.5 mM EDTA PH 8.0	25 μl 10 mg/ml EtBr
	0.25 µg/ml EtBr	Adjust to 1 litre with dH_2O
10x TBS (Tris buffered	200 mM Tris base	24.2 g Tris base
saiine)	1.368 IVI Naci	80 g Naci
		adjust Philo 7.6
		adjust to 1 litre with dH_2O
1x TBS/Twoop 20		100 ml 10y TRS
1x 103/1ween-20	0.1% Tween-20	1 ml Tween-20
	0.170 Ween 20	$900 \text{ ml} \text{ dH}_2\text{O}$
1 5 M Tris/HCl	121 14 g/l Tris base	90 855 g Tris base
		adjust to pH 6.8 with HCl
		adjust to 500 ml with dH_2O
		and autoclave
3 M Tris/HCl	121.14 g/l Tris base	181.71 g Tris base
		adjust to pH 8.3 with HCl
		adjust to 500 ml with dH_2O
		and autoclave
Western blot blocking	5% Skimmed milk powder	5 g Skimmed milk powder
buffer	1x TBS	100 ml 1x TBS/Tween-20
	0.1% Tween-20	
Western blot gel loading	250 mM Tris/HCI 6.8	$3.4 \text{ ml} \text{dH}_2\text{O}$
--------------------------	-----------------------	--
buffer 5x	10% SDS	1.6 ml 1.5 M Tris/HCl 6.8
	0.5% Bromophenol blue	1 g SDS
	50% glycerol	0.05 g Bromophenol blue
	500 mM DTT	5 ml glycerol
		25 μ l of 1 M DTT added to
		25 μl 5x gel loading buffer
		before use
Western blot running	250 mM Tris	30.29 g Tris
Buffer 10x	2 M Glycine	144.13 g Glycine
	10% SDS	10 g SDS
		adjust to 1 litre with dH ₂ O
		and autoclave
		Dilute 1:10 with dH ₂ O to
		use
Western blot transfer	25 mM Tris	3.03 g Tris
buffer	200 mM Glycine	14.41 g Glycine
	20% Methanol	200 ml Methanol
		800 ml dH₂O

Table 2.13: List of recipes for all buffers and solutions

^{*}1x PBS was prepared by the media kitchen of the Department of Infection, Immunity

and Inflammation, University of Leicester

2.6 Antibodies

Reagent	Catalogue number	Company
Anti-human CD14	120-000-305	Miltenyi Biotec, Bergisch
Monoclonal anti- human versican antibody	MAB3054	R&D Systems, Abingdon, UK
Rabbit anti-Actin polyclonal antibody	A-2066	Sigma-Aldrich, Dorset, UK
Rabbit anti-β-Catenin polyclonal antibody	C-2206	Sigma-Aldrich, Dorset, UK
Polyclonal anti-Rabbit Immunoglobulins/HRP	P0448	Dako, Cambridgeshire, UK
Anti-rat IgG conjugated to FITC	F-1763	Sigma-Aldrich, Dorset, UK
Rat IgG ₁ isotype control	MAB005	R&D Systems, Abingdon, UK

Table 2.14: List of antibodies used for immunoblotting and flow cytometry experiments

2.7 Reporter plasmids

2.7.1 pGL4.10 [luc2] Luciferase reporter vector (#E6651, Promega,

Southampton, UK)

The pGL4 luciferase reporter vectors are reporter vectors optimized for expression in mammalian cells (Fig 2.1). Numerous configurations of pGL4 vectors are available, including those with the synthetic firefly luc2 (*Photinus pyralis*) and Renilla hRluc (*Renilla reniformis*) luciferase genes, which have been optimized for more efficient expression in mammalian cells. Furthermore, both the reporter genes and the vector backbone such as Ampicillin (Amp^r) gene and mammalian selectable marker genes have been engineered to reduce the number of consensus transcription factor binding

sites, reducing background and the risk of anomalous transcription. The multiple cloning region of the pGL4 vectors has been synthetically constructed and is based on the multiple cloning region of the pGL3 vectors (Figure 2.1 and 2.2). However, differences between the two multiple cloning regions exist. The pGL4 vector multiple cloning region includes the following restriction sites: Bgll, Sfil, Acc651, Kpnl, EcolCRI, Sacl, Nhel, Xhol, EcoRV, Bglll, and HindIII. The Bgll, Sfil and EcoRV restriction sites have been added to the pGL4 vector multiple cloning region and are not found in the pGL3 vector multiple cloning region. This increases the number of choices in choosing restriction sites during cloning. The *Mlul* and *Smal* restriction sites found in the pGL3 multiple cloning region have not been included in the pGL4 multiple cloning region. The two *Bgll/Sfil* restriction sites in pGL4 facilitate the moving the DNA of interest such as response elements, enhancers and promoters between vectors. Additionally, transfers between pGL4 vectors using either the Bgll or Sfil restriction enzymes retain the desired orientation of the DNA of interest due to the unique DNA recognition properties of Ball and Sfil. Also pGL4 vectors contain a SV40 late poly (A) signal which increases the level of luciferase expression due to polyadenylation signals (Fig 2.1). In addition to the features listed, all pGL4 vectors contain a synthetic poly(A) signal/transcriptional pause site which is located upstream of the multiple cloning region. This site is present to reduce the effects of spurious transcription on luciferase reporter gene expression (Fig 2.1). In this project the Sfil restriction site of pGL4 luciferase reporter vector was chosen for versican promoter cloning (See section 3.3).



Figure 2.1: Generic pGL4.10 [luc2] vector map



Figure 2.2: The multiple cloning region of the pGL4 vectors

2.7.2 pRL-TK vector (#E2241, Promega, Southampton, UK)

The pRL-TK vector serves as an internal control reporter construct which can be used in combination with any experimental reporter vector (e.g. pGL4) to co-transfect mammalian cells. The pRL-TK vector contains a cDNA encoding Renilla luciferase, originally cloned from the marine organism *Renilla reniformis* (sea pansy) which can be used to normalize the firefly luciferase activity from the targeting constructs. In this project, HMDM were transfected with versican promoter pGL4.10 [luc2] reporter vector and co-transfected with Renilla plasmid (pRL-TK) to normalise the transfection results.

2.8 Equipment

Equipment	Model	Company
Balance	BD202	Mettler Toledo, Greifensee, Switzerland
Centrifuges	5717R	Eppendorf, Hamburg, germany
Micro plate reader (ELISA	Model 680	Bio-Rad Laboratories, Hertfordshire,
reader)		UK
Incubators	Hera cell	Heraeus, Hanau, Germany
Ø	Galaxy R	New Brunswick Scientific / RS Biotech
LightCycler Systems for Real-Time PCR	Version 3	Roche, Burgess Hill, UK
Luminometer	Sirius	Berthold detection system,
		Pforzheim, Germany
Microscopes	H600	Hund, Wetzlar, Germany
	Wilovert S	Hund, Wetzlar, Germany
NanoDrop spectrophotometer	1000	NanoDrop products, Wilmington, USA
Oxygen analyser	Mini O ₂	Analox sensor technology, North Yorkshire, UK
pH-meter	MP225	Mettler Toledo, Greifensee, Germany
Shaker	Innova44	New Brunswick Scientific, Hertfordshire, UK
	Celloshaker	Glotec-Fischer, Reiskirchen, Germany
Thermocycler (PCR	TC-3000	Techne, Cambridge, UK
machine)	Progene 231- 103	Techne, Cambridge, UK
UV- visible	Cary 50 Bio	Varian, Santa Clara, USA
Spectrophotometer		
Western blot apparatus	Xcell II Minicell	Novex San Diego, USA

Table 2.15: List of equipment

Chapter 3: Methods

3.1 Cell culture

3.1.1 Isolation of peripheral blood mononuclear cells (PBMCs)

Blood samples from healthy donors were collected into a standard laboratory syringe which contained sufficient heparin to give a final concentration of 10 U/ml, to prevent blood clotting. The blood sample was diluted 1:1 with Hank's Balanced Salt Solution and then 30 ml of diluted blood was gently layered onto 15 ml of Ficoll-PaqueTM Plus in a 50 ml centrifuge tube, followed by 30 minutes centrifugation (minimum acceleration and brake off) at 400 g at room temperature (RT). Following centrifugation a thick concentrated PBMC layer, containing monocytes and lymphocytes, was visible at the interface between the original blood sample and the Ficoll-Paque Plus. The monocyte/lymphocyte layer was carefully transferred to a new 50 ml centrifuge tube and washed twice with Hank's Balanced Salt Solution and spun down at 400 g for 10 minutes at RT. Finally the PBMC were washed once with Iscove's Modified Dulbecco's medium and spun down at 400 g for 10 minutes at RT. Isolated cells were resuspended in 10ml of Iscove's Dulbecco's medium containing 2.5% human AB serum, 1% L-glutamine, and 1% penicillin and streptomycin (See tables 2.1, 2.8 and 2.9 for the reagents and materials) followed by cell counting using an improved Neubauer haemocytometer with a chamber depth of 0.1 mm (Staples et al., 2007) (See section 3.1.2).

Note: Iscove's Dulbecco's medium containing 2.5% human AB serum, 1% L-glutamine, and 1% penicillin and streptomycin is described as complete Iscove's medium in this project. Hank's Balanced Salt Solution and complete Iscove's medium were used at RT.

3.1.2 Determination of cell concentration

An aliquot of the cells was diluted 1:10 in hank's balanced salt solution (HBSS). 10 μ l of the diluted cells was transferred to an improved Neubauer haemocytometer with a chamber depth of 0.1 mm. 25 central squares of the grid pattern bounded by triple lines (1mm x 1mm) were counted and the cell concentration (cells/ml) calculated as follows:

Cells/ml = Cells counted x 10⁴ x dilution factor

- 10⁴ = Conversion factor of volume of counting chamber into ml
- Dilution factor=10

3.1.3. Generation of adherence-purified human monocyte- derived macrophages (HMDM)

The PBMC concentration was adjusted to 1×10^6 cell/ml in complete Iscove's medium. 6-well adherent Nunclon plates were seeded with 2ml of medium per well giving 2×10^6 cells per well and incubated at 37° C, 5% CO₂ for 2 hours. After 2 hours, the medium containing the non-adherent cells (mainly lymphocytes) was removed and replaced with 2ml of fresh complete Iscove's medium. Adherent monocytes were incubated at 37° C with 5% CO₂ and 20.9% O₂ for 5 days to allow differentiation into macrophages (Burke et al., 2003; Staples et al., 2011). Previous analysis of maturation markers (CD68) and morphology of monocyte-derived macrophages (MDMs) over a period of 3-5 days incubation concluded that this length was appropriate for the generation of macrophages (Staples et al., 2007) (See table 2.1 and 2.8 for the reagents and materials).

3.1.4 Magnetic cell sorting system (MACS)

The Miltenyi Biotec MACS technology separates different cell populations based on the expression of specific surface antigens on these cells using microbeads. MACS microbeads are super-paramagnetic particles that are coupled to monoclonal antibodies. CD14⁺ purified monocytes were prepared according to the manufacturer's instructions, as described below.

PBMC pellet was re-suspended in D-PBE buffer (See table 2.13), using 80 μ l per 10⁷ cells. Microbeads conjugated to monoclonal anti-human CD14 antibody (See table 2.14) were added (20 μ l per 10⁷ cells) followed by 15 minutes incubation at 4°C. After

incubation, cells were washed by adding 1.5 ml of D-PBE buffer per 10⁷ cells and spun down for 5 minutes at 300g. The cell pellet was re-suspended in 500 μ l of D-PBE buffer and then loaded into a MACS MS column, located in the magnetic field of the VarioMACS separator (See table 2.8). The MS column was washed twice with 500 µl of D-PBE buffer, which was allowed to run through into a discard pot. Cells pass through the column slowly so that CD14 positive cells bound to the magnetic beads attach to the column. The column was then washed twice with D-PBE to ensure that there were no CD14 negative cells trapped within the column. To retrieve the CD14 positive cells (monocytes) from the column, the MS column was removed from the VarioMACS unit and placed over a 15 ml sterile centrifuge tube. The CD14 expressing monocyte/microbeads complex was eluted by adding 1 ml of D-PBE to the column and forcefully applying a plunger. The collected CD14⁺ purified monocytes and CD14⁻ cells (mainly lymphocytes) were washed once with 1 ml of complete Iscove's medium and spun down at 300 g for 5 minutes at RT. Cells then were re-suspended in 1 ml of complete Iscove's medium and counted (See section 3.1.2). The CD14⁺ purified monocytes and CD14⁻ cells were seeded in 2ml complete Iscove's medium at a density of 1×10^{6} cells per well in 6-well adherent Nunclon plates and incubated at 37° C with 5% CO₂ and 20.9% O₂ for 5 days.

3.1.5 Culture of U937

U937 cells are a non-adherent human monocytic cell line derived from a histiocytic lymphoma. U937 cells were cultured in 75cm² tissue culture flasks using filtered RPMI 1640 medium containing 10% FCS, 1% L-glutamine, 1% penicillin and streptomycin and 1% non-essential amino-acid (See table 2.1 and 2.9). Cells were split once every 3 days

to keep the cells healthy; an aliquot of cells was diluted 1:1 with 1% filtered Trypan blue and then counted as described in section 3.1.2. Trypan blue stains the cytoplasm of apoptotic and dead cells blue but it is unable to penetrate the cell membrane of living cells. After determining the cell concentration, cells were diluted to 2.0 x 10^5 cells/ml and 10 ml transferred to a 75cm² flask. The flask was labelled with the date split and the cell concentration and returned to incubation at 37°C with 5% CO₂ and 20.9% O₂ (See table 2.1 and 2.8 for the reagents and materials).

Note: RPMI 1640 medium containing 10% FCS, 1% L-glutamine, 1% non-essential amino-acids and 1% penicillin/streptomycin has been described as complete RPMI1640 medium in this project. Complete RPMI1640 medium was used at RT.

3.2 Cell treatment

3.2.1 Normoxic & Hypoxic Incubation

Depending on the experiment, cells were either cultured under normal oxygen concentrations (normoxia) in a humidified atmosphere of 95% air (20.9% O₂), 5% CO₂, or under hypoxia in a humidified multi-gas oxygen control Galaxy R incubator at 0.2% O₂, 5% CO₂, 94.8% N₂ (See table 2.15). Oxygen levels indicated on the incubator display screens were verified using a separate oxygen analyser (Analox). Cells were incubated under normoxia or hypoxia for 18 hours, 24 hours or 5 days depending on the experiment.

3.2.2 Lipopolysaccharide (LPS) stimulation of adherence-purified HMDM The LPS used for cell stimulation was commercially isolated and purified from *Salmonella abortus equii* (SAE) or *Salmonella minnesota* (SMI) (See table 2.1), and were diluted with RPMI 1640 medium to make 10 ng/µl working stock solutions. After 5 days incubation under normal oxygen tensions, 2×10^6 adherent HMDM were treated with either SAE or SMI LPS at final a concentration of 100 ng/ml and incubated under normoxia or hypoxia for a further 18 hours (See section 3.2.1) prior to RNA isolation (See section 3.4).

3.2.3 Cobalt chloride (CoCl₂) stimulation of adherence-purified HMDM

CoCl₂ is one the common used agent which stabilises HIF-1 (Maxwell et al. 1999). The manufacturer's CoCl₂ (See table 2.1) was dissolved by dH₂O to prepare laboratory stock solution with a concentration of 30 mM. CoCl₂ was added to 2×10^6 adherent HMDM at final concentration of 300 μ M, for 18 hours in normoxia, prior to RNA isolation (See section 3.4).

3.2.4 Desferrioxamine (DFO) stimulation of adherence-purified HMDM

DFO is an iron chelating agent which primarily uses for HIF-1 stabilisation (Maxwell et al., 1999). The manufacturer's DFO (See table 2.1) was dissolved by dH₂O to prepare laboratory stock solution with a concentration of 20 mM. DFO was added to 2×10^{6} adherent HMDM at final concentration of 200 μ M, for 18 hours in normoxia, prior to RNA isolation (See section 3.4).

3.2.5 Ferric citrate stimulation of adherence-purified HMDM

The manufacturer's ferric citrate (See table 2.1) was dissolved by dH_2O to prepare laboratory stock solution with a concentration of 20 mM. For assessment of iron chelating activity of the DFO and its effect on versican gene expression, cells were treated simultaneously with 200 μ M DFO and ferric citrate at different concentrations (e.g. 100 μ M, 200 μ M, 400 μ M) to saturate the agent and reverse its activity.

3.2.6 Actinomycin D treatment of adherence-purified HMDM

Actinomycin D is an antineoplastic antibiotic that inhibits transcription by binding DNA at the transcription initiation complex and preventing elongation by RNA polymerase (Sobell, 1985). The manufacturer's actinomycin D (See table 2.1) was dissolved by sterile DMSO to generate a laboratory stock with a concentration of 1 mg/ml. For mRNA half-life analysis, actinomycin D at a final concentration of 10 μ g/ml was added to 2 × 10⁶ adherent HMDM after incubation for 5 days in normoxia. Cells were incubated under either normoxia or hypoxia (See section 3.2.1) for a further 2 hours, 8 hours, 12 hours, 24 hours and 36 hours prior to RNA isolation (See section 3.4).

3.2.7 PI3-Kinase inhibitor treatment of adherence-purified HMDM

LY294002 and wortmannin are shown to act in vivo as highly selective inhibitors of PI3-Kinase. The manufacturer's LY294002 and wortmannin (See table 2.1) were dissolved by sterile DMSO to generate laboratory stock solutions with a concentration of 10 mM and 0.2 mM respectively. After 5 days incubation under normal oxygen tensions, 2 × 10^{6} adherent HMDM were treated with LY290042 at final concentrations of 2 μ M and 5 μ M for 18 hours in normoxia and hypoxia (See section 3.2.1), prior to RNA isolation

(See section 3.4). Wortmannin at a final concentration of 200 nM was added to 2×10^6 adherent HMDM after incubation for 5 days in normoxia. Dilutions were prepared such that all wells received 2 µl of DMSO/inhibitor solution. The same volume of DMSO (2 µl) was added to cells as a carrier control. Cells then were incubated under normoxia or hypoxia (See section 3.2.1) for a further 18 hours prior to RNA isolation (See section 3.4).

3.3 General molecular cloning

3.3.1 Amplification of the versican promoter by polymerase chain reaction

Polymeras chain reaction (PCR) is a powerful method developed by Kary Mullis in the 1980s and occurs in three stages; denaturation of the template DNA, primer annealing, and primer extension. PCR is based on the ability of DNA polymerase to synthesize new strand of DNA complementary to the offered template strand by using two primers which are short pieces (usually around 20 base pairs) of synthetic DNA. Amplification of the targeting sequence on the template is achieved by cycles of reaction that elongate the primers according to the targeting sequence by DNA polymerase. This requirement makes it possible to delineate a specific region of template sequence that the researcher wants to amplify.

Forward and reverse primers which were used to amplify regions of the versican promoter for PCR cloning are shown in table 2.12. Below is the reaction mix and PCR set up to amplify different regions of the versican promoter using a thermocycler (See table 2.3 and 2.13 for reagents and materials). **Note:** in order to generate a 29-bp of random negative control, a reaction mix including 1 μ l of 10x buffer with MgSO₄ (1x), 1 μ l of forward (2.5 ng/ μ l) and 1 μ l of reverse primers (2.5 ng/ μ l) were made up to 10 μ l with Sterile dH₂O. The reaction then was placed in a thermocycler (PCR machine) for 1 minute at 95°C to anneal the primers. All PCR reaction mixes were prepared on ice.

Reagent	Volume	Concentration
REDAccuTaq [®] LA 10x Buffer	5 μΙ	1 x
dNTP (10mM)	2.5 μΙ	0.5 mM
Human genomic DNA (50pg/μl)	1 µl	50 pg
Forward primer (20mM)	1 µl	0.4 mM
Reverse primer (20mM)	1 µl	0.4 mM
Sterile dH ₂ O	37 μl	-
REDAccuTaq [®] LA DNA Polymerase	2.5 μl	0.05 U/µl
Final Volume	50 μl	-

Table 3.1: PCR reaction mix for 4.140-Kb (-3.956Kb/+184bp), 830-bp (-646bp/+184bp), 480-bp (-296bp/+184bp) and 240-bp (-56bp/+184bp) of versican promoter construct

Reagent	Volume	Concentration
10x Pfu buffer with MgSO ₄	5 μΙ	1 x
dNTP (10 mM)	1 μl	0.2 mM
Genomic DNA (50 pg/μl)	1 µl	50 pg
Forward primer (20 mM)	1 µl	0.4 mM
Reverse primer (20 mM)	1 µl	0.4 mM
Sterile dH ₂ O	50.5 μl	-
Pfu DNA polymerase	0.5 μl	0.025 U/μl
Final Volume	50 μl	-

Table 3.2: PCR reaction mix for 210-bp (-56bp/+154bp), 160-bp (-56bp/+104bp) and 110-bp (-56bp/+54bp), 210-bp (-26bp/+184bp), 180-bp (-26bp/+154bp), 130-bp (-26bp/+104bp), 80-bp (-26bp/+54bp) constructs of versican promoter

Segment	Temperature	Time	Cycles
Initial denaturation	96°C	30 sec	1
Denaturation	94°C	15 sec	
Annealing	68°C	30 sec	40
Extension	68°C	15 min	
Final Extension	68°C	20 min	1
Final Extension	4°C	-	-

Table 3.3: PCR set-up for amplification of 4.140-Kb (-3.956Kb/+184bp), 830-bp (-626bp/+184bp), 480-bp (-298bp/+184bp) and 240-bp (-56bp/+184bp) of versican promoter

Note: Annealing temperature and Extension time used for PCR amplification of 4.140-

Kb was 66° C and 5 min respectively.

Segment	Temperature	Time	Cycles
Initial denaturation	95 <i>°</i> C	1 min	1
Denaturation	94°C	30 sec	
Annealing	57°C	30 sec	30
Extension	70°C	30 sec	
Final Extension	70°C	10 min	1
Final Extension	4°C	-	-

Table 3.4: PCR set-up for amplification of 210-bp (-56bp/+154bp), 160-bp (-56bp/+104bp) and 110-bp (-56bp/+54bp) 210-bp (-26bp/+184bp), 180-bp (-26bp/+154bp), 130-bp (-26bp/+104bp), 80-bp (-26bp/+54bp) constructs of versican promoter

3.3.2 Agarose gel electrophoresis

Electrophoresis is used to separate DNA fragments by applying an electric field to a gel. DNA is negatively charged, independent of the pH of the solution, and migrates towards the positive pole of an electric field. The rate of migration of DNA in a gel depends on the concentration of the agarose, the pore size of the gel, the applied voltage, and conformation and length of the DNA molecules. In general bigger fragments migrate slower through the gel.

The sizes of PCR amplified fragments of the versican promoter were confirmed by running on an agarose gel and compared with a DNA marker (See table 2.3 and 2.13 for all the reagents and buffers). For agarose gel electrophoresis, a solution of agarose in 100ml of 1x TBE containing 0.25 μ g/ml ethidium bromide (See table 2.13 and 3.5)

was heated in a microwave oven until the agarose was dissolved. The solution was left to cool for a few minutes and then poured into a gel tray with a 1.0 mm comb in it and let the gel set. Ethidium bromide (EtBr) is a fluorescent dye which intercalates between stacked bases of double stranded DNA and allows visualisation of nucleic acids by exposure to UV light. Once the agarose solution was solidified and complete wells were formed, the comb was removed and the gel was placed in the electrophoresis tank which was filled with sufficient 1x TBE buffer (containing EtBr) to cover the gel (See table 2.13). The DNA samples of amplified fragments of versican promoter were diluted with 1:6 with 6x gel loading buffer and slowly loaded into the complete wells. The size of a fragment is estimated by comparison with a DNA molecular weight marker which contains fragment of known length. 250 ng of DNA molecular weight marker was loaded at the same time as the samples into a different well. A voltage of 100 V was applied for 2 hours. The gel run was terminated when the bromophenol blue migrated about 3/4 of the length of the gel. The gel was then examined under UV light and photographed.

Amplified fragment of	Agarose
versican promoter	concentration (%w/v)
4.140-Kb (-3.956Kb/+184bp)	0.6
830-bp (-646bp/+184bp)	1.5
480-bp (-296bp/+184bp)	1.5
240-bp (-56bp/+184bp)	1.5
210-bp (-56bp/+154bp)	1.5
160-bp (-56bp/+104bp)	2.0
110-bp (-56bp/+54bp)	2.0
210-bp (-26bp/+184bp)	1.5
180-bp (-26bp/+154bp)	2.0
130-bp (-26bp/+104bp)	2.0
80-bp (-26bp/+54bp)	2.0

Table 3.5: Electrophoretic separation of PCR amplified fragments of the versican promoter

3.3.3 Restriction digestion

Restriction endonucleases can be used for cleaving double-stranded DNA molecules at specific positions which are called restriction sites. The cleavage site is located within the recognition sequence and depending on the enzyme, DNA fragments with blunt ends or cohesive (sticky) ends are generated (See table 2.3).

In this project, for versican promoter cloning, the *Sfil* restriction site was chosen, which is cleaved by the *Sfil* restriction enzyme isolated from an *E.coli* strain that carries the *Sfil* gene from *Streptomyces fimbriatus*. The *Sfil* endonuclease cleaves DNA at the sequence 5'-GGCCNNNN \downarrow NGGCC-3', (where N is any base and \downarrow is the point of cleavage) and creates sticky ends. The *Sfil* restriction enzyme was used for the digestion of PCR amplified fragments of the versican promoter which carry *Sfil* restriction sites and also the pGL4.10 luciferase reporter plasmid (See section 2.7.1). Restriction digestion was carried according to the manufacturer's instructions (See table 3.6 and 3.7). The reaction was incubated at 50°C for 2 hours to allow digestion. After 2 hours, digested DNAs were separated on agarose gels as described in section 3.3.2.

Note: Concentrations of the amplified versican promoter PCR products were determined by comparing to known concentrations of DNA molecular weight marker separated on agarose gels (See section 3.3.2).

Reagents	Volume	Final concentration
<i>Sfi</i> l Restriction Enzyme (20 U/μl)	1 µl	1 U/ μl
DNA	11 µl	1-5 μg / 20 μl
10x <i>Sfi</i> l Buffer	2.0 μl	1 x
10x BSA	2.0 μl	1 x
Sterile dH ₂ O	4.0 μl	-
Final volume	20 µl	-

Table 3.6: Sfil digestion reaction mixture for PCR amplified fragments of versican promoter

Reagents	Volume	Final concentration
<i>Sfi</i> l Restriction Enzyme (20 U/μl)	1 µl	0.8 U/ μl
pGL4.10 vector (1 µg/µl)	3 μΙ	12 ng/ 25 μl
10x <i>Sfi</i> l Buffer	2.5 μl	1 x
10x BSA	2.5 μl	1 x
Sterile dH ₂ O	16 μl	-
Final volume	25 μl	-

Table 3.7: Sfil digestion reaction mixture for pGL4.10 luciferase reporter plasmid

3.3.4 QIAquick DNA Agarose Gel Extraction

The digested pGL4.10 vector and PCR amplified fragments of versican promoter sequences were extracted and purified from the agarose gel using the QIAquick Gel Extraction Kit (Qiagen) according to the manufacturer's instructions (See table 2.7).

First, 3 volume of QG Buffer was added to 1 volume of extracted gel slice (e.g. 300 μ l of QG Buffer to 100 mg of extracted gel) in a 1.5 ml sterile centrifuge tube and then incubated at 50°C for 10 minutes. When the gel slice was dissolved 1 volume of isopropanol was added to 1 volume of extracted gel slice (e.g. 100 μ l of isopropanol to 100 mg of extracted gel) to increase the yield of DNA fragments. Then the resulting mixture was transferred to a QIAquick spin column and centrifuged at 13000 g for 1 minute to bind DNA to the column. 750 μ l of PE Buffer containing ethanol was then added to the column and spun down at 13000 g for 1 minute to wash the DNA. Then, the column was placed over a new sterile 1.5 ml centrifuge tube and finally, to elute DNA, 30 μ l of EB Buffer was added to the centre of column and after 1 min in RT it was centrifuged at 13000 g for 1 minute. The purified DNA fragments were analysed by agarose gel electrophoresis (See section 3.3.2).

3.3.5 Ligation of DNA into plasmid vector

DNA ligation is the process of joining together two DNA molecule ends, most commonly to ligate an insert DNA molecule into a plasmid vector, ready for bacterial transformation. It involves creating a phosphodiester bond between the 3' hydroxyl of one nucleotide and the 5' phosphate of another. This reaction is usually catalyzed by a DNA ligase enzyme such as T4 DNA ligase which is the most commonly used DNA

ligase to ligate sticky or blunt ends. Typically, insert and plasmid vector DNA are individually cut to yield complementary ends, then both are added to a ligation reaction to be circularised by DNA ligase (See table 2.3 and 2.8 for the reagents and materials).

2.5-Kb of versican promoter cloned into pGL3-basic luciferase reporter plasmid was kindly provided by Dr Burke. In this project the digested pGL4.10 plasmid vector and digested fragments of versican promoter sequence were ligated at X:Y ratio according to the manufacturer's instructions (See table 3.9, Fig 2.1 and 2.2 for the cloning region of the pGL4.10 vector) where X is the amount of vector plasmid and Y the amount of insert DNA (See table 3.8). To prepare the ligation mixture, first PGL4.10 plasmid vector, versican DNA fragments and sterile dH₂O were transferred into a 0.5 ml sterile centrifuge tube and incubated in the Thermocycler (PCR machine) at 42°C for 2 minutes and then let it to cool down in RT. The rest of the reagents were then added and incubated at 4°C for 16 hours to complete ligation and followed by storage at - 20°C (See table 3.9).

Note: Concentrations of the digested fragment of versican promoter products and digested pGL4.10 plasmid vector were estimated by comparing to known concentrations of DNA molecular weight marker separated on agarose gels (See section 3.3.2).

Digested fragment of versican promoter	pGL4 vector X	Insert DNA Y	pGL4 vector (ng)	Insert DNA (ng)	pGL4 vector (µl)	Insert DNA (µl)
4.140-kb (-3.956kb/+184bp)	1	2	100	200	4	6.6
(30ng/µl)						
830-bp (-646bp/+184bp)	1	3	50	30	1	10
(3ng/µl)						
480-bp (-296bp/+184bp)	1	10	100	100	2	5.5
(18ng/µl)						
240-bp (-56bp/+184bp)	1	20	50	50	1	10
(5ng/µl)						
210-bp (-56bp/+154bp)	1	10	50	25	3.5	8.3
(3ng/µl)						
160-bp (-56bp/+104bp)	1	10	50	12.5	3.5	1.5
(8ng/µl)						
110-bp (-56bp/+54bp)	1	10	50	12.5	3.5	3.1
(4ng/μl)						
210-bp (-26bp/+184bp)	1	10	50	25	3.5	3.1
(8ng/µl)						
180-bp (-26bp/+154bp)	1	10	50	20	3.5	2.5
(8ng/µl)						
130-bp (-26bp/+104bp)	1	10	50	12.5	3.5	3.1
(4ng/μl)						
80-bp (-26bp/+54bp)	1	10	50	10	3.5	5
(2ng/µl)						

<u>**Table 3.8**</u>: Different ratios of vector DNA (pGL4.10 vector): insert DNA (digested versican promoter product). pGL4.10 vector was used at 25 ng/ μ l, 50 ng/ μ l and 14 ng/ μ l.

Reagents	Volume	Final concentration		
Digested pGL4.10	See table above	2.5-5ng		
Digested versican promoter	See table above	0.5-10ng		
Sterile dH ₂ O	Variable	-		
Incubate in a PCR machine at 42°C for 2 minutes				
T4 DNA Ligase (4U/μl)	2	0.4 U/μl		
10x Reaction buffer	2	1 x		
ATP (10mM)	2	1mM		
Final volume	20	-		
incubate at 4°C for 16 hours to complete ligation				

Table 3.9: Reaction mixture for ligation of the digested versican promoter sequences into the pGL4.10 vector

Note: The pGL4.10 digested plasmid and 29-bp random sequence were ligated at 1:10 ratio where 1 is the amount of vector plasmid and 10 the amount of insert DNA. The ligation mixture included 2 μ l of T4 DNA ligase (0.4 U/ μ l, Stratagene), 2 μ l of 10x Buffer (1x), 2ul of ATP (1mM), 4 μ l of pGL4.10 vector plasmid, 2 μ l of 29-bp random sequence (0.5 ng/ μ l) and were made up to 20 μ l with sterile dH₂O. The water, plasmid and versican promoter DNAs were first incubated at 42°C for 2 minutes and were let to cool down. The rest of the reagents were then added and left at 4°C for 16 hours followed by storage at -20°C.

3.3.6 Transformation of plasmid-DNA into DH5-α competent cells

Bacterial transformation is a technique to introduce a foreign plasmid into bacteria and to use those bacteria to amplify the plasmid in order to make large quantities of it. This is based on the natural function of a plasmid.

For this project, Library Efficiency DH5- α Competent Cells were used as these cells are suitable for cloning experiments in which limiting amount of plasmid DNA will be used (See table 2.7 and 2.8 for the reagents and materials). For the transformation, first the ligation product (pGL4.10 plasmid-versican promoter sequence) was diluted 1:5 with TE buffer (10mM Tris-HCI [pH7.5] and 1mM EDTA). A 200 µl of aliquot of DH5- α competent cells was thawed on ice and 100 µl of cells were transferred into 14 ml Falcon tubes. 1 µL (1~10 ng) of the diluted ligation product and 5 µl (50 pg) of the control pUC19 DNA were added into each tube and gently shaken to mix. The bacteria were incubated on ice for 30 minutes, then heat-shocked at 42°C in a water bath for 45 seconds followed by 2 minutes incubation on ice. Then, 900 µl of room temperature S.O.C medium was added to each tube, and the competent cells were incubated in 37°C with shaking at 225 rpm for 1 hour. 100 µl, 300 µl, and 500 µl of competent cells were plated out onto LB-agar plates containing 100 µg/ml ampicillin for selection (See table 2.10) and the plates incubated overnight at 37°C.

Transformation efficiency of competent cells was determined by transforming 50 pg of

the control pUC19 DNA into competent cells, with 100 μ l of cells being plated out;



3.3.7 Purification of plasmid-DNA

Plasmid DNA was prepared using the Qiagen EndoFree[™] Plasmid Maxi Kit according to the manufacturer's instructions. The plasmid purification protocol is based on a modified alkaline lysis procedure. Insoluble complexes containing chromosomal DNA, salt, detergent and proteins are removed by passing the lysate through a filter. The filtered lysate is incubated on ice with a specific removal buffer which prevents the binding of LPS molecules to an anion-exchange column in the subsequent step allowing purification of DNA containing less than 0.1 endotoxin units per µg of DNA. Plasmid DNA is bound to an anion-exchange column under low salt and pH conditions and RNA, proteins, dyes and low-molecular-weight impurities are removed by a medium-salt wash. Plasmid-DNA is eluted in a high-salt buffer and then concentrated and desalted by isopropanol precipitation. A final wash step with 70% ethanol removes residual salt. The purified DNA is briefly air-dried and dissolved in a small volume of TE buffer (See table 2.7 and 2.8 for the reagents and materials).

The procedure is as follows. A starter culture of 3 ml LB medium containing 100 μ g/ml ampicillin (See table 2.10) was inoculated with a single bacteria colony picked from a selective plate and incubated ~8 hours at 37°C shaking at 224 rpm. The starter culture was diluted 1:1000 into 100 ml of selective LB medium containing 100 µg/ml ampicillin and grown 12-16 hours at 37°C shaking at 224 rpm. 50 ml of over-night grown bacterial culture were centrifuged at 6000 g for 30 minutes at 4°C. The bacteria pellet were lysed by re-suspending in 10 ml of buffer P1 followed by adding 10 ml of buffer P2. The buffer and bacteria were mixed gently by inverting the tube 5 times and then incubated 5 minutes at RT. During the incubation time, the QIAfilter cartridge was prepared by screwing the cap onto the outlet nozzle of it and placing into a convenient tube. 10 ml of chilled buffer P3 was added to the lysate and was mixed gently by inverting the tube 5 times followed by transferring lysate into the QIAfilter cartridge and incubated in RT for 10 minutes. After incubation, the QIAfilter cartridge was placed over a sterile 50 ml centrifuge tube and the cap was removed and then the lysate was filtered into the tube by applying a plunger. The filtered lysate was mixed gently with 2.5 ml of buffer ER and incubated on ice for 30 minutes. A QIAGEN-tip column was equilibrated by applying 10 ml of QBT buffer and allowed to empty by gravity flow. Then, the filtered lysate was transferred into the QIAGEN-tip column and allowed to enter the resin by gravity flow. The QIAGEN-tip column was washed twice with 30 ml buffer QC and the DNA was eluted with 15 ml buffer QN into a sterile 50 ml centrifuge tube. Addition of 10.5 ml isopropanol and centrifugation at 15000 g for 30

minutes at 4°C precipitated the DNA. The supernatant was decanted and the DNA pellet was washed with 5 ml of RT endotoxin-free 70% ethanol and centrifuged at 10000 g for 10 minutes at 4°C. The DNA-pellet was air dried for 10-15 minutes and resuspended in 300-450 μ l endotoxin-free buffer TE and stored at -20^oC.

A small volume of purified plasmid DNA was used to determine DNA concentration (See section 3.3.8). DNA sequencing to confirm the sequence of cloned fragments was done by the Protein Nucleic Acid Chemistry Laboratory (PNACL), University of Leicester.

3.3.8 Quantification of nucleic acids

Absorption spectroscopy is a non-destructive method of measuring the concentration of deoxyribonucleic acid (DNA). Nucleic acids absorb light maximally at a wavelength of 260 nm, and this physical property can be used as a basis to determine the concentration of the nucleic acids in a given solution. DNA samples are sometimes contaminated with trace of protein which has a maximum absorption at 280nm. So the purity of the nucleic acids can be estimated by calculating the ratio of readings taking at 260 nm and 280 nm (OD₂₆₀:OD₂₈₀). Pure preparations of DNA have OD₂₆₀/OD₂₈₀ values of ~1.8 whereas contamination with protein results in lower values.

The measurement of purified DNA (pGL4.10 plasmid-versican promoter) was carried out in a special quartz cuvette since glass or plastic cuvettes show a high selfabsorption at 260 nm. In general, to save the samples, they were diluted 1:100 with TE buffer. Then the DNA concentration was determined by spectrophotometer at OD_{260} and calculated using the following formula:

Concentration of the nucleic acid (μ g/ml) = OD₂₆₀ x ϵ x d

- ϵ : extinction coefficient (50 μ g/ml for dsDNA)
- d: dilution factor

3.4 Quantification of mRNA

3.4.1 Isolation of total RNA

Total RNA was isolated from either normoxic or hypoxic adherent HMDM (See section 3.1.3) and CD14⁺ purified monocytes and CD14⁻ (lymphocyte) (See section 3.1.4) using TRI reagent lysis and isopropanol precipitation according to the manufacturer's instructions as described below (Staples et al., 2000 & 2007 & 2011) (See table 2.2 and 2.8 for the reagents and materials).

Cells were lysed in 500 μ l TRI reagent. The lysate, containing RNA, DNA and protein was transferred into a new autoclaved 1.5 ml centrifuge tube and allowed to stand for 5 minutes at RT to permit complete dissociation of nucleoprotein complexes. At this stage the lysate was either stored at -80°C or used for mRNA isolation. 100 μ l of chloroform was added to the lysate and vortexed vigorously for 15 seconds to form a white emulsion which was allowed to stand for 10 minutes at RT. The resulting mixture was centrifuged at 12000 g for 15 minutes at 4°C. This separates the mixture into 3 phases; a colourless upper aqueous phase containing RNA, an interphase containing DNA and a red organic phase containing protein. The top aqueous layer was transferred into a new autoclaved 1.5 ml centrifuge tube and 500 μ l of isopropanol was added for RNA precipitation. The sample was incubated at RT for 10 minutes and then centrifuged at 12000 g for 10 minutes at 4°C. The supernatant was discarded and the RNA pellet was washed with 1 ml of 75% Ethanol and spun down at 12000 g for 5 minutes at 4°C. The supernatant was discarded and the pellet air-dried for 10 minutes at RT under a laboratory fume hood. The RNA pellet was re-suspended in 20 μ l of DEPC-dH₂O and either immediately reverse transcribed (See section 3.4.2) or stored at -80°C.

3.4.2 Reverse transcription

In an enzymatic reaction, isolated total RNA is reverse transcribed into cDNA. For this reaction, 9 μ l of total RNA (See section 3.4.1) was transferred into an autoclaved 0.5 ml centrifuge tube and incubated in a thermocycler (PCR machine) for 5 minutes at 70°C to denature RNA secondary structure. The sample was cooled to 42°C and then 11 μ l of reaction mix (See table 3.10) was added and the reaction was incubated at 42°C for 1 hour to allow reverse transcription of RNA into cDNA. The reaction was terminated by heating to 90°C for 4 minutes to denature the reverse transcriptase enzyme. The cDNA was stored at -20°C or used immediately for Real-Time- PCR (See table 2.2, 2.8 and 2.13 for the reagents and materials).

Reagent	volume	Final concentration
Hexanucleotides (0.2 μg/μl)	1 µl	10 ng/µl
10 mM dNTP mix	2 µl	1 mM
DEPC-dH ₂ O	2.4 μl	-
5x AMV Buffer	4.0 μl	1 x
AMV Reverse Transcriptase (10 U/μl)	0.8 µl	0.4 U/µl
RNasin [®] Ribonuclease Inhibitor (40 U/μl)	0.8 μl	1.6 U/µl
RNA	9 µl	-
Total volume	20 µl	-

Table 3.10: Reaction mix for reverse transcription. Reagents and reaction mix were kept on ice.

3.4.3 Real-Time PCR

Semi-quantitative PCR was performed using a Roche LightCycler, an integrated thermal cycler and fluorimeter that amplifies the target nucleic acid and monitors the development of the amplification products via measurement of fluorescence. Amplification of target DNA is carried out in the presence of the fluorescent dye SYBR Green which binds to the minor groove of the DNA double helix. The fluorescent intensity is directly proportional to the amount of the amplified product. Subsequent to an amplification run, a melting curve analysis of the PCR product is performed to allow discrimination between specific products and non-specific products. The melting cycle comprises three steps; rapid heating to 94°C to denature all DNA, cooling the

reaction below the annealing temperature of the DNA fragments, and finally slowly heating to monitor the double-stranded DNA as it melts at a characteristic temperature.

For each PCR assay, four 1:5 serially diluted standard samples from a pool of the cDNAs were prepared to be tested. These four standard PCRs were plotted by LightCycler software to calculate the relative concentration of each tested cDNA. Then 3 μ l of samples (standard samples and tested cDNA) were transferred into a Roche LightCycler capillary and 17 μ l of PCR master mix (See table 3.11) was added to each reaction capillary. The PCR master mix contains SYBR Green JumpStart with Tag DNA polymerase which is inactive at RT, due to the presence of heat-labile blocking groups on some of the amino acid residues of the enzyme and subsequently minimizes nonspecific amplification products as no elongation takes place during the period when primers can bind non-specifically. The polymerase is activated by removing the blocking groups at high temperatures in the initial denaturation step of the Real-Time PCR. One extra capillary was prepared using 3 μ l of autoclaved dH₂O with 17 μ l of PCR master mixture as a negative control to confirm that the PCR master mix is not contaminated with DNA. Then all the capillaries were capped and spun down at 400 g for 10 seconds at 4°C. LightCycler capillaries were transferred to a Roche PCR LightCycler real-time PCR instrument for amplification (See table 2.2, 2.8, 2.11 and 2.15 for the materials, reagents and equipment). The reaction mix and set up was prepared according to the following tables;

Reagent	Volume	Final concentration
2x SYBR® Green JumpStart™ Taq ReadyMix	10 µl	1 x
5' Primer (5 μM)	2 µl	0.5 μΜ
3' Primer (5 μM)	2 μΙ	0.5 µM
Sterile dH ₂ O	3 μΙ	-
Total volume	17 μl	_
DNA template	3 μΙ	-

Table 3.11: Reaction master mix for Real-Time PCR. Primers have been listed in table 2.4.1

Segment	Temperature	Time	Cycles
Initial denaturation	95℃	10 min	1
Denaturation	96°C	10 sec	
Annealing	60°C	10 sec	45
Extension	72°C	25 sec	

<u>Table 3.12</u>: PCR cycle used for amplification and detection mRNA of β 2-M, Versican, VEGF and GLUT-1 genes

3.4.4 Quantification of mRNA

First Real-Time PCR was performed using primers for the house keeping gene β -2 microglobulin (β -2M) which is present in all nucleated cells (Gussow et al., 1987; Lee et al., 2007). Different studies have shown that β -2M expression does not alter under low oxygen conditions (Jogi et al., 2002; Foldager et al., 2009; Watson et al., 2009) and previously has been used as an internal control in 0.2% O₂ (Staples et al., 2011). Then the amount of cDNA of interest was determined in a second Real-Time PCR using versican primers. In each run, four internal standards were used consisting of serial 1:5 dilutions of a cDNA pool containing cDNA from each sample (See section 3.4.3). These four standard samples were plotted by LightCycler software to calculate the relative concentration of each cDNA of interest. Then each sample was normalised by dividing the calculated cDNA value of gene of interest by the calculated amount of β -2M of the same sample. The following formulas were used to calculate the relative level and hypoxic induction of the gene of interest (versican in this example);

Versican calculated value for normoxic cDNA	 Relative versican abundance in normoxia 	
β -2M calculated value for normoxic cDNA		
Versican calculated value for hypoxia cDNA β-2M calculated value for hypoxia cDNA	= Relative versican abundance in hypoxia	
Relative versican abundance in hypoxia Relative versican abundance in normoxia	= Versican hypoxic fold induction	

3.5 Transfection

3.5.1 JetPEI[™] DNA transfection reagent

PBMCs were transfected using JetPEI[™] transfection reagent to assess the hypoxic inducibility of versican promoter constructs. JetPEI[™] is a powerful and useful delivery vehicle and ensures effective and reproducible DNA and oligonucleotide transfection into mammalian cells with low toxicity. JetPEI[™] is a linear polyethylenimine (PEI), synthesized and purified by PolyPlus LTD which condenses DNA into positively charged particles. The particles then interact with anionic cell surface residues such as proteoglycans and phospholipids and are taken into the cell via endocytosis. Once inside the lysosomes, PEI works as a proton sponge, resulting in an influx of counterions by accepting protons leading to endosomal pH increasing. Raising endosomal pH could alter protein folding and inactivate endosomal enzymes results in protection of DNA from degradation. This mechanism also leads to endosome swelling and rupture, permitting the release of the JetPEI/DNA complexes into the cytoplasm, thereby allowing nuclear transport for subsequent transcription.

The overall charge of the JetPEI[™]/DNA complexes is therefore crucial for efficient transfection. It is determined by the DNA to reagent ratio. This ratio, which represents the ionic balance within the complexes, is defined as the N/P ratio which refers to the number of nitrogen residues (N) in the JetPEI[™] per phosphate (P) of DNA. To obtain positively charged complexes, an N/P ratio of >3 is required. The following formula, recommended by PolyPlus, was used in order to calculate the N/P ratio which is taking into account the volume of JetPEI[™] reagent used for a given amount of DNA;
N/P ratio =
$$\frac{7.5 \text{ x } \mu \text{ l of jetPEI}^{\text{TM}}}{3 \text{ x } \mu \text{ of DNA}}$$

• Factor 7.5: concentration of nitrogen residues in jetPEI™

• Factor 3: 1 µg of oligonucleotide contains 3 nmoles of anionic phosphate

In preliminary investigation to optimise the transfection procedure, HMDM generated from PBMC were also transfected by $JetPEI^{TM}$ (See table 2.7) according to the manufacturer's instruction. 2×10^6 PBMC cultured in 6-well adherent plate with complete Iscove's Dulbecco's medium and after 5 days incubation under normal oxygen condition (See section 3.1.3), HMDM were transfected with DNA-plasmid and aid of JetPEI DNA transfection reagent is described below. In order to normalise the transfection results, HMDM were co-transfected with Renilla plasmid (pRL-TK) which contains Renilla luciferase and serves as an internal control for normalization (See section 2.7.2). Cells were then incubated under normoxia or hypoxia for 24 hours or 5 days (See section 3.2.1) prior to luciferase assay (See section 3.5.3).

In this project, my preliminary experiments established that the best PBMC transfection results were obtained using an N/P ratio of 8 (1 μ g of DNA, 3.2 μ l of JetPEITM). PBMC were isolated (See section 3.1.1) and seeded in 4ml complete Iscove's medium at a density of 4×10⁶ per well in 6-Well Costar ultra low attachment plates and incubated at 37^oC with 5% CO₂ and 20.9% O₂ for 5 days to allow differentiation of

monocytes to macrophages. On the day of transfection, cells were de-adhered by repetitive pipetting using a 1 ml Gilson and transferred into a sterile 50 ml centrifuge tube. Cells were returned to 37°C prior to transfection. For each transfection (using two wells per transfection) 2 μ g of DNA was diluted with 200 μ l of 150mM NaCl into a sterile 0.5 ml centrifuge tube. In a new sterile 0.5 ml centrifuge tube, 6.4 µl of JetPEI™ reagent was diluted with 200 μ l of 150 mM NaCl. Both tubes were gently mixed and spun down briefly. The 200 µl of diluted JetPEI[™] solution were then added all at once to the 200 µl of diluted DNA solution and immediately mixed by vortexing and spun down briefly to bring the drops to the bottom of the tube. The JetPEI[™] / DNA mixture was incubated for 30 minutes at RT and then 400 µl of mixture was added to 4 ml of complete Iscove's medium containing 4×10⁶ PBMC which had been previously deadhered as described above. Transfected cells were plated in 2 ml at a density of 2×10⁶ per well in 6-Well Nunclon plates (adherent plates) and returned to 37°C with 5% CO₂ and 20.9% O₂ for 1 hour. Cells were then incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 24 hours or 5 days (See section 3.2.1) prior to luciferase assay (See section 3.5.3).

3.5.2 Reporter gene assay

In order to study the activity of a specific promoter, this promoter is cloned into a vector containing a reporter gene (e.g. luciferase). Transfection of the vector into cells results in expression of luciferase using the host cell's transcription factors as well as its transcription and translation machinery.

Luciferase catalyses the formation of light from ATP and luciferin (Reaction is shown below). The emitted light intensity is linearly related to the amount of luciferase expressed and is measured using a luminometer (See table 2.15). The measurements are expressed in relative light units (RLU) per second.



3.5.3 Luciferase assay system

PBMC were transfected with versican promoter regions which were cloned into the pGL4.10 luciferase reporter vector (See section 3.3 and 3.5.1) and cultured under normal oxygen concentration or hypoxia (See section 3.2.1) prior to luciferase assay. The assay was performed according to the manufacturer's instructions (See table 2.7 and 2.8 for the reagents and materials).

Medium was first removed from the wells and the cells were washed twice with 1 ml of 1x PBS. To each well 250 μ l of 1x Lysis buffer were added. The cell lysates were incubated at -80°C for 10 minutes followed by thawing at RT for 30 minutes. 100 μ l of luciferase assay reagent (LAR) was added to a 5 ml polystyrene Falcon tube and then was measured as background in the luminometer (See table 2.15). The luminometer measured the emitted light units for 30 seconds and converted the collected light units into RLU/s. Then 20 μ l of the cell lysate were added to the tube containing 100 μ l

LAR and mixed gently and measured in the luminometer. All transfection experiments were performed triplicate. Additionally, the protein concentration of each lysate was determined using a Pierce 660 nm Protein Assay to normalize luciferase values (See section 3.5.4).

In addition, a dual luciferase assay system (See table 2.7) was used according to the manufacturer's instructions to measure the luciferase signals expressed by HMDM which were transfected with DNA-plasmid (e.g. versican promoter pGL4.10 plasmid) and co-transfected with pRL-TK (See section 2.7.2). Medium was first removed from the wells and to each well 250 μ l of 1x Lysis buffer were added. The cell lysates were incubated at -80°C for 10 minutes followed by thawing at RT for 30 minutes. The luminometer measured the emitted light units for 30 seconds and converted the collected light units into RLU/s. To measure the firefly luciferase expressed by pGL4.10 plasmid, 20 µl of the RT cell lysate were added to the Falcon tube containing 100 µl luciferase assay reagent II (LAR II) and mixed gently and measured in the luminometer. Then by adding 100 μ l of Stop & Glo reagent to the same tube, the firefly luciferase activity is quenched and the Renilla luciferase expressed by pRL-TK, will be activated and measured in the luminometer. For each experiment, 100 μ l of LAR II and then 100 μ l of Stop & Glo reagent were sequentially added to the 5 ml polystyrene Falcon tube and the RLU quantified, and the values used as firefly and Renilla background values.

3.5.4 Measuring protein concentration

The Thermo Scientific Pierce 660 nm protein assay is a quick, ready-to-use colorimetric method for total protein quantification (See table 2.7). The assay is reproducible, rapid and more linear than coomassie-based Bradford assays and compatible with higher concentrations of most detergents, reducing agents and other commonly used reagents. The Pierce 660nm protein assay is based on the binding of a proprietary dyemetal complex to protein in acidic conditions that causes a shift in the dye's absorption maximum, which is measured at 660nm. The dye-metal complex is reddishbrown and changes to green upon protein binding. The colour change is produced by deprotonation of the dye at low pH facilitated by interactions with positively charged amino acid groups in proteins. Therefore, the dye interacts mainly with basic residues in proteins, such as histidine, arginine and lysine and to a lesser extent tyrosine, tryptophan and phenylalanine.

Protein concentrations are estimated by reference to absorbance obtained for a series of standard protein dilutions assayed alongside the unknown samples. In this project, bovine serum albumin (BSA) was used as standard protein and the assay was performed according to the manufacturer's instructions (See table 2.7). First a serial dilution of BSA in 1x luciferase Lysis buffer was prepared as follow: 50 µg/ml, 100 µg/ml, 250 µg/ml, 500 µg/ml and 750 µg/ml (See table 2.13). For the assay, 10 µl of standards, luciferase cell lysate and 1x Lysis buffer as blank were pipetted into a 96well plate. 150 µl of protein assay reagent was then added to each well and the plate was covered and mixed on a plate shaker at 200 rpm for 1 minute following 5 minutes incubation at RT. After incubation the protein concentration was measured in an ELISA

reader at 660 nm (See table 2.15). To calculate the protein expression, a standard curve was prepared by plotting the average blank-corrected 660 nm measurement for each BSA standard versus its concentration in μ g/ml. Then the standard curve was used to determine the protein concentration of each luciferase cell lysate sample.

3.6 Nuclear Extraction

The method originally described by Schreiber et al., (1989) was used for nuclear protein isolation, with slight modifications as described below.

The non-adherent monocytic cell line U937 was cultured in 75cm² tissue culture flasks using complete RPMI1640 medium (See section 3.1.5). Cells were counted using Trypan blue (See section 3.1.2) and 5.0 x 10^5 cells/ml were transferred into 10 ml of fresh complete RPMI1640 medium in the 75cm² flask which was incubated either under normoxia or hypoxia for 24 hours prior to nuclear extraction (See section 3.2.1). In order to inhibit proteolysis, two freshly cocktail of proteases inhibitors; solution 1 from nuclear extraction buffer A and solution 2 from nuclear extraction buffer B were prepared fresh before starting and kept on ice prior to use (See table 2.13). U937 cells were removed from normoxia and hypoxia incubators and transferred into a sterile 15 ml centrifuge tube followed by 5 minutes centrifugation at 400 g. The supernatant was decanted and cell pellet was washed twice with 10 ml of ice-cold 1x PBS and centrifuged at 400 g for 5 minutes. Then, cell pellet was re-suspended in 1 ml of icecold 1x PBS and transferred into a 1.5 ml sterile centrifuge tube followed by 5 minutes centrifugation at 4°C for 400 g. The supernatant was removed and the cell pellet was re-suspended in 4 packed cell volumes of ice-cold solution 1. The pellet was incubated

on ice for 5 minutes during which the cells swell due to the hypotonic buffer. After ice incubation, Nonidet NP40 (Igepal) was added to a final concentration of 0.6 % and the mixture was then vortexed for 10 seconds and centrifuged at 12000 g for 30 seconds at 4°C. The supernatant was removed (cytoplasmic extract) and transferred to a new sterile 1.5 ml centrifuge tube and kept in -80°C. The pellet was washed with 1 ml of 1x PBS and spun down at 12000 g for 30 seconds at 4°C. This stage was applied to make sure there would not be any trace of cytoplasmic proteins in the nuclear extract proteins. Then the supernatant was removed and the pellet was re-suspended in 1 packed volume of ice-cold solution 2 followed by addition of glycerol to a final concentration of 20%. The mixture was incubated at 4°C for 15 minutes on a rotator. After incubation, the mixture was centrifuged at 12000 g for 5 minutes at 4°C and supernatant (nuclear extract) was transferred to a new sterile 1.5 ml centrifuge tube and kept in -80^oC and used in western blot analysis (See section 3.7). 2 μ l of cytoplasm nuclear extract proteins were applied to a Nanodrop cuvette-free or spectrophotometer to determine the protein concentration of each sample. 2 μ l of solution 2 were used as a blank to zero the Nanodrop (See table 2.15).

Note: please see table 2.4, 2.8 and 2.13 for the reagents, buffers and materials which were used for nuclear protein extraction

3.7 Western blotting

Western blotting is a technique used to identify and locate proteins based on their ability to bind to specific antibodies. Western blot analysis can detect a protein of interest from a mixture of a great number of proteins. In western blotting, prior to protein immobilization on the PVDF or nitrocellulose membrane, sample proteins are separated using SDS polyacrylamide gel electrophoresis (SDS-PAGE) which can give information about the size of the protein with comparison to a size marker or ladder, and also give information on protein expression with comparison to a control such as untreated sample or another cell type or tissue.

The Xcell II Minicell electrophoresis and blotting system was used according to the manufacturer's instruction (See table 2.15). For electrophoresis, nuclear protein extract samples (90 μ g) (See section 3.6) were diluted 1:6 with 6x loading buffer to a maximum of 18 μ l (Kept on ice at all times) following 5 minutes incubation at 95°C in the thermocycler and then loaded to the gel. Samples were then size fractionated on SDS-PAGE gel containing 8% (resolving gel) and 4% polyacrylamide (stacking gel) in 1x western blot running buffer for 60 minutes at 40 mA. Pre-stained high molecular weight protein ladder was also used to assess product size. Following electrophoresis, proteins were electro-blotted onto Hybond-P PVDF membrane in western blot transfer buffer for 90 minutes at 20 V.

Note: please see table 2.5 and 2.13 for the reagents and which were used for western blotting.

3.7.1 Immunodetection of β-catenin

Membranes were washed three times in 1x TBS/Tween 20 buffer for 5 minutes at RT and then were blocked overnight at 4°C in western blot blocking buffer. Next day, the membranes were washed twice for 10 minutes in 1x TBS/Tween 20 buffer followed by incubation for 1 hour at RT in a 1:2000 dilution of rabbit polyclonal anti- β -Catenin antibody in western blot blocking buffer. The membranes then were washed twice for 20 minutes and twice for 10 minutes in 1x TBS/Tween 20 buffer. Then the membranes were probed with secondary antibody for 1 hour at RT in a 1:4000 dilution of polyclonal anti-rabbit immunoglobulins/HRP in western blot blocking buffer. Again, the membranes were washed twice for 20 minutes and twice for 10 minutes in 1x TBS/Tween 20 buffer and the antibody-labelled proteins were detected using ECL Advance western blotting detection reagents (Amersham) according to the manufacturer's instructions and visualised on film (See table 2.5).

Note: please see table 2.13 and 2.14 for the buffers and antibodies which used in Immunodetection of hypoxic accumulation of β -catenin

3.7.2 Re-probing the membranes

Membranes were re-probed for β -actin protein as an internal control for western blotting analysis. For re-probing, the membranes were washed three times for 10 minutes in 1x TBS/Tween 20 buffer following incubation overnight at 4°C in western blot blocking buffer. Rabbit polyclonal anti-actin antibody was used at a 1:2000 dilution in western blot blocking buffer for 1 hour at RT. The membranes were probed with secondary antibody for 1 hour at RT in a 1:10000 dilution of polyclonal anti-rabbit

immunoglobulins/HRP in western blot blocking buffer. All the immunodetection procedure was performed as described in section 3.7.1.

Note: please see table 2.13 and 2.14 for the buffers and antibodies which used in Immunodetection of β -actin protein.

3.8 Flow cytometric analysis of versican protein expression

Flow cytometry is commonly used to identify and quantify cell surface or intracellular markers expressed by cells in mixed cell populations, following staining of cells using fluorescent monoclonal antibodies specific for the target markers. The cells may be alive or fixed at the time of measurement. Cells are passed single-file through a laser beam by continuous laminar flow of a fine stream of the cell suspension. Each cell scatters some laser light and also emits fluorescent light from the fluorochrome, excited by the laser. The cytometer typically measures several parameters simultaneously for each cell, including: Forward scatter intensity (approximately proportional to cell size), 90 degree or right angle side scatter intensity (approximately proportional to the quantity of granular structures within the cell) and florescence intensities at several wavelengths (allowing the determination of the density of antigen on or in the cell and the number of cells expressing the antigen) which is also known as mean florescence intensity.

In order to detect the expression of versican protein, PBMC were treated with brefeldin A (See table 2.6 and 2.13). Brefeldin A is an inhibitor of intracellular protein transport (Hunziker et al., 1992). Incubation of cells in culture with brefeldin A leads to

blockade of protein transport to the golgi complex (GC) and accumulation of proteins in the endoplasmic reticulum (ER), leading to inhibition of protein secretion. Thus the protein of interest accumulates within the cell (Sciaky et al., 1997; Nebenfuhr et al., 2002).

PBMCs were isolated (See section 3.1.1) and seeded in 2ml complete Iscove's medium at a cell density of 2×10⁶ per well in 6-Well Costar ultra low attachment plates and incubated at 37°C with 5% CO₂ and 20.9% O₂ for 5 days to allow differentiation of monocytes to macrophages (See section 3.1.3). After 5 days, PBMCs were incubated under either normoxia (20.9% O_2) or hypoxia (0.2% O_2) for 4 days (See section 3.2.1). Cells were then treated with 2 μ g/ml brefeldin A and incubated under the same oxygen tension for a further 24 hours prior to quantification of versican protein by fluorescence-activated cell sorting (FACS). First cells were harvested and transferred into the 5 ml polystyrene Falcon tube followed by 5 minutes centrifugation at 200 g. Cell pellets were then washed with 1 ml of 1x PBS and centrifuged at 200 g for 5 minutes. Supernatant was removed and cells were fixed using 1% paraformaldehyde (formalin) in 1x PBS/2%FCS followed 20 minutes incubation in RT and then centrifuged at 200g for 5 minutes. Supernatant was discarded and to permit intracellular staining cells were permeabilised with 0.1% saponin in 1x PBS/2%FCS and incubated for 15 minutes at RT. Saponin a detergent-like molecule, has been used to cause permeabilisation by solubilising cell membrane cholesterol, thereby, allowing cytometric analysis of nuclear and other intracellular antigens (Jacob et al., 1991; Krutzik and Nolan, 2003). Cell suspensions were then centrifuged at 200 g for 5 minutes and cell pellets were washed twice with 1 ml of PBS/2%FCS followed by 5

minutes centrifugation at 200 g. Cell pellets were re-suspended in 1 ml of PBS/2%FCS and then equally divided into two 5 ml polystyrene Falcon tubes. Cells were then incubated on ice for 45 minutes in a 1:50 dilution of monoclonal anti- human versican antibody or rat lgG1 isotype control with 0.1% saponin in 1x PBS/2%FCS buffer. Cell suspensions were centrifuged at 200 g for 5 minutes and then washed with 1 ml of PBS/2%FCS followed 5 minutes centrifugation at 200 g. Cell pellets were re-suspended in 500 µl of 0.1% saponin in 1x PBS/2%FCS buffer and incubated with 1:40 diluted antirat lgG conjugated to FITC on ice for 45 minutes. Cell suspensions were centrifuged at 200 g for 5 minutes and pellets were washed with 1 ml of PBS/2%FCS followed by 5 minutes centrifugation at 200 g. Cell pellets were then re-suspended in 500 µl of 1x PBS and stored at 4°C in the dark, prior to cell acquisition on a FACSCalibur flow cytometer followed by analysis using CellQuest software (Becton Dickinson). The instrument settings were obtained from published work (Staples et al., 2011) and shown in table 3.13.

Parameter	Voltage	AMPGain	Mode	Threshold
FCS	E-1	6.53	Lin	52
SSC	250	1.0	Log	0
FL1	490	1.0	Log	0

Table 3.13: Instrument settings for the flow cytometric acquisition of human PMBCs. FCS: Forward Scatter, SSC: Side Scatter, FL1: Filter 1.

Note: please see table 2.6, 2.13 and 2.14 for all the reagents, buffers and antibodies used in flow cytometry experiments.

3.9 Database analysis

The versican proximal promoter sequence was searched using MatInspector (Genomatix,http://www.genomatix.de/online_help/help_matinspector/matinspector_ help.html) and transcription element search system (TESS) software (http://www.cbil.upenn.edu/cgibin/tess/tess?RQ=WELCOME) for putative transcription factor binding sites.

3.10 Statistics

Statistical analysis was done using the statistical software GraphPad Prism 5 (GraphPad Software Inc., San Diego, USA) Student's one tailed and two-tailed, paired t-test was selected for the analysis after inputting the data. P-value was calculated for significance level. The results were considered statistically significant if ***, $p \le 0.001$; **, $p \le 0.01$; and *, $p \le 0.05$.

4. <u>Results</u>

Chapter 4: Results

Hypoxic regulation of versican transcription in primary human macrophages

4.1 Introduction

Hypoxia, which refers to low oxygen tension in biological systems, is a feature of many pathological tissues such as solid tumours (Hockel et al., 1996; Vaupel et al., 2001), wounds (Lee et al., 2009), atherosclerotic plaques (Simanonok et al., 1996), arthritic joints (Mapp et al., 1995) and sites of infection and inflammation (Wenger et al., 2005). The median oxygen tension in normal tissues is usually between 20 and 70 mmHg but in ischaemic pathological sites can reach zero mmHg (Vaupel et al., 2001). However, hypoxia is also found in healthy tissues such as spleen (oxygen level as low as 0.5%) (Caldwell et al., 2001). Cells of the monocyte/macrophage lineage are involved in all of the above pathologies (Lewis et al., 1999; Griffiths et al., 2000; Lewis et al., 2000). It has been known for some time that macrophages accumulate in poorly vascularised, hypoxic sites and respond rapidly to the hypoxia present by altering their expression of a wide array of genes (Murdoch et al., 2005; Burke et al., 2003)

Unpublished work by Staples and B. Burke et al., (personal communication) found that the extracellular matrix (ECM) proteoglycan versican is one such hypoxia inducible gene which can be up-regulated up to 700-fold at the mRNA level in human peripheral blood mononuclear cells (PBMC) after 5 days incubation under 0.2% O₂ (Fig 1.8). Versican is a member of the large aggregating chondroitin sulphate proteoglycans, and is distributed in a wide variety of tissues during development and injury (Yee et al.,

2007). Recent studies by Asplund et al., (2010) showed the hypoxic inducibility of the versican gene in human monocyte-derived macrophages (HMDM) after 24 hours under $0.5\% O_2$.

Here, the effect of short term (18 hours) hypoxia (0.2% O₂) on versican mRNA expression was first investigated in adherence-purified HMDM in different donors. Then in order to investigate whether macrophages are the principle cell type showing hypoxic up-regulation of versican in our PBMC cultures, Real Time RT-PCR was used to compare the level of versican mRNA expression in CD14⁺ monocyte-derived macrophages with lymphocytes (CD14⁻). In addition, the decay of versican mRNA was investigated in normoxic and hypoxic adherence-purified HMDM.

4.2 Short term hypoxia induces versican mRNA in adherence-purified HMDM

The effect of 18 hours (short term) hypoxia (0.2% O₂) was investigated on versican mRNA expression in primary HMDM, purified by adherence from PBMC from 15 different healthy donors. As described in section 3.1.1,PBMCs were cultured in 2ml complete Iscove's medium at a cell density of 2×10^6 per well in 6-well adherent Nunclon plates and incubated at 37° C with 5% CO₂ and 20.9% O₂ for 5 days to allow differentiation of monocytes into macrophages (See section 3.1.3). Lymphocytes are non-adherent so this procedure typically yields a cell population of >95% monocytemacrophages (Burke et al., 2003). After 5 days incubation, cells were either cultured under normal oxygen concentrations (20.9% O₂) or hypoxia (0.2% O₂) for a further 18

hours prior to RNA isolation followed by determination of β -2M (a housekeeping gene) and versican mRNA levels by Real-Time RT-PCR (See section 3.4).

As previously described, first Real-Time PCR was performed on cDNA samples using primers for the house keeping gene β -2M. Studies in our lab (Staples et al., 2011) have shown that β -2M has a constant level of expression under normoxia and hypoxia and has also been used as an internal control in a variety of studies (Jogi et al., 2002; Watson et al., 2009). As figure 4.1 shows, to obtain relative concentration, four fivefold serially-diluted standards (STD1-4; 1, 0.2, 0.04 and 0.008) were prepared and were run in parallel with the samples and negative control; PCR H2O (PCR -ve control) and the standard curve were plotted by the Lightcycler (Roche) software to calculate the relative concentration of β -2M cDNA for each sample (See section 3.4.3). The specificity and purity of the quantified samples was confirmed by examining the melting curve. As figure 4.2 indicates, all the PCR products (standards and samples) showed a similar melting temperature except the negative control (PCR -ve) which by comparison with previous experiments in which the product size had been confirmed on agarose gels, confirms that the correct PCR product was amplified and that no contamination was observed in the reaction. Then the amount of cDNA of the gene of interest, versican, was determined in second Real-Time PCR using versican primers, on the same standards, samples and negative control (Fig 4.3 and 4.4). As previously described in section 3.4.4, to normalise the amount of versican cDNA for each sample (normoxia or hypoxia), the calculated cDNA value of versican was divided by the calculated amount of β -2M cDNA in the same sample. To calculate the hypoxic fold induction of versican in adherence-purified HMDM of a particular donor, normalised

abundance of versican in hypoxia was divided by normalised abundance of versican in normoxia.



Figure 4.1: LightCycler PCR amplification of β-2M cDNA in normoxic (N, 20.9% O_2) and hypoxic (H, 0.2% O_2) adherence-purified HMDM. The figure shows four standards (STD1-4) produced by serial 1:5 dilution (1, 0.2, 0.04 and 0.008) which were prepared and run in parallel with the samples (HMDM N and HMDM H) and negative control, H₂O (PCR -ve). The standard curve was plotted by the software to calculate the relative concentration of β-2M cDNA for each sample.



Figure 4.2: LightCycler melting curve of amplified β-2M cDNA in the samples described in Fig 4.1. The figure shows similar melting temperatures in the four standards (STD1-4) and samples (HMDM N and HMDM H) but not in the negative control; H_2O (PCR –ve) which, by comparison with previous experiments in which the product size had been confirmed on agarose gels, confirms that the correct PCR product was amplified. The small non-specific product peak in the negative control could be due to primer dimer formation and confirms no contamination in the reaction.



Figure 4.3: LightCycler PCR amplification of versican cDNA in normoxic (N, 20.9% O_2) and hypoxic (H, 0.2% O_2) adherence-purified HMDM. The figure shows four standards (STD1-4) with a serial dilution of 1:5 (1, 0.2, 0.04 and 0.008) which were prepared and run in parallel with the samples (HMDM N and HMDM H) and negative control, H_2O (PCR -ve). The standard curve was plotted by the software to calculate the relative concentration of versican DNA for each sample.



Figure 4.4: LightCycler melting curve of amplified versican cDNA in samples detailed in Fig 4.3. The figure shows similar melting temperatures in the four standards (STD1-4) and samples (HMDM N and HMDM H) but not in the negative control; H_2O (PCR -ve) which, by comparison with previous experiments in which the product size had been confirmed on agarose gels, confirms that the correct PCR product was amplified and that no contamination was observed.

This method was applied to 15 different donors to confirm the hypoxic up-regulation of versican mRNA. As figure 4.5 shows, I observed substantial induction of versican mRNA (average 34-fold) in all 15 different donors after 18 hours hypoxia despite high variability in hypoxic fold induction among these donors, ranging from 11 to 125 fold. These data clearly confirmed the hypoxic up-regulation of versican mRNA in adherence-purified HMDM.



Figure 4.5: Effect of short term (18 hours) hypoxia on versican mRNA expression in adherence-purified HMDM. Real-Time RT-PCR analysis was used to determine the effect of 18 hrs hypoxia on versican mRNA expression in 15 different donors. 2 x 10^6 cells per well were exposed to normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 18 hours. All data were normalised to β-2M mRNA levels in the same samples determined by separate PCR. Fold induction was calculated by dividing the normalized versican mRNA level in hypoxia by the normalised versican mRNA level in normoxia.

4.3 Comparison of versican mRNA hypoxic up-regulation in macrophages and lymphocytes

Real-Time RT-PCR was used to compare the induction of versican mRNA expression in macrophages and lymphocytes, to determine the relative importance of these cell types in preliminarily experiments using PBMC cultures (Fig 1.8), which contain a mixture of monocyte-macrophages and lymphocytes, with the majority, typically 90-95%, being lymphocytes (Burke et al., 2003). As described in section 3.1.4, PBMCs were first prepared and then CD14⁺ monocytes were separated from CD14⁻ cells (mainly lymphocytes) using Miltenyi MACS magnetic beads linked to an antibody specific for the monocyte-specific surface antigen CD14. Cells were then seeded in 2ml complete Iscove's medium at a cell density of 1×10^6 per well in 6-well adherent Nunclon plates and incubated at 37° C with 5% CO₂ and 20.9% O₂ for 5 days to allow differentiation of CD14⁺ purified monocytes into macrophages (See section 3.1.3). After 5 days incubation, cells were either cultured under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for a further 18 hours prior to RNA isolation followed by determination of versican mRNA level by Real-Time RT-PCR (See section 3.4).

As figure 4.6 indicates, 18 hours of hypoxia significantly induced versican mRNA expression in adherence-purified HMDM (34-fold) and CD14⁺ magnetic bead purified monocyte-derived macrophages (25-fold) but not in lymphocytes (CD14⁻ cells). These data clearly suggest that macrophages, and not lymphocytes, are the main type of peripheral blood mononuclear cell responsible for hypoxic up-regulation of versican.



Figure 4.6: Effect of short term (18 hours) hypoxia on versican mRNA expression in human primary macrophages and Lymphocytes. Versican mRNA levels were quantified by Real-Time RT-PCR in HMDM (n=15), CD14⁺ magnetic bead purified monocyte-macrophages and CD14⁻ cells (lymphocytes) (n=3). 2 x 10⁶ cells per well were exposed to normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 18 hours. All data were normalised to β2-M mRNA levels determined by separate PCR. Data are from independent experiments using different donors and are expressed as means ± SEM. Data were further analysed for significant induction using paired t-tests. *** = p <0.001,* = p <0.05 when compared to the normoxic samples

4.4 Hypoxic up-regulation of versican mRNA occurs at the transcriptional level

To determine whether the observed up-regulation of versican mRNA by hypoxia is due to increased transcription or increased versican mRNA stability, the decay of versican mRNA was investigated in normoxia and hypoxia in adherence-purified HMDM. In order to block further transcription, cells were treated with actinomycin D, and the amounts of versican mRNA at a series of time points were determined by RT-PCR. For these experiments, adherence-purified HMDM were isolated and cultured as normal (See sections 3.1.1 and 3.1.3) and after 5 days maturation, were incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for a further 18 hours prior to actinomycin D treatment. Cells were then treated with actinomycin D at a final concentration of 10 μ g/ml and returned to normoxia or hypoxia for a further 2 hours, 8 hours, 12 hours, 24 hours and 36 hours respectively prior to RNA isolation followed by determination of versican mRNA level by Real-Time RT-PCR.

As Figure 4.7 shows, very similar levels of versican mRNA degradation were observed in normoxia and hypoxia at all time points up to 36 hours. Since these data demonstrate that versican mRNA stability is not increased by hypoxia, it appears that the increase observed in versican mRNA in hypoxic adherence-purified HMDM is due to transcriptional up-regulation rather than increased mRNA half-life by hypoxia.



Figure 4.7: Effect of hypoxia on versican mRNA decay in adherence-purified HMDM. Adherence-purified HMDM were quantified by Real-Time RT-PCR to assess the decay of versican mRNA in normoxia and hypoxia. 2×10^6 cells per well were first treated with actinomycin D at a final concentration of 10 µg/ml and then were exposed to normoxia (20.9% O₂) or hypoxia (0.2% O₂) for a further 2h, 8h, 12h, 24h and 36h prior to RNA isolation. Data were normalised to β2-M mRNA levels, and are shown as a percentage of the initial time zero (Actinomycin D addition time) mRNA level in normoxia or hypoxia. Data from 5 independent experiments using different donors are expressed as means ± SEM.

As a control to validate the RNA samples generated by this actinomycin D mRNA stability approach, Real-Time RT-PCR using GLUT-1 primers was performed on the same samples to determine the decay of GLUT-1 mRNA, a gene known to be transcriptionally induced by hypoxia (Burke et al., 2003). As figure 4.8 shows that, as for versican, similar levels of GLUT-1 mRNA degradation were observed in normoxia and hypoxia at all time points examined. This data confirms, as expected, that hypoxia-induced increases in GLUT-1 mRNA are mediated by transcriptional up-regulation rather than increased mRNA stability in hypoxia.



Figure 4.8: Effect of hypoxia on GLUT-1 mRNA decay in adherence-purified HMDM. Adherence-purified HMDM were quantified by Real-Time RT-PCR to assess the decay of GLUT-1 mRNA in normoxia and hypoxia. Experimental details as explained for figure 4.7. Data from 5 independent experiments using different donors are expressed as means ± SEM.

4.5 Summary

In this chapter, the effect of 18 hours period of hypoxia on versican mRNA expression was investigated in adherence-purified HMDM generated from PBMC in 15 different donors. As figure 4.5 showed, Real-Time RT-PCR on adherence-purified HMDM in 15 different donors indicated substantial induction of versican mRNA after 18 hours hypoxia. There was high variability in hypoxic fold induction among these donors (average induction 34-fold, range 11-125 fold). Further study was carried out to investigate which cell type was principally responsible for versican expression, since adherence-purified HMDM cultures are typically no more than 95% macrophages (Burke et al., 2003). As figure 4.6 showed, 18 hours of hypoxia significantly induced versican mRNA expression in adherence-purified HMDM (34-fold) and CD14⁺

monocyte-derived macrophages (25-fold) but not in lymphocytes (CD14⁻), indicating that macrophages as the principle cell type showing hypoxic up-regulation of versican mRNA. In addition, analysis of the decay of the mRNA of versican (Fig 4.7) and GLUT-1 (Fig 4.8) showed similar kinetics of mRNA degradation in normoxic and hypoxic adherence-purified HMDM at all time points which clearly confirmed that hypoxic upregulation of versican and GLUT-1 mRNA is not due to changes in mRNA stability and must occur at the transcriptional level. The data presented in this chapter, showing that the strong up-regulation of versican mRNA by hypoxia occurs at the transcriptional level, indicate that analysis of the versican promoter must be carried out to elucidate the mechanism of up-regulation. This work is described in chapter 6.

Chapter 5: Results

Hypoxic regulation of versican protein synthesis in human primary macrophages

5.1 Introduction

Some previous studies have investigated the level of versican protein expression in human (Zimmermann et al., 1994; Kischel et al., 2010) and mouse tissues (Russell et al., 2003). Du Cros et al., (1995) also showed versican protein expression in skin and blood vessels of adult human tissue and during hair follicle development in mouse. Further study by Domenzain et al., on media from subconfluent cultures by immunoblotting, demonstrated that invasive human melanoma SK-ml-131 cells express versican protein isoforms V0 and V1 (Domenzain et al., 2003 & 2009). A recent study observed that versican is present in the CD68-positive macrophage-rich areas of human carotid lesions (Asplund et al., 2010). However, to date there has been only one study addressing the question of whether versican protein is induced by hypoxia. Asplund et al., using Immunoblotting, did not show hypoxic induction of versican protein level in normoxia and hypoxia in HMDM. They incubated HMDM for 24 hours under normoxia $(21\% O_2)$ or hypoxia $(0.5\% O_2)$ and then cell lysates were separated by SDS–PAGE on 5% gels and immunoblotted for versican protein. However further investigation, using radioactive quantification of versican-bound GAG chains using ³⁵Ssulphate and D-[6-³H]-glucosamine hydrochloride showed 3 and 7-fold increases in GAG chains respectively, in cells exposed to hypoxia for 24 hours compared to cells incubated in normoxia (Asplund et al., 2010).

In this project, further study was carried out to investigate in more detail the effect of hypoxia on versican protein synthesis in human primary macrophages using intracellular flow cytometry. Based on previous studies on quantification of versican protein in conditioned medium (Serra et al., 2005; Hasegawa et al., 2007) and using intracellular flow cytometry (Wang et al., 2008), PBMCs were treated with brefeldin A, an intracellular protein transport inhibitor which causes blockade of protein transport to the Golgi complex and accumulation of proteins in the endoplasmic reticulum, leading to inhibition of protein secretion (Hunziker et al., 1992; Sciaky et al., 1997; Nebenfuhr et al., 2002). Here, versican protein expression was analysed in PBMC after 24 hours or 5 days incubation under normoxia (20.9% O₂) or hypoxia (0.2% O₂) using intracellular staining followed by flow cytometry. In addition, the level of versican protein expression and hypoxic induction in monocytes/macrophages was compared to those in lymphocytes.

5.2 Long term hypoxia induces versican protein expression in human primary macrophages

The effect of 5 days (long term) hypoxia (0.2% O_2) was investigated on versican protein expression in PBMC from 5 different healthy donors using intracellular flow cytometry. In order to optimize the protocols being used, pilot experiments were performed to investigate whether brefeldin A is necessary for the detection of intracellular protein as described previously. As described in section 3.1.1, PBMCs were cultured in 2ml complete Iscove's medium at a cell density of 2×10⁶ per well in 6-Well Costar ultra low attachment plates and incubated under normoxia (20.9% O_2) for 5 days. After 5 days incubation, PBMCs were incubated under either normoxia or hypoxia (0.2% O_2) for a

further 5 days prior to intracellular staining. As described in section 3.8, in order to quantify intracellular protein level, a cell permeabilisation protocol was used prior to addition of a 1:50 dilution of rat monoclonal anti- human versican antibody or rat IgG1 isotype control followed by incubation with a 1:40 dilution of anti-rat IgG conjugated to FITC. Samples were then acquired using a FACSCalibur flow cytometer, followed by analysis with CellQuest software (See section 3.8).

As figure 5.1 shows, brefeldin A-untreated normoxic (A) and hypoxic (C) PBMCs were plotted onto a dot-scatter plot which exhibited lymphocytes with low forward scatter (FSC), side scatter (SSC) and monocytes/macrophages with higher FSC and SSC (Smyth et al., 2005; Alexis et al., 2006; Lay et al., 2009). When a rough population of monocyte/macrophages (R1) was gated onto a histogram-plot, despite having some background with isotype control (green line) which could be due to non-specific binding, no difference of versican fluorescent intensity (n=3) were observed in hypoxia in comparison with normoxia (blue line). Increasing the size of the monocyte/macrophage region on the dot-scatter plot did not show any increase in hypoxia versican fluorescent intensity (data not shown).



Figure 5.1: Effect of long term (5 days) hypoxia on versican protein expression in brefeldin A-untreated PBMC-derived macrophages. Untreated PBMCs were subjected to flow cytometry to analyse the effect of 5 days hypoxia on versican protein expression. 2×10^6 cells per well were cultured under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 5 days prior to flow cytometry analysis. Normoxic and hypoxic PBMCs were plotted onto a dot-scatter plot (A and C). The monocyte/macrophage region (R1) with high forward (FSC) and side scatter (SSC) was gated onto a histogram-plot and then examined for normoxic and hypoxic versican fluorescent intensity (B and D) (blue line) against isotype control (green line). A representative example of 3 independent experiments is shown.

Identical studies, except for the addition of brefeldin A, were performed on PBMC treated with brefeldin A to investigate the effect of 5 days (long term) hypoxia on versican protein expression. Human PBMCs were isolated and cultured as described in section 3.1.1. After 5 days incubation, PBMCs were incubated under either normoxia $(20.9\% O_2)$ or hypoxia $(0.2\% O_2)$ for 4 days prior to brefeldin A treatment. Cells were then treated with 2 µg/ml brefeldin A and incubated under the same oxygen tension for a further 24 hours prior to quantification of protein by intracellular flow cytometry as described in section 3.8.

As figure 5.2 (plots A and C) demonstrates, when a rough population of monocytes/macrophages (R1) was gated onto a histogram-plot, despite having some isotype background which could be due to non-specific binding (green line), a remarkable increase of versican fluorescent intensity was observed (n=5) in hypoxia (D) (blue line) in comparison with normoxia (B) (blue line) against isotype control (green line). As figure 5.2, graph E shows, flow cytometric analysis of versican protein expression on PBMC derived macrophages from 5 different donors indicated significant induction of versican protein (3-fold) after 5 days in hypoxia. In order to eliminate any possible background due to non-specific binding, the versican fluorescent intensity was subtracted from isotype fluorescent intensity in normoxia and hypoxia in each separate experiment. As well as demonstrating clear hypoxic induction, these experiments demonstrated the need for brefeldin A in order to allow accumulation of versican protein within the cells to allow detection and quantification by flow cytometry.



Figure 5.2: Effect of long term (5 days) hypoxia on versican protein expression in brefeldin A treated PBMC-derived macrophages. Treated PBMCs were subjected to flow cytometry to analyse the effect of 5 days hypoxia on versican protein expression. 2×10^6 cells per well were cultured under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 4 days followed brefeldin A treatment and incubated under the same oxygen tension for a further 24 hours. Normoxic and hypoxic PBMCs were plotted onto a dot-scatter plot (A and C). The monocyte/macrophage region (R1) with high forward (FSC) and side scatter (SSC) was gated onto a histogram-plot and then examined for normoxic and hypoxic versican fluorescent intensity (B and D) (blue line) against isotype control (green line). A representative example of 5 independent experiments is shown. (E) Versican protein fold induction in region R1 cells from 5 independent experiments using different donors is expressed as means ±SEM. Isotype control levels were subtracted from the data. The normoxic value in each experiment was assigned an arbitrary value of 1. Data is from 5 independent experiments and were further analysed using paired two-tailed t-tests. ** p < 0.01

5.3 Short term hypoxia did not induce versican protein expression in human primary macrophages

The effect of 24 hours (short term) hypoxia (0.2% O_2) was also investigated on versican protein expression in PBMC treated with brefeldin A from 3 different healthy donors using intracellular flow cytometry. PBMCs were cultured in 2ml complete Iscove's medium at a cell density of 2×10⁶ per well in 6-Well Costar ultra low attachment plates and incubated under normal oxygen condition for 5 days (See section 3.1.3). After 5 days, PBMCs were treated with 2 µg/ml brefeldin A and incubated under either normoxia (20.9% O_2) or hypoxia (0.2% O_2) for 24 hours prior to quantification of protein by intracellular flow cytometry as described in section 3.8.

As figure 5.3 shows, normoxic (A) and hypoxic (C) PBMCs were first plotted onto a dot-scatter plot. When the same region of monocytes/macrophages (R1) as previously showed in Fig 5.1 and 5.2 (plot A and C) was selected and gated onto the histogramplot, despite having some background with isotype control (green line) which could be due to non-specific binding, no increase of versican fluorescence intensity (n=3) was observed in hypoxia (D) in comparison with normoxia (B) (blue line) against isotype control (green line). Increasing the monocyte/macrophage region size on the dot scatter plot did not result in any increase in versican fluorescent intensity in hypoxia (data not shown). This finding is concordant with the observations of Asplund et al., (2010) who failed to show hypoxic up-regulation of versican protein in HMDM after 24 hours.



Figure 5.3: Effect of short term (24 hours) hypoxia on versican protein expression in PBMC derived macrophages treated with brefeldin A. PBMCs were subjected to flow cytometry followed to analyse the effect of 24 hours hypoxia on versican protein expression. 2×10^6 cells per well were treated with brefeldin A and then exposed to normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 24 hours. Experimental details as explained for figure 5.2. A representative example of 3 independent experiments is shown.

5.4 Hypoxia-induced increases of versican protein expression are confined to monocyte/macrophage cells

To investigate whether the hypoxic up-regulation of versican protein in monocytes/ macrophages demonstrated by intracellular flow cytometry (Fig 5.2, plot E) is affected by monocyte maturation/differentiation, as indicated by side scatter (cell granularity) and forward scatter (cell size) (Alexis et al., 2006; Lay et al., 2009), different subpopulations of 5 days old normoxic and hypoxic monocyte/macrophages were analysed for versican fluorescent intensity.

When looking at macrophages with different forward signals, the cells with high forward and side scatter (large, more mature macrophages) showed a higher versican specific mean fluorescent intensity compared to cells with a low forward and side scatter (small, less mature monocyte/macrophages) (Fig 5.4, plot C and F). To further analyse this, the monocyte/macrophage populations on the dot-scatter plots were divided into 3 separate regions (R3-R1) on the basis of increasing FSC, as shown in figure 5.4, plots A and D. The proportion of cells in each of these regions differed in normoxia and hypoxia (Fig 5.4, plots B and E), likely reflecting the effect of prolonged hypoxia on the increase in cell size (Liu et al., 2006) which is associated with macrophage maturation (Alexis et al., 2006; Lay et al., 2009). In both normoxia and hypoxia, increasing cell size (forward scatter) significantly correlated with increasing versican mean fluorescent intensity (Fig 5.4, plots C and F) but not with isotype control (data not shown). Comparison of R1 mature macrophages from normoxia and hypoxia (Fig 5.4 Plots C and F) with R2 and R3 regions, demonstrated a statistically significant overall increase in versican protein expression. Similar hypoxia-induced increases in
versican mean fluorescent intensity were also observed in the monocyte/macrophages in regions R2 and R3. Interestingly, very low levels of versican fluorescent intensity, and no hypoxic induction, were detected in the population assumed to contain the majority of lymphocytes (R4; Fig 5.4 plots C and F) however, histogram-plot analysing of cells in R4 showed few strongly positive cells (data not shown). As expected, this matches well with the low versican mRNA level, and lack of hypoxic induction, which was found in CD14⁻ cells (lymphocytes) (Fig 4.6). Taken together, this data suggests that hypoxic up-regulation of versican protein is confined to monocyte/macrophage lineage cells and is not detectable in lymphocytes.



Figure 5.4: Intracellular flow cytometry analysis of hypoxic regulation of versican protein in human PBMC. PBMCs were subjected to flow cytometry to analyse the effect of 5 days hypoxia on versican protein expression. 2 x 10⁶ cells per well were cultured under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 4 days followed by brefeldin A treatment and incubated under the same oxygen tension for a further 24 hours. Normoxic and hypoxic PBMCs were plotted onto a dot-scatter plot (A and D). Monocyte/macrophages are subdivided into 3 regions R3-R1 in respect of increasing cell size (forward scatter, FSC). Lymphocytes are included in region 4 (A and D). Dot to calculate percentage scatter plots were used the of the total monocyte/macrophages population present in regions R1, R2, and R3 in normoxia (B) or hypoxia (E). Monocyte/macrophage regions R3-R1 and lymphocytes region (R4) were gated onto the histogram-plot and then examined for normoxic and hypoxic versican fluorescent intensity (C and F) against isotype control. Isotype control levels were subtracted from the data. Data are from 5 independent experiments and were further analysed using paired two-tailed t-tests,* = p < 0.05

5.5 Summary

In this chapter, the level of versican protein expression was quantified in normoxic or hypoxic PBMC after 24 hours and 5 days incubation using intracellular staining followed by flow cytometry. Preliminary investigations showed the role of brefeldin A in blocking the protein secretion into the culture medium when the results from untreated PBMC (Fig 5.1) were compared to PBMC treated with brefeldin A (Fig 5.2). Flow cytometry analysis of PBMCs treated with brefeldin A followed 24 hours incubation under normoxia or hypoxia did not show any increase in versican fluorescent intensity in hypoxia in comparison with normoxia (Fig 5.3 plots B and D) whereas further investigation demonstrated significant induction of versican protein expression (3-fold) in monocyte/macrophages after 5 days hypoxia (n=5) (Fig 5.2, graph E). Also further investigation on PBMCs after 5 days hypoxia demonstrated that increasing monocyte/macrophage cell size (forward scatter) on the dot scatter plot is associated with increasing versican fluorescent intensity in normoxia and hypoxia (Fig 5.4, plot A, D, C and F). In addition, as figures 5.4, plots C and F show, no hypoxic induction of versican protein was detected in region which contains the lymphocyte population (R4) although a very low level of versican mean fluorescent intensity was observed in both normoxia and hypoxia. The data suggested that versican protein expression is largely confined to monocyte/macrophage and is not detectable in lymphocytes.

Chapter 6: Results

Versican promoter analysis

6.1 Introduction

The versican promoter which was characterised by Naso et al., (1994), contains a typical TATA box located approximately 16 base pairs upstream of the transcription start site and several putative regulatory sites for a number of transcription factors such as AP-2 and SP-1 (Fig 1.6). Furthermore, a reporter construct driven by an 876-bp (-632/+240 relative to the transcriptional start site) piece of the versican promoter showed significant expression in HeLa cells and IMR-90 embryonic lung fibroblasts. Deletion constructs revealed enhancer and repressor elements on the 876-bp of versican promoter which were responsible for the basal expression in different cells (Naso et al., 1994). In addition, oligonucleotide arrays on human carcinoma cell lines such as MCF7, suggested a role for the Pax3 and P53 transcription factors in versican regulation and expression (Mayanil et al., 2001; Yoon et al., 2002). In addition, two putative binding sites for the TCF/LEF which are localized at position -546-bp and -492bp on the versican promoter, are revealed by Rahmani et al., (2005) (Fig 1.6). Sitedirected mutagenesis of the TCF sites in the versican promoter markedly reduced reporter luciferase activity in smooth muscle cells (SMC) and further investigation confirmed the role of the β -catenin / TCF transcription factor complex in expression of versican in SMC (Rahmani et al., 2005). In addition, studies in human melanoma cells identified several transcriptional regulatory elements in a 620-bp (-618/+2) versican promoter construct, including SP-1, AP-2 and AP-1 binding sites (Domenzain et al., 2009). Further investigation using mutagenesis of the AP-1 site at position -36 suggested that AP-1 is the main regulatory factor which directs versican production during melanoma progression (Domenzain et al., 2009).

In this project, promoter reporter constructs were used to determine the parts of the versican promoter responsible for the hypoxic up-regulation of versican in human primary macrophages. A 2.5-Kb piece of DNA containing the versican promoter was initially cloned into pGL3-basic in Dr Burke's lab prior to this project, and during this project I produced a series of versican promoter constructs including 4.140-Kb (-3.956kb/+184bp), 830-bp (-646bp/+184bp), 480-bp (-296bp/+184bp) and 240-bp (-56bp/+184bp) which were cloned into the more modern, improved pGL4.10 luciferase reporter vector. These constructs were then transfected into HMDM and tested for inducibility by hypoxia. In addition, to determine which parts of the proximal promoter are required for increased expression in hypoxia, a series of versican promoter deletion constructs were made and transfected into HMDM. Also sequence analysis was carried out using Genomatix MatInspector and TESS software to search for putative binding sites for hypoxia-inducible transcription factors.

6.2 PCR amplification of different regions of versican promoter

To determine the DNA sequences which are required for hypoxic expression of versican in HMDM, the different segments of versican promoter were amplified from human genomic DNA first by PCR and then were cloned into the pGL4.10 luciferase reporter plasmid (See section 3.3 for general molecular cloning and table 2.12 for primers).

A 4.140-Kb (-3.956kb/+184bp) region of the versican promoter was PCR-amplified using -3.956kb versican-5' (forward) and +184bp versican-3' (reverse) primers and then run on a 0.6% agarose gel against a 1-Kb DNA molecular weight ladder to confirm the correct size of DNA fragment is amplified (Fig 6.1).



Figure 6.1: 0.6% agarose gel electrophoresis of PCR amplification of the 4.140-Kb (- 3.956kb/+184bp) versican promoter sequence from human genomic DNA. Lane 1 = 1-kb DNA ladder, lane 2 = 4.140-Kb

In addition, shorter sections of the versican promoter including 830-bp (-646bp/+184bp), 480-bp (-296bp/+184bp) and 240-bp (-56bp/+184bp) were PCRamplified and run on a 1.5% agarose gel. The sizes of the amplified segments of the versican promoter were confirmed against a 100-bp DNA molecular weight ladder (Fig 6.2).



Figure 6.2: 1.5% agarose gel electrophoresis of PCR amplification of the 240-bp (-56bp/+184bp), 480-bp (-296bp/+184bp) and 830-bp (-646bp/+184bp) versican promoter sequences from human genomic DNA. Lane 1 = 100-bp DNA ladder, Lane 2 = 240-bp, Lane 3 = 480-bp, Lane 4 = 830-bp.

In order to define the DNA sequence of the versican promoter responsible for hypoxic expression in HMDM, a deletion series of the 240-bp (-56bp/+184bp) of versican proximal promoter was generated, including, 210-bp (-56bp/+154bp), 160-bp (-56bp/+104bp), (-56bp/+54bp), 110-bp 210-bp (-26bp/+184bp), 180-bp (-26bp/+154bp), 130-bp (-26bp/+104bp), 80-bp (-26bp/+54bp). As figure 6.3 shows, two amplified 210-bp (-56bp/+154bp and -26bp/+184bp), 180-bp, 160-bp, 130-bp, 110-bp and 80-bp of versican promoter sequences were successfully PCR-amplified and run on a 2.0% agarose gel and compared to a DNA molecular weight ladder (HyperLadder™ V, Bioline) to confirm that the correct fragments of the versican promoter have been generated (Fig 6.3).



Figure 6.3: 2% agarose gel electrophoresis of PCR amplification of the 110-bp (-56bp/+54bp), 160-bp (-56bp/+104bp), 210-bp (-56bp/+154bp), 80-bp (-26bp/+54bp), 130-bp (-26bp/+104bp), 180bp (-26bp/+154bp) and 210-bp (-26bp/+184bp) versican promoter sequences from human genomic DNA. Lane 1 = HyperLadder V, Lane 2 = 110-bp, Lane 3 = 160-bp, Lane 4 = 210-bp, Lane 5 = 80-bp, Lane 6 = 130-bp, Lane 7 = 180-bp and Lane 8 = 210-bp

In addition, a 29-bp random nucleotide sequence (CTAGCTAGCTAGCTAGCTAGCTAGCTAGC) was cloned into the pGL4.10 luciferase reporter plasmid for use as a negative control in transfection experiments. The annealed 29-bp random sequence was loaded into a 2.0% agarose gel and the size was confirmed in comparison with a DNA molecular weight ladder (HyperLadder[™] V, Bioline) (Figure 6.4).



Figure 6.4: 2% agarose gel electrophoresis of the annealed 29-bp random nucleotide sequence (CTAGCTAGCTAGCTAGCTAGCTAGCTAGCTAGC). Lane 1 = HyperLadder V, lane 2 = 29-bp

6.3 Cloning and screening of different fragments of versican promoter

The successfully amplified different regions of the versican promoter, and the pGL4.10 [luc2] vector, were digested by the *Sfil* restriction enzyme isolated from an E. coli strain that carries the *Sfil* gene from *Streptomyces fimbriatus* (See section 2.7.1 and 3.3.3). The digested versican promoter DNA sequences and pGL4.10 vector were run on agarose gels and extracted with the QIAquick Gel Extraction Kit, and the concentration estimated by comparison with known standards in ethidium bromide-stained agarose gels. Purified DNA fragments were cloned into the pGL4.10 vector and DNA sequencing was performed to confirm the sequences of the different fragments cloned (See section 3.3 for general molecular cloning).

In order to confirm that a positive 4.140-Kb construct had been generated, all three putative constructs obtained were digested using *Sfil* and run on a 0.6% agarose gel beside an undigested 4.140-Kb construct to determine if the correct size insert has been cloned (Fig 6.5). As figure 6.5 indicates, all 3 digested 4.140-Kb versican promoter pGL4.10 constructs showed the correct insert size at 4-Kb in comparison with DNA ladder. Due to the fact that the insert and vector are similar in size, an incomplete separation of 4.140-Kb versican promoter and pGL4.10 vector was observed (Fig 6.5).

6. <u>Results</u>



Figure 6.5: 0.6% agarose gel electrophoresis of restriction digested plasmid preps of **4.140-Kb** of versican promoter pGL4.10 constructs. Three putative purified 4.140-Kb versican pGL4.10 clones were digested with *Sfil* and run with an undigested 4.140-Kb construct. Lane 1 = 1-kb DNA ladder, Lane 2 = undigested 4.140-Kb versican pGL4.10 construct, Lane 3 = digested 4.140-Kb versican pGL4.10 construct, Lane 4 = undigested 4.140-Kb versican pGL4.10 construct, Lane 6 = undigested 4.140-Kb versican pGL4.10 construct, Lane 7 = digested 4.140-Kb versican pGL4.10 construct

In addition, to confirm that the 830-bp, 480-bp and 240-bp fragments of the versican promoter had been successfully cloned into the pGL4.10 plasmid, three clones of each construct were digested using *Sfil* and run on 1.5% agarose gels beside undigested constructs. As figure 6.6 indicates, of three putative clones of the 480-bp versican pGL4.10 construct, only one was digested by *Sfil* and showed the correct size insert. Also all 3 selected clones of the 240-bp (Fig 6.7) and 830-bp (Fig 6.8) versican pGL4.10 constructs showed the correct insert size when they were digested by *Sfil* and compared to undigested constructs and DNA ladder.

6. <u>Results</u>



Figure 6.6: 1.5% agarose gel electrophoresis of restriction digested plasmid preps of **480-bp of versican promoter pGL4.10 constructs**. Three putative purified 480-bp versican pGL4.10 clones were digested with *Sfil* and run with an undigested 480-bp construct. Lane 1 = 1-Kb DNA ladder, Lane 2 = digested 480-bp versican pGL4.10 construct (No insert), Lane 3 = undigested 480-bp versican pGL4.10 construct, Lane 4 = digested 480-bp versican pGL4.10 construct, Lane 6 = digested 480-bp versican pGL4.10 construct (No correct insert), Lane 7 = undigested 480-bp versican pGL4.10 construct, Lane 8 = 100-bp DNA ladder



Figure 6.7: 1.5% agarose gel electrophoresis of restriction digested plasmid preps of **240-bp versican promoter pGL4.10 constructs**. Three putative purified 240-bp versican clones were digested with *Sfil* and run with an undigested 240-bp construct. Lane 1 = 100-bp DNA ladder, Lane 2 = digested 240-bp versican pGL4.10 construct, Lane 3 = undigested 240-bp versican pGL4.10 construct, Lane 4 = digested 240-bp versican pGL4.10 construct, Lane 6 = digested 240-bp versican pGL4.10 construct, Lane 7 = undigested 240-bp versican pGL4.10 construct, Lane 8 = 1-kb DNA ladder

6. <u>Results</u>



<u>Figure 6.8:</u> 1.5% agarose gel of restriction digested plasmid preps of 830-bp versican promoter pGL4.10 constructs. Three putative purified 830-bp versican clones were digested with *Sfil* and run with an undigested 830-bp construct. Lane 1 = 1-Kb DNA ladder, Lane 2 = undigested 830-bp versican pGL4.10 construct, Lane 3 = digested 830bp versican pGL4.10 construct, Lane 4 = undigested 830-bp versican pGL4.10 construct, Lane 5 = digested 830-bp versican pGL4.10 construct, Lane 6 = undigested 830-bp versican pGL4.10 construct, Lane 7 = digested 830-bp versican pGL4.10 construct

Due to detection of normoxic and hypoxic luciferase activity in the 240-bp of versican proximal promoter construct, further work was carried out to generate a deletion series of the 240-bp, by inserting shorter sequences of versican promoter into the pGL4.10 vector. Three putative clones of the 110-bp versican pGL4.10 construct (Fig 6.9) were analysed by *Sfil* restriction digestion and run on a 2.0% agarose gel beside an undigested 110-bp construct to confirm successful cloning. As figure 6.9 shows, only two digested purified DNA-plasmid constructs showed a band of the correct size. Also three and two selected clones of the 160-bp and 210-bp (-56bp/+154bp) constructs showed one and two positive clones respectively on 2.0% agarose gels (Fig 6.9). All digested versican promoter pGL4.10 constructs were examined against the same

undigested constructs alongside 1-Kb and 100-bp DNA ladder to confirm the correct size of insert.



Figure 6.9: 2% agarose gel electrophoresis of restriction digested plasmid preps of 110-bp, 160-bp and 210-bp (-56bp/+154bp) versican promoter pGL4.10 constructs. Three putative purified 110-bp and 160-bp clones and two purified 210-bp clones digested with Sfil and run with an undigested relative construct. Lane 1 = 1-Kb DNA ladder, Lane 2 = undigested 110-bp versican pGL4.10 construct, Lane 3 = digested 110bp versican pGL4.10 construct (No insert), Lane 4 = undigested 110-bp versican pGL4.10 construct, Lane 5 = digested 110-bp versican pGL4.10 construct, Lane 6 = undigested 110-bp versican pGL4.10 construct, Lane 7 = digested 110-bp versican pGL4.10 construct, Lane 8 = 100-bp DNA ladder, Lane 9 = undigested 160-bp versican pGL4.10 construct, Lane 10 = digested 160-bp versican pGL4.10 construct (No insert), Lane 11 = undigested 160-bp versican pGL4.10 construct, Lane 12 = digested 160-bp versican pGL4.10 construct (No insert), Lane 13 = undigested 160-bp versican pGL4.10 construct, Lane 14 = digested 160-bp versican pGL4.10 construct, Lane 15 = 100-bp DNA ladder, Lane 16 = undigested 210-bp versican pGL4.10 construct, Lane 17 = digested 210-bp versican pGL4.10 construct, Lane 18 = undigested 210-bp versican pGL4.10 construct, Lane 19 = digested 210-bp versican pGL4.10 construct, Lane 20 = 100-bp DNA ladder

Also 4 selected clones of the 80-bp versican promoter pGL4.10 construct were analysed by *Sfil* restriction digestion against undigested 80-bp versican promoter construct on a 1.5% agarose gel. As figure 6.10 shows, three out of four digested 80-bp versican promoter pGL4.10 construct were positive and showed the correct insert size. Analysis of four putative clones of the 130-bp, 180-bp constructs (Fig 6.11) and three putative clones of the 210-bp (-26bp/+184bp) of versican promoter pGL4.10 construct (Fig 6.12) showed three, two and three positive clones, respectively, in comparison with undigested constructs and DNA ladders.



Figure 6.10: 2% agarose gel electrophoresis of restriction digested plasmid preps of **80-bp versican pGL4.10 construct**. Four putative purified 80-bp clones digested with *Sfil* and run with an undigested 80-bp construct. Lane 1 = 1-Kb DNA ladder, Lane 2 = undigested 80-bp versican pGL4.10 construct, Lane 3 = digested 80-bp versican pGL4.10 construct (No insert), Lane 4 = undigested 80-bp versican pGL4.10 construct, Lane 5 = digested 80-bp versican pGL4.10 construct, Lane 6 = undigested 80-bp versican pGL4.10 construct, Lane 7 = digested 80-bp versican pGL4.10 construct, Lane 8 = undigested 80-bp versican pGL4.10 construct, Lane 9 = digested 80-bp versican pGL4.10 construct, Lane 10 = 100-bp DNA ladder



Figure 6.11: 2% agarose gel electrophoresis of restriction digested plasmid preps of 130-bp and 180-bp versican pGL4.10 construct. Four putative purified 130-bp and 180-bp versican pGL4.10 clones digested with Sfil and run with an undigested relative construct. Lane 1 = 100-bp DNA ladder, Lane 2 = undigested 130-bp versican pGL4.10 construct, Lane 3 = digested 130-bp versican pGL4.10 construct (No insert), Lane 4 = undigested 130-bp versican pGL4.10 construct, Lane 5 = digested 130-bp versican pGL4.10 construct, Lane 6 = undigested 130-bp versican pGL4.10 construct, Lane 7 = digested 130-bp versican pGL4.10 construct, Lane 8 = undigested 130-bp versican pGL4.10 construct, Lane 9 = digested 130-bp versican pGL4.10 construct, Lane 10 = 100-bp DNA ladder, Lane 11 = undigested 180-bp versican pGL4.10 construct, Lane 12 = digested 180-bp versican pGL4.10 construct (No insert), Lane 13 = undigested 180-bp versican pGL4.10 construct, Lane 14 = digested 180-bp versican pGL4.10 construct, Lane 15 = undigested 180-bp versican pGL4.10 construct, Lane 16 = digested 210-bp versican pGL4.10 construct (No insert), Lane 17 = undigested 180-bp versican pGL4.10 construct, Lane 19 =digested 180-bp versican pGL4.10 construct, Lane 19 = 100-bp DNA ladder



Figure 6.12: 2% agarose gel electrophoresis of restriction digested plasmid preps of **210-bp (-26bp/+184bp) versican pGL4.10 construct**. Three putative purified 210-bp versican pGL4.10 clones digested with *Sfil* and run with an undigested 210-bp construct. Lane 1 = 100-bp DNA ladder, Lane 2 = undigested 210-bp versican pGL4.10 construct, Lane 3 = digested 210-bp versican pGL4.10 construct, Lane 4 = undigested 210-bp versican pGL4.10 construct, Lane 6 = undigested 210-bp versican pGL4.10 construct, Lane 7 = digested 210-bp versican pGL4.10 construct, Lane 7 = digested 210-bp versican pGL4.10 construct.

In order to generate a random pGL4.10 construct, 3 putative clones were digested using *Sfil* and loaded onto a 2.0% agarose gel. Digested random pGL4.10 constructs were compared to undigested construct in addition to 100-bp DNA ladder. Weak bands indicate that three positive clones had been obtained (Fig. 6.13).



Figure 6.13: 2% agarose gel electrophoresis of restriction digested plasmid preps of random pGL4.10 construct. Three putative purified 29-bp random pGL4.10 clones digested with *Sfil* and run with an undigested random construct. Lane 1 = HyperLadder V, Lane 2 = undigested 29-bp random pGL4.10 construct, Lane 3 = digested 29-bp random pGL4.10 construct, Lane 5 = digested 29-bp random pGL4.10 construct, Lane 5 = digested 29-bp random pGL4.10 construct, Lane 6 = undigested 29-bp random pGL4.10 construct, Lane 7 = digested 29-bp random pGL4.10 construct

6.4 Transfection optimisation using JetPEI[™] in HMDM

As pilot experiments, several transfections using a 2.5-Kb versican promoter pGL3basic luciferase reporter construct kindly provided by Dr Burke were carried out on HMDM. Cells were transfected with this firefly luciferase expressing reporter plasmid and co-transfected with the pRL-TK plasmid (Promega) which contains a Renilla luciferase reporter gene and serves as an internal control for normalization of transfections (See section 3.5 for transfection).

Five-day old HMDM (prepared as described in section 3.1.1) were cultured at a cell density of 2×10^6 per well in, a volume of 2ml in 6-well adherent Nunclon plates. Cells were transfected with 2 µg, 1.5 µg or 1 µg of 2.5-Kb versican promoter pGL3-basic construct and co-transfected with either 20 ng, 15 ng or 25 ng of pRL-TK plasmid, giving firefly luciferase to Renilla luciferase construct ratios of 1:100, 1:100, and 1:40, respectively, using JetPEI DNA transfection reagent (PolyPlus) (See section 3.5.1). Cells were then incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 24 hours prior to luciferase assay (See section 3.5.3).

HMDM transfected with 2 µg versican promoter-plasmid/20 ng pRL-TK using 6.4 µl JetPEI reagent (using one well per transfection), expressed firefly luciferase reporter activity after 24 incubation under normoxia and hypoxia in three independent experiments with different donors (Table 6.1). Although the level of firefly luciferase expression was higher in hypoxia (overall 1.9-fold), the level of internal control Renilla luciferase expression in all three experiments was too low (in many cases below the background level obtained using only LAR II and Stop & Glo reagent, in the absence of

cell lysate (See section 3.5.3)) to be reliable. In further optimisation experiments, HMDM transfected with 1.5 μ g of versican promoter-plasmid/15 ng pRL-TK using 4.8 µl JetPEI reagent (using one well per transfection) expressed versican luciferase reporter construct using three different donors in independent experiments, although no luciferase increase was observed in hypoxic HMDM. In addition, the level of Renilla luciferase expression was also close to the background level (Table 6.1). To optimise the level of Renilla luciferase expression, further experiments in HMDM from three different donors, using 1 µg DNA promoter-plasmid/25 ng pRL-TK with 3.2 µl JetPEI reagent (using one well per transfection), showed very high luciferase expression in normoxia and hypoxia, and Renilla luciferase levels well above background. As table 6.1 shows, although the level of firefly luciferase using 1 μ g of versican promoterplasmid/25 ng pRL-TK was higher in hypoxia (average 2-fold), the level of Renilla luciferase was also increased by hypoxia in all three independent experiments. Overall the pRL-TK plasmid either failed to show signals that could be distinguished from background or was affected by hypoxia, so it was not useful for normalization. In an attempt to circumvent the Renilla normalization problem, the Thermo Scientific Pierce 660 nm protein assay was used to normalize in some transfections, which is not a perfect solution but at least allowed correction for the number of cells in each well.

2.5-Kb versican promoter pGL3 construct (μg)	pRL-TK plasmid (ng)	JetPEI DNA transfection reagent (µl)	Firefly Luciferase (RLU/s)		Renilla Luciferase (RLU/s)		Background (RLU/s)
			20.9% O ₂	0.2% O ₂	20.9% O ₂	0.2% O ₂	
			24 hrs	24 hrs	24hrs	24hrs	
2	20	6.4	2828	4588	414	417	401
2	20	6.4	6224	14543	369	384	398
2	20	6.4	9042	17474	454	496	445
1.5	15	4.8	1766	796	383	388	358
1.5	15	4.8	4274	3892	343	368	281
1.5	15	4.8	1161	1284	464	422	342
1	25	3.2	46487	113097	2600	6659	311
1	25	3.2	7840	19366	1844	3508	265
1	25	3.2	7295	10823	611	13292	301

Table 6.1: Transfection optimization in HMDM using different amounts of 2.5-Kb versican promoter pGL3-basic construct and JetPEI DNA transfection reagent. HMDM were incubated under normoxia for 5 days and co-transfected with different amounts of 2.5-Kb versican promoter pGL3-basic construct and the internal control pRL-TK Renilla luciferase plasmid using different amounts of JetPEI DNA transfection reagent. Transfected cells were incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 24 hours prior to luciferase assay. Renilla luciferase values were compared to the background RLU obtained using only LAR II and Stop & Glo reagent, in the absence of cell lysate. Data are from 3 independent experiments using different donors and experiments were performed in triplicate.

Taken as a whole, the initial experiments on the 2.5-Kb versican promoter pGL3-basic luciferase reporter construct confirmed the basal promoter activity of this construct and showed hypoxia-induced increases (2-fold). These findings led me to further dissect the versican promoter in order to try to determine the DNA sequences which are required for the expression and hypoxic regulation of versican in HMDM, by making a series of versican reporter constructs including 4.140-Kb (-3.956Kb/+184bp), 830-bp (-646bp/+184bp), 480-bp (-296bp/+184bp) and 240-bp (-56bp/+184bp) which were generated and cloned into the pGL4.10 reporter vector via the *Sfil* restriction site.

6.5 Effect of 24 hours hypoxia on versican promoter reporter constructs expression in HMDM

Based on the data from the initial optimisation experiments described in section 6.4, five-day old HMDM which were cultured at a cell density of 2×10^6 per well, were transfected with 1 µg of versican reporter constructs and co-transfected with 25 ng pRL-TK using 3.2 µl of JetPEI reagent (using one well per transfection). Cells were then incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 24 hours prior to luciferase assay (See section 3.5).

In these experiments, the Renilla luciferase expression by the internal control plasmid pRL-TK was unfortunately not high enough above background to be useful. Therefore the "raw" (non-normalised) firefly luciferase values were plotted (Fig 6.14). Negative control promoterless plasmids, pGL3-basic and pGL4.10 (Promega), did not show luciferase expression that could be distinguished from background (data not shown).

As figure 6.14 shows, 24 hours hypoxia did not induce luciferase activity in HMDM transfected with the 4.140-Kb pGL4.10 or 480-bp pGL4.10 constructs, whereas cells transfected with 2.5-Kb pGL3-basic or 830-bp pGL4.10 constructs showed hypoxia-increased luciferase expression in addition to the 240-bp pGL4.10, the shortest versican promoter reporter construct. As expected, hypoxia significantly induced the level of luciferase expression in HMDM transfected with PGK, a hypoxia-inducible positive control luciferase reporter construct (Ameri et al., 2002). Although little hypoxia inducibility was observed after 24 hours with any construct, this data indicated that all the versican promoter constructs were functional.



<u>Figure 6.14</u>: Effect of 24 hours hypoxia on versican promoter reporter construct activity in HMDM. HMDM incubated under normal oxygen condition for 5 days, and transfected with 1 µg of versican reporter construct DNA and 25 ng of pRL-TK Renilla normalization plasmid using 3.2 µl of JetPEI reagent according to the manufacturer's protocol. Cells were incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for a further 24 hours prior to luciferase assay. Luciferase raw value data from minimum 3 independent experiments using different donors are expressed as means ± SEM. Data were further analysed for significant induction using paired two-tailed t-tests.* = p <0.05 when compared to the normoxic PGK.

6.6 Effect of 5 days hypoxia on versican promoter reporter constructs expression in HMDM

Preliminary investigations by Staples and Burke et al., (unpublished data) had shown that versican mRNA is up-regulated much more highly by long term hypoxia (5 days) than by short term hypoxia in PBMC (Fig 1.8). In addition, my early investigation on versican protein, showed hypoxic induction of versican protein level in macrophages after 5 days (Fig 5.2, graph E). Therefore, the effect of 5 days hypoxia was investigated on HMDM transfected with versican promoter reporter constructs in addition to the PGK positive control plasmid (Fig. 6.15). Negative controls (pGL3-basic and pGL4.10 [luc2]) were also used and showed negligible expression similar to background even after 5 days (data not shown). Five-day old HMDM which were cultured at a cell density of 2×10^6 per well, were transfected with 1 µg of versican reporter constructs and co-transfected with 25 ng pRL-TK using 3.2 µl of JetPEI reagent (using one well per transfection) (See section 3.5). Cells were then incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 5 days prior to luciferase assay.

As expected 5 days hypoxia significantly induced PGK luciferase expression in HMDM (Fig 6.15). In addition, all versican promoter reporter constructs including 4.140-Kb pGL4.10, 2.5-Kb pGL3-basic, 830-bp pGL4.10 and 480-bp pGL4.10 showed hypoxiainduced luciferase activity after 5 days. As figure 6.15 indicates, also 5 days hypoxia markedly induced 240-bp pGL4.10 (2.7-fold), the shortest versican reporter construct which determined me for further investigation on the versican promoter in order to elucidate the DNA sequences which are required for the hypoxic activation of versican in HMDM. Although hypoxia inducibility was observed after 5 days with any construct (Fig 6.15), overall these data confirmed that the all versican promoter luciferase constructs are functional however the data could not be normalised by internal control due to very low level of Renilla luciferase expression from the co-transfected pRL-TK plasmid.



<u>Figure 6.15</u>: Effect of 5 days hypoxia on versican promoter reporter construct activity in HMDM. HMDM were generated from 2×10^6 per well PBMC in 6-well adherent Nunclon plates, incubated under normal oxygen condition for 5 days, and transfected with 1 µg of versican reporter construct DNA and 25ng of pRL-TK Renilla normalization plasmid using 3.2 µl of JetPEI reagent according to the manufacturer's protocol. Cells were incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for a further 5 days prior to luciferase assay. Luciferase raw value data from minimum 3 independent experiments using different donors are expressed as means ± SEM. Data were further analysed for significant induction using paired one-tailed t-tests.* = p <0.05 when compared to the normoxic PGK.

To address the problem of poor Renilla luciferase expression levels, for further transfection experiments, luciferase values were normalized to the protein concentrations obtained using the Thermo Scientific Pierce 660 nm protein assay. Also, in order to minimise variability from cell plating, pipetting inconsistencies, and toxicity during transfection, PBMCs were first transfected and then split equally into adherent plates which were incubated under normoxia or hypoxia for 5 days prior to luciferase assay (See section 3.5). In addition, to solve the problem of the negative controls expressing too little luciferase to be detected, a 29-bp random nucleotide sequence cloned into pGL4.10 to use as a negative control in further transfection experiments (See section 3.3), with the intention of producing detectable signals. In addition, sequence analysing using Genomatix MatInspector and TESS software did not show any putative transcription factor binding sites within the 29-bp random nucleotide sequence.

6.7 Expression of versican promoter reporter constructs in hypoxia

As described above, high levels of luciferase activity were detected after 5 days with all the versican promoter constructs analysed, including the shortest, the 240-bp (-56bp/+184bp) construct which was markedly induced by hypoxia (2.7-fold) (Fig 6.15). Therefore, to further elucidate the DNA sequences responsible for hypoxic activity of the versican promoter in HMDM a deletion series was made to dissect the putative transcription factor binding sites (Fig 6.17) within the -56bp/+184bp construct.

Shorter derivatives including 210-bp (-56bp/+154bp), 210-bp (-26bp/+184bp), 180-bp (-26bp/+154bp), 160-bp (-56bp/+104bp), 130-bp (-26bp/+104bp), 110-bp (-

56bp/+54bp) and 80-bp (-26bp/+54bp), were generated by PCR from human genomic DNA with the appropriate sets of primers and cloned into the pGL4.10 [luc2] using standard techniques. In order to transfect PBMC, cells were isolated (See section 3.1.1) and seeded in 4ml complete Iscove's medium at a cell density of 4×10⁶ per well in 6-Well Costar ultra low attachment plates and incubated at 37°C with 5% CO₂ and 20.9% O₂ for 5 days to allow differentiation of monocytes to macrophages. After 5 davs incubation, 4×10^6 PBMCs were transfected with 2 µg of versican promoter reporter construct DNA followed by 6.4 µl of JetPEI reagent (using two well per transfection) and were plated in 2 ml at a cell density of 2×10⁶ per well in 6-Well Nunclon adherent plates and returned to 37° C with 5% CO₂ and 20.9% O₂ for 1 hour followed by incubation under either normoxia (20.9% O_2) or hypoxia (0.2% O_2) for 5 days (See section 3.5.1). After 5 days, medium was removed and cells were washed twice with 1x PBS, lysed, and luciferase activities were quantified using luciferase assay reagents according to the manufacturer's protocol (See section 3.5.3). In addition, protein concentrations were measured using the Thermo Scientific Pierce 660 nm protein assay and luciferase values were normalized to the obtained protein concentrations (See section 3.5.4).

As figure 6.16 indicates, versican promoter reporter construct -56bp/+184bp (240-bp) in addition to those carrying deletions at the 3' end of the 240-bp starting sequence, containing the sequences -56bp/+154bp (210-bp) and -56bp/+104bp (160-bp), showed high levels of luciferase expression in both normoxia and hypoxia after 5 days. However, luciferase activity was markedly lower in the -56bp/+54bp (110-bp) construct which additionally showed significant hypoxia-induced luciferase expression

in comparison with the random construct. Overall, these data indicate that the region +54 to +104 is important for high level expression in both normoxia and hypoxia in HMDM, although hypoxic induction was still observed with the -56bp/+54bp construct, suggesting that the region +54 to +104 is not the only region involved in hypoxic inducibility (Fig 6.16).

Next, a series of 5' deletion constructs of the 240-bp (-56bp/+184bp) versican promoter was also generated containing the sequences -26bp/+184bp (210-bp), - 26bp/+154bp (180-bp), -26bp/+104bp (130-bp) and -26bp/+54bp (80-bp) and transfected into HMDM followed by 5 days incubation under normoxia and hypoxia. As figure 6.16 shows, although hypoxia activity was significantly lower in the shortest versican reporter construct -26bp/+54bp in comparison with -56bp/+54bp, transfection analysis data demonstrated that luciferase expression is markedly lower in the constructs beginning at -26 compared to that beginning at -56, suggesting that the sequence from -56 to -26 encompasses a region required for high level activity of these versican promoter reporter constructs in HMDM. In addition, as figures 6.16 shows, although normoxic and hypoxic expression of luciferase was observed in the 29-bp random negative control construct, the luciferase activity was significantly lower than the versican promoter constructs.



Figure 6.16: Effect of 5 days normoxia and hypoxia on 240-bp (-56bp/+184bp) and deleted versican promoter reporter construct in HMDM. PBMCs were transfected with deleted versican promoter or 29-bp negative control random pGL4.10 constructs to assess the level of luciferase activity in normoxic and hypoxic HMDM. 4×10^6 PBMCs were transfected with 2 µg of versican reporter or 29-bp random constructs and plated in 2 ml at a cell density of 2×10^6 per well in 6-Well Nunclon adherent plates. Cells were then incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 5 days prior to luciferase assay. Data were normalised to the obtained protein concentrations. Data from minimum 6 independent experiments each done in triplicate using cells from different donors are expressed as means ± SEM. Data were further analysed for significant induction using paired two-tailed t-tests.* = p <0.05 when compared to the random construct in hypoxia.

In addition as figure 6.17 indicates, analysis of the 240-bp versican promoter sequence using Genomatix MatInspector and TESS software revealed binding sites for a number of transcription factors which are known to be hypoxia-inducible: CREB (Leonard et al., 2008) and AP-1 at -34, AP-1 at -4 (Uenoyama et al., 2006), SP1 at +25 (Xu et al., 2000; Szalad et al., 2009), E2F at +80 (O'Connor and Lu, 2000; Cummins and Taylor, 2005) and a putative binding site at +60 for the master regulator of hypoxia-inducible gene expression, the transcription factor HIF-1 (Wenger et al., 2005).



Figure 6.17: Putative transcription factor binding sites within the 240bp (-56+184) versican promoter sequence. Genomatix MatInspector and TESS software revealed putative transcription binding sites for hypoxia inducible factor (HIF), cAMP responsive element binding (CREB), activator Protein 1 (AP1), SP1, nuclear factor 1 (NF-1) and E2F within the 240bp (-56+184) versican promoter sequence. The TATA box and transcription start site were identified by Naso et al., (1994).

Overall the data first showed that region +54 to +104, which contains a putative HIF-1 binding site (Fig 6.17), is crucial for high level versican promoter activity in hypoxia in HMDM (Fig 6.16). In addition, constructs which lack the -56 to -26 region showed reduced hypoxia activity suggesting that transcription factors which bind in this region are important for versican promoter activity in HMDM. Thus, the data show that both the -56 to -26 and +54 to +104 regions are required together in the same construct for high expression from the versican promoter in hypoxic HMDM (Fig 6.16).

6.8 Summary

In this chapter, in order to determine the DNA sequences of the versican promoter which are responsible for hypoxic regulation, a 2.5-Kb versican promoter pGL3-basic construct (provided by Dr B. Burke) and a series of versican promoter pGL4.10 constructs including 4.140-Kb, 830-bp, 480-bp and 240-bp (made during this project) were transfected into HMDM. Initial transfections using the 2.5-Kb versican promoter pGL3-basic construct showed detectable levels of firefly luciferase in HMDM which were transfected with 2 µg or 1.5 µg of versican promoter construct and cotransfected with 15ng or 20ng of pRL-TK plasmid DNA respectively (Table 6.14). However, the level of Renilla luciferase expressed from the normalisation control pRL-TK plasmid was so close to the background as to make the data of no use for normalization. Further experiments using 1 µg of versican promoter pGL3-basic construct and 25 ng of pRL-TK plasmid showed increased versican luciferase expression in hypoxia (2-fold) in comparison with normoxia with a level of Renilla luciferase expression which was above background however it showed hypoxia induced increases (Table 6.14). Based on these preliminary investigations on the 2.5-

Kb versican promoter pGL3-basic luciferase reporter construct, further study analysing the raw value of luciferase expression by a series of versican promoter constructs showed basal luciferase activity in the 4.140-Kb, 830-bp, 480-bp and 240-bp versican promoter reporter constructs and confirmed that they are all functional after 24 hours or 5 days exposure to normoxia and hypoxia (Fig 6.14 and 6.15). But these data could not be normalised against an internal control due to the problematic hypoxic induction of luciferase expression by the pRL-TK plasmid. Therefore in further transfection experiments, luciferase values were normalized to the obtained protein concentrations using a protein assay. In addition, as figure 6.14 and 6.15 showed, hypoxia significantly induced PGK luciferase expression, a hypoxia-inducible positive control in HMDM whereas the level of luciferase expressed by negative controls reporter plasmids, pGL3-basic and pGL4.10 [luc2] was undetectable (data not shown). In order to have a negative control which produces detectable luciferase activity, a 29bp random construct was transfected into HMDM.

Based on initial transfection experiments using a 240-bp construct (-56bp/+184bp), the shortest versican promoter luciferase reporter construct which showed hypoxiainduced increases (2.7-fold) after 5 days (Fig 6.15), HMDM were transfected with a deletion series of this 240-bp construct, to analyse which parts are required for high level activity of the versican promoter in hypoxia. As figure 6.16 showed, hypoxic promoter activity markedly dropped in the -56 to +54 construct compared to constructs which contain the region +54+104, indicating that the sequence from +54 to +104 is crucial for high level versican promoter activity in hypoxic HMDM. Analysis of this promoter sequence (Fig 6.17) identified several putative binding sites for

hypoxia inducible transcription factors, the most important one being HIF-1 at +60bp, the master regulator of hypoxia-inducible gene expression (Wenger et al., 2005; Semenza, 2009). The data also showed that promoter activity is significantly reduced in constructs which lack -56 to -26, suggesting that this region also contains regulatory factors which are important for versican promoter activity in hypoxic HMDM (Fig 6.16). Taken together, the data suggested that both the -56 to -26 and +54 to +104 regions are required together in the same construct for high expression from the versican promoter constructs.

Chapter 7: Results

Role of HIF-1 in regulation of versican gene expression in human monocyte-derived-macrophages

7.1 Introduction

A recent study by Asplund et al., (2010) using siRNA in THP-1 cells suggested that the hypoxic inducibility of the versican gene in human monocyte-derived macrophages (HMDM) is controlled by hypoxia inducible factor-1 (HIF-1). HIF-1, a principle mediator of angiogenesis and glycolysis (Semenza, 2003), is the main hypoxia-inducible transcription factor, and regulates numerous genes in response to hypoxic stress (Semenza, 2004). HIF-1 is a heterodimeric molecule which is composed of α and β subunits. HIF-1 β is constitutively expressed and found in the nucleus whereas HIF-1 α is rapidly degraded under normal oxygen levels by the ubiquitin-proteasome pathway which is directed by the tumour suppressor VHL (von Hippel-Lindau protein) (Adams et al., 2009; Semenza, 2009). In the absence of oxygen (i.e. in hypoxia) HIF-1 α protein degradation is reduced in many cell types including macrophages (Talks et al., 2000; Burke et al., 2002). HIF-1 α then accumulates and translocates into the nucleus where it binds to the HIF-1 β subunit and transactivates certain genes via binding to the hypoxia responsive elements (HRE) in their promoters (Semenza, 2009). In addition, accumulating evidence has demonstrated that HIF-1 also responds to non-hypoxic stimuli such as lipopolysaccharide (LPS), cobalt chloride (CoCl₂) and the iron chelator desferrioxamine (DFO) resulting in the transcription of hypoxia-inducible genes under normal oxygen concentrations (Dery et al., 2005; Frede et al., 2006). CoCl₂ and DFO are chemical reagents that are commonly used to induce HIF-1 α protein in normoxia (Guo

et al., 2006). Maxwell et al., (1999) and Yuan et al., (2003) demonstrated that cobalt chloride and DFO are capable of inducing HIF by stabilizing HIF-1 α using mechanisms other than hypoxia.

In this project study, I investigated whether versican promoter reporter constructs containing potential HIF binding sites are induced by HIF-1 α over-expression. Furthermore, the role of HIF-1 in versican mRNA expression was studied in normoxic HMDM by inducing HIF-1 using LPS, CoCl₂ and DFO treatment. In addition as previously shown the role of PI3-kinase in normoxic up-regulation of versican in SMC (Rahmani et al., 2005), here also the possible role of this pathway was investigated in hypoxic up-regulation of versican mRNA in HMDM.

7.2 HIF-1 α over-expression does not affect the expression of versican promoter reporter constructs in normoxic HMDM

Versican promoter analysis to locate the HRE consensus sequences proposed by Wenger et al., (2005), identified one putative HIF binding site in the sense (position: +60, 5'-GG<u>CGTG</u>CG-3') (Fig 6.17) and two in the antisense strand of the promoter (position: -507, 5'-AA<u>CACG</u>CT-3' and -373, 5'-TG<u>CACG</u>GG-3'). To investigate whether these putative HIF binding sites are induced by HIF-1, HIF-1 α was over-expressed in normoxic HMDM which were co-transfected with the 2.5-Kb or 240-bp versican promoter luciferase reporter constructs. In order to transfect PBMC, cells were isolated (See section 3.1.1) and seeded in 4ml complete Iscove's medium at a cell density of 4×10⁶ per well in 6-Well Costar ultra low attachment plates as described previously. After 5 days incubation, 2×10⁶ PBMCs were transfected with 1 µg of DNA of

versican luciferase reporter construct (2.5-Kb or 240-bp), Phosphoglycerate kinase-1 (PGK), empty pGL4.10 [luc2] or pGL3-basic plasmids and co-transfected with 300 ng of HIF-1 α over-expressing plasmid (pCDNA3.1/HIF-1 α ; a kind gift from Professor Chris Pugh, University of Oxford) or the negative control plasmid (pCDNA 3.1 CAT) followed by 3.2 μ l of JetPEI reagent (one well per transfection) according to the manufacturer's protocol (See section 3.5). After 24 hours incubation in normoxia (20.9% O₂), medium was removed and HMDM were washed twice with 1x PBS, lysed, and luciferase activities were quantified. Luciferase values were normalized to the obtained protein concentrations (See sections 3.5.3 and 3.5.4).

As figure 7.1 indicates, HIF-1 α over-expression in normoxic HMDM did not affect the expression of the 2.5-Kb (-2.316Kb/+184bp) versican promoter luciferase construct, which contains potential HIF-1 binding sites in the sense (position: +60) and antisense (position: -507 and -373) strands. In contrast, the control PGK-1 reporter construct, which is known to be regulated by HIF-1 (Carmeliet et al., 1998; Ryan et al., 1998; Ameri et al., 2002), showed significant induction in comparison with the versican reporter construct, demonstrating that the HIF-1 α over-expression plasmid was functional (Fig 7.1). Since the 2.5-Kb versican promoter is cloned into the pGL3-basic reporter vector, pGL3-basic empty plasmid was used as a negative control and as expected it was not induced by HIF-1 α over-expression (Fig 7.1).


Figure 7.1: Effect of over-expression of HIF-1α on the 2.5-Kb (-2.316Kb+184bp) versican promoter construct in normoxic HMDM. HIF-1α was over-expressed in normoxic HMDM which were transfected with the 2.5-Kb versican promoter luciferase reporter construct to investigate whether the construct is responsive to HIF-1. 2×10⁶ PBMCs were transfected with 1 µg of 2.5-Kb versican promoter luciferase pGL3-basic construct, PGK (positive control) or empty pGL3-basic plasmid (negative control). Cells were then co-transfected with 300ng of HIF-1α over-expressing plasmid (pCDNA3.1/HIF-1α) or the negative control plasmid (pCDNA 3.1 CAT) with aid of 3.2 µl of JetPEI reagent according to the manufacturer's protocol and were plated in 2 ml per well at a cell density of 1×10⁶ /ml in 6-Well Nunclon adherent plates. Cells were then incubated under normoxia (20.9% O₂) for 24 hours prior to luciferase assay. Data were normalised to the obtained protein concentrations. Triplicate data from 3 independent experiments using different donors are expressed as means ± SEM. The pGL3-basic value in each experiment was assigned an arbitrary value of 1. Data were further analysed for significant induction using paired two-tailed t-tests. **= p < 0.01

Similarly, as figure 6.17 showed, although the putative HIF-1 binding site within the +54 to +104 region (position: +60) is present in the 240-bp versican promoter construct, induction was not observed in normoxic HMDM co-transfected with the HIF-1 α over-expressing plasmid (Fig 7.2). The pGL4.10 empty plasmid was used as a negative control because the 240-bp construct is cloned into it. In contrast, the positive control PGK-1 reporter construct, which contains HIF-1 binding sites, was significantly induced by HIF-1 α over-expression (Fig 7.2). Overall this data suggest that either these HIF binding sites on the sense and antisense strand of the versican promoter are not functional HIF-1 binding sites, or hypoxic activation of versican promoter is not controlled by HIF-1.



<u>Figure 7.2</u>: Effect of over-expression of HIF-1 α on the 240-bp (-56bp+184bp) versican promoter construct in HMDM. HIF-1 α was over-expressed in normoxic HMDM which were transfected with the 240-bp the versican promoter luciferase reporter construct to investigate whether the construct is responsive to HIF-1. Experimental details as for figure 7.1. Triplicate data from 3 independent experiments using different donors are expressed as means ± SEM. The pGL4.10 value in each experiment was assigned an arbitrary value of 1. Data were further analysed for significant induction using paired two-tailed t-tests. * = p < 0.05

7.3 LPS slightly induce versican mRNA expression in normoxic HMDM

Previous studies have shown that LPS induces HIF-1 α mRNA and protein, as well as expression of a range of HIF-1 inducible genes including vascular endothelial growth factor (VEGF), in macrophages (Blouin et al., 2004; Oh et al., 2008). So to investigate whether LPS might induce versican mRNA, adherence-purified HMDM were treated with LPS in normoxia followed by analysis of versican mRNA levels by Real Time RT-PCR. As previously described, PBMC were cultured at a cell density of 2x10⁶ per well in 6-well adherent plates and incubated for 5 days under normal oxygen condition. After 5 days incubation, adherence-purified HMDM were treated at final concentrations of 100 ng/ml with LPS from either Salmonella abortus equii (SAE) or Salmonella minnesota (MINN), two different gram negative bacteria species which both activate macrophages through TLR4 ligation (Gomes et al., 2010) (See section 1.1.2). Cells were then incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for a further 18 hrs prior to RNA isolation (See section 3.4).

Treatment of HMDM with either LPS from two different bacterial species or hypoxia caused similar induction of the mRNA for the known HIF-1 up-regulated gene, VEGF, which has been shown to be induced by LPS via a HIF-1 dependent mechanism (Ramanathan et al., 2007) (Fig 7.3). However, in the same RNA samples, LPS only slightly induced versican mRNA (MINN: 2.4-fold, SAE: 1.8-fold), in contrast to hypoxia (15.6-fold) (Figure 7.4). This data suggests that hypoxic induction of versican mRNA is likely to be a HIF-1 independent mechanism.

7. <u>Results</u>



Figure 7.3: Effect of LPS on VEGF mRNA expression in comparison with hypoxia. Adherence-purified HMDM were analysed by Real Time RT-PCR to determine the effect of two different preparations of LPS (SAE: Salmonella abortus equii LPS; MINN: Salmonella Minnesota LPS) on VEGF mRNA expression in comparison with hypoxia. 2 x10⁶ cells per well were treated with MINN LPS and SAE LPS at 100 ng/ml and then exposed to normoxia (20.9% O₂) or hypoxia alone (0.2% O₂) for a further 18 hours prior to RNA isolation. Data from 3 independent experiments using different donors were normalised to β-2M mRNA levels and are expressed as means ± SEM. The normoxic value in each experiment was assigned an arbitrary value of 1. Data were further analysed using paired two-tailed t-tests.* = p <0.05 compared to normoxia.



<u>Figure 7.4</u>: Effect of LPS on versican mRNA expression in comparison with hypoxia. Adherence-purified HMDM were analysed by Real Time RT-PCR to determine the effect of two different preparations of LPS (SAE: Salmonella abortus equii LPS; MINN: Salmonella Minnesota LPS) on versican mRNA expression in comparison with hypoxia. Experimental details as explained for figure 7.3. Data from 3 independent experiments and are expressed as means ± SEM. The normoxic value in each experiment was assigned an arbitrary value of 1. Data were further analysed using paired two-tailed t-tests.* = p < 0.05

7.4 Desferrioxamine induces versican mRNA in HMDM but not cobalt chloride

To further investigate whether HIF-1 is involved in the hypoxic induction of versican gene expression, adherence-purified HMDM were treated with the hypoxia mimetic agents DFO and CoCl₂ which stabilise the hypoxia-inducible α subunit of HIF-1, increasing HIF-1 protein levels in normoxia, leading to induce the expression of HIF-1 dependent hypoxia-inducible genes (Wang and Semenza, 1993_b; Minchenko et al., 2002 & 2004; Mojsilovic-Petrovic et al., 2007). As previously described, PBMC were cultured at a cell density of 2x10⁶ per well in 6-well adherent plates and incubated for 5 days under normal oxygen conditions. After 5 days incubation, adherence-purified HMDM were treated with either DFO or CoCl₂ at final concentrations of 200 μ M and 300 μ M respectively and incubated under normoxia (20.9% O₂) or hypoxia (0.2% O₂) for a further 18 hours (See section 3.2) prior to RNA isolation followed by determination of versican, VEGF and GLUT-1 mRNA levels by real time RT-PCR.

Hypoxic inducibility of versican, VEGF and GLUT-1 was first confirmed. As expected, 18 hours hypoxia significantly induced versican (15.5-fold), VEGF (11.7-fold) and GLUT-1 mRNA (15.5-fold) (Fig 7.5). As figure 7.6 indicates, treatment of normoxic HMDM with DFO also significantly induced versican mRNA expression in normoxia (10-fold), as well as inducing the two known HIF-1 inducible control mRNAs, coding for VEGF (11-fold) and GLUT-1 (29.5-fold), in the same RNA samples. In contrast, as figure 7.7 shows, CoCl₂ markedly induced VEGF (10.6-fold) and GLUT-1 mRNAs (6.3-fold) but not versican mRNA expression, in the same RNA samples. DFO and CoCl₂ have been previously shown to induce VEGF and GLUT-1 via the induction of HIF-1 protein

(Hayashi et al., 2004; Mace et al., 2007; Lin et al., 2008). Therefore these data suggest that versican is regulated differently to VEGF and GLUT-1, via mechanisms which can be activated by hypoxia and DFO but not by CoCl₂. Since cobalt chloride is known to induce HIF-1 protein (Maxwell et al., 1999), HIF-1 is clearly not sufficient in itself to induce the versican promoter, suggesting that a HIF-1 independent mechanism up-regulates versican hypoxic induction in HMDM.



Figure 7.5: Effect of 18 hours hypoxia on versican, VEGF and GLUT-1 mRNA expression in adherence-purified HMDM. HMDM were analysed by Real Time RT-PCR to determine the effect of 18 hours 0.2% O₂ on versican mRNA expression in comparison with VEGF and GLUT-1. 2 x 10⁶ cells per well were exposed to normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 18 hours prior to RNA isolation. Data from minimum 4 independent experiments using different donors were normalised to β-2M mRNA levels and are expressed as means ± SEM. The normoxic value in each experiment was assigned an arbitrary value of 1. Data were further analysed using paired two-tailed t-tests. ** p < 0.01,* = p <0.05 compared to the normoxia samples.



Figure 7.6: Effect of desferrioxamine on versican, VEGF and GLUT-1 mRNA expression in adherence-purified HMDM. HMDM were analysed by Real Time RT-PCR to determine the effect of DFO on versican mRNA expression in comparison with VEGF and GLUT-1. Experimental details as explained for figure 7.5. Data were further analysed using paired two-tailed t-tests. ** p < 0.01,* = p <0.05 compared to the normoxia samples.



Figure 7.7: Effect of cobalt chloride on versican, VEGF and GLUT-1 mRNA expression in adherence-purified HMDM. HMDM were analysed by Real Time RT-PCR to determine the effect CoCl₂ on versican mRNA expression in comparison with VEGF and GLUT-1. Experimental details as explained for figure 7.5. Data were further analysed using paired two-tailed t-tests.* = p <0.05 compared to the normoxia samples.

7. <u>Results</u>

7.5 Desferrioxamine induces versican mRNA due to iron chelating

It is well known that desferrioxamine (DFO) is an iron chelating agent which binds and removes free iron from the bloodstream (Piga et al., 2003). In addition, studies have suggested that DFO stabilises HIF- α in normoxia by acting as an iron chelator that depletes the iron required for HIF- α hydroxylation (Maxwell et al., 1999; Yuan et al., 2003). Therefore following the finding of induction of versican mRNA by DFO in normoxic adherence-purified HMDM, further investigation was carried out to confirm that induction could be blocked by addition of iron. As previously described, PBMC were cultured at a cell density of 2x10⁶ per well in 6-well adherent plates and incubated for 5 days under normal oxygen condition. After 5 days incubation, adherence-purified HMDM were treated simultaneously with DFO at a final concentration of 200 μ M and ferric citrate at final concentrations of 100 μ M, 200 μ M and 400 µM (See section 3.2.5) to saturate DFO and nullify the effects of its ironchelating activity. Cells were then incubated under normoxia (20.9% O_2) for a further 18 hours prior to RNA isolation followed by determination of versican mRNA levels by Real Time RT-PCR.

As expected, DFO treatment alone significantly induced versican mRNA (Fig 7.8) whereas versican gene expression was gradually and significantly reduced as the concentration of the iron was increased and finally reached a similar level of versican mRNA expression in untreated normoxic cells. This data shows that the induction of the versican gene by DFO was completely blocked by the addition of iron, confirming that the iron-chelating activity of DFO, rather than any other activity or contaminant, was responsible for versican induction.

7. <u>Results</u>





Adherence-purified HMDM were analysed by Real Time RT-PCR to determine the effect of added iron on versican mRNA induction by desferrioxamine (DFO). 2 x 10^6 cells per well were treated simultaneously with DFO at final concentration of 200 μ M and ferric citrate at final concentration of 100 μ M, 200 μ M and 400 μ M and then were incubated in normoxia (20.9% O₂) for a further 18 hours prior to RNA isolation. Data from 3 independent experiments using different donors were normalised to β -2M mRNA levels and are expressed as means ± SEM. The normoxic value in each experiment was assigned an arbitrary value of 1. Data were further analysed using paired two-tailed t-tests. *** = p <0.001 ** p < 0.01,* = p <0.05

7.6 PI3-kinase inhibitors reduce hypoxic induction of versican mRNA in adherence-purified HMDM

It has been previously reported that versican can be up-regulated in normoxic smooth muscle cells (SMC) by the PI3-kinase pathway, via accumulation of β -catenin in the nucleus and binding of a β -catenin / TCF complex to the versican promoter (Rahmani et al., 2005). In addition, several studies have shown that hypoxia activates the PI3-Kinase signaling pathway leads to induce increases of β-catenin mRNA and protein (Alvarez-Tejado et al., 2001; Liu et al., 2010) which enhances expression of various genes such as MMP7 in human primary macrophages (Deguchi et al., 2009) and IL-8 (Xu et al., 2004). To investigate the possible role of this pathway in hypoxic upregulation of versican, HMDM were treated with two distinct specific inhibitors of PI3kinase, LY290042 and wortmannin, to analyse their effects on hypoxia-induced increases in versican mRNA in HMDM. As previously described, PBMC were cultured at a cell density of 2x10⁶ per well in 6-well adherent plates and incubated for 5 days under normal oxygen conditions. Five day matured adherence-purified HMDM were treated with LY290042 (2 μ M and 5 μ M) or wortmannin (200 nM) and then incubated under either normoxia (20.9% O₂) or hypoxia (0.2% O₂) for a further 18 hours (See section 3.2) prior to RNA isolation followed by determination of versican and GLUT-1 mRNA levels by Real Time RT-PCR.

As figure 7.9 indicates, reduced induction of versican mRNA by hypoxia was markedly observed in HMDM treated with two specific inhibitors of PI3-Kinase, LY294002 (2 μ M: 3-fold and 5 μ M: 4.5-fold reduction) and wortmannin (2.6-fold reduction) in comparison with DMSO treated cells although this data was not statistically significant

due to high variability in hypoxic fold induction among these donors as it was earlier shown in this project (Fig 4.5). In addition, since previous studies showed that PI3-Kinase activation by hypoxia increased HIF-1α synthesis in different cell lines (Mazure et al., 1997; Chen et al., 2001; Xu et al., 2004; Lee et al., 2006), in this project, HMDM treated with PI3-Kinase inhibitors were investigated for a known HIF-1 inducible mRNA, coding for GLUT-1. As figure 7.9 shows, no reduction of hypoxic induction was observed in GLUT-1 mRNA following treatment with PI3-Kinase inhibitors, suggesting that the previous findings, showing the possible activation of PI3-K/HIF-1 pathway by hypoxia, could be confined to specific cell types. Overall this data suggests a possible role for the PI3-Kinase pathway in hypoxic up-regulation of versican mRNA in HMDM and shows that versican is affected differently by PI3-Kinase inhibitors compared to a classical HIF-1 regulated gene, GLUT-1.





As previously discussed, in the absence of PI3-Kinase activity, β -Catenin, a cytoplasmic phosphoprotein, is phosphorylated by GSK-3, which is targeted for degradation by the ubiquitin-proteasome system (Polakis, 2000). Several studies suggested a correlation between β-catenin accumulation in the nucleus and hypoxic activation of PI3-kinase through GSK-3β degradation (Demir et al., 2009; Du et al., 2010; Mazumdar et al., 2010). More in-depth studies showed β -Catenin can interact with transcription factors such as TCF/LEF (Clevers and van de Wetering, 1997; Cui et al., 2010) and control the transcriptional activation of various genes such as versican in smooth muscle cells (Rahmani et al., 2005). In the present study, in a preliminary investigation, I investigated the possible hypoxic up-regulation of nuclear β -catenin protein level in U937 cells, a human monocytic cell line since they are able to grow rapidly and reach a high level of density in comparison with HMDM which take 5 days to be generated from PBMC. For this experiment, U937 cells were cultured at a cell density of 5x10⁶ per flask and incubated under either normoxia (20.9% O2) or hypoxia (0.2% O2) for 24 hours prior to nuclear protein extraction followed by immunoblotting for β -catenin and actin (See sections 3.6 and 3.7).

As figure 7.10 shows, hypoxia markedly up-regulated the nuclear β -catenin protein in U937 cells in comparison with normoxia, suggesting the possible involvement of this signaling pathway in hypoxic regulation of nuclear β -catenin in U937 cells. This data is in accordance with previous studies which showed hypoxia-induced increases in nuclear β -catenin in several cell lines and indicated a correlation between β -catenin accumulation and activation of PI3-kinase by hypoxia (Deguchi et al., 2009; Liu et al., 2010; Mazumdar et al., 2010).

7. <u>Results</u>



Figure 7.10: Effect of hypoxia on β-catenin protein expression in U937 cell nuclear extracts. Nuclear extracts were immunoblotted to assess the presence of β-catenin in normoxia and hypoxia. 5×10^5 cells per ml in 10 ml complete RPMI1640 were exposed to normoxia (20.9% O₂) or hypoxia (0.2% O₂) for 24 hours prior to nuclear protein extraction. 100 µg of nuclear protein was then immunoblotted for β-catenin and actin using 1:2000 and 1:4000 dilutions of primary antibody, respectively, followed by a 1:10000 dilution of secondary antibody. A blot representative of 2 independent experiments is shown.

7.7 Summary

In this chapter, study was first investigated whether the versican promoter was induced by HIF-1 α over-expression as a previous study suggested that the hypoxiainduced increases in versican mRNA are controlled by HIF-1 (Asplund et al., 2010). As figures 7.1 and 7.2 showed, HIF-1 α over-expression did not affect the expression of the 2.5-Kb or 240-bp versican promoter constructs, which both contain HIF binding sites. However, the positive control PGK-1 reporter construct, which contains HIF-1 binding sites, was strongly induced by HIF-1 α over-expression in all experiments, proving that the HIF-1 α over-expression construct was functional and effective at the ratios of reporter to over-expression plasmid DNA. In addition, as figure 7.4 showed, LPS treatment of HMDM slightly induced versican mRNA in contrast with the mRNA of the HIF-1 regulated gene VEGF which has been shown to be up-regulated by LPS via HIF-1 (Ramanathan et al., 2007). To further investigate the possibility of up-regulation of versican mRNA expression by HIF-1, HMDM were treated with DFO and CoCl₂, two hypoxia mimetic agents. As figures 7.6 and 7.7 showed, only DFO, and not CoCl₂, significantly induced versican mRNA, whereas the HIF-1 regulated genes VEGF and GLUT-1 were up-regulated by both DFO and CoCl₂, in the same RNA samples. These findings suggested that hypoxic induction of versican mRNA is regulated differently to VEGF and GLUT-1, via mechanisms which can be activated by hypoxia and DFO but not by CoCl₂. In addition, as figure 7.8 indicated, the level of versican mRNA induction was significantly dropped when the iron concentration was increased, strongly suggesting that the iron chelating activity of DFO induced versican mRNA in HMDM, rather than any other activity, or chemical or microbial contaminant. Further investigation, using two inhibitors of the PI3-kinase pathway, LY294002 and wortmannin, markedly reduced induction of versican mRNA by hypoxia (Fig 7.9), suggesting a role for PI3kinase in hypoxic up-regulation of versican mRNA in adherence-purified HMDM. Finally, accumulation of β -catenin protein was shown in the nucleus of hypoxic U937 cells (Fig 7.10) that it is in line with previous studies which showed hypoxia induced nucleus accumulation of β -catenin in different cell lines (Liu et al., 2010; Mazumdar et al., 2010).

Chapter 8: Discussion

Versican is a large extracellular matrix proteoglycan which is expressed from the early stages of development (Westergren-Thorsson et al., 1998) through to adult tissues (Perides et al., 1992; Zimmermann et al., 1994). Versican expression is also elevated in pathological sites, including in healing wounds (Landolt et al., 1995; Theocharis et al., 2000) and in a variety of tumours such as breast (Kischel et al., 2010) brain (Schwartz and Domowicz, 2004) and melanoma (Domenzain et al., 2003). On the other hand, macrophages accumulate in poorly vascularised and hypoxic sites including solid tumours, wounds, and sites of infection and inflammation where they can be exposed to low levels of oxygen for long periods (Lewis et al., 1999; Vaupel et al., 2001). In addition, several studies have demonstrated that versican expression is related to tumour progression in many malignant tumours (Ricciardelli et al., 1998; Touab et al., 2002; Voutilainen et al., 2003) where hypoxia is present. Furthermore macrophages also accumulate in atherosclerotic plaques in which both hypoxia and high versican expression have been reported (Husain et al., 1999; Talusan et al., 2005; Kenagy et al., 2006). Moreover unpublished data by Staples and Burke et al., (B. Burke, personal communication) showed that hypoxia produced very high versican mRNA fold inductions in macrophages (data not shown). Taking into account the above findings linking disease outcomes with macrophages, hypoxia, and versican expression, in this study I set out to establish how versican expression is regulated by hypoxia in macrophages, principally by a thorough deletion analysis of the versican promoter, which has not previously been carried out.

A recent study by Asplund et al., (2010) demonstrated that versican mRNA and protein are induced by exposure to hypoxia (0.5% O₂) for 24 hours in primary HMDM. The previously mentioned unpublished data by Staples and Burke et al., (B. Burke, personal communication) showed that extended periods of severe hypoxia (0.2% O₂, 5 days) in HMDM (data not shown) and PBMC (Fig 1.8) produce far higher versican mRNA fold inductions (140 and 700-fold respectively) than those which were obtained with overnight hypoxia (37 and 20-fold respectively) or which Asplund et al., (2010) obtained in HMDM with 24 hours hypoxia (40-fold). This finding is similar to those in our recent publication which showed that 5 days of hypoxia induced much higher mRNA levels of the pro-angiogenic cytokine VEGF and the glucose transporter GLUT-1 in PBMC compared to 24 hours hypoxia (Staples et al., 2010). No published work to date has investigated the effect of long term hypoxia on versican gene expression in primary HMDM, and compared it with short-term hypoxia. This was one of the key aims of my studies.

Initially, I investigated the effect of 18 hours hypoxia (0.2% O_2) on versican mRNA expression in HMDM from 15 different healthy donors. Substantial hypoxic induction of versican mRNA was observed in all 15 different donors after 18 hours exposure to 0.2% O_2 (Figure 4.5). Despite high variability in hypoxic fold induction among these donors (average 34-fold, range 11-125-fold), these data clearly confirmed the hypoxic up-regulation of versican mRNA in adherence-purified HMDM. In addition, since similar hypoxic inductions were observed when some of the same donors were used in further experiments, the high degree of variability in hypoxic fold induction of versican mRNA could possibly be due to single nucleotide polymorphisms (SNP) in the versican

promoter, although no studies to date have reported these. Overall this finding is similar to Asplund's observation which showed an average of 40-fold hypoxic induction in HMDM from 8 different donors after 24 hours exposure to 0.5% O₂, although the method of blood preparation and generation of adherence purified HMDM were slightly different to this project (Asplund et al., 2010). Furthermore my experiments on versican protein expression, using intracellular flow cytometry analysis, did not show hypoxia-induced increases of versican protein after 24 hours (Fig 5.3) whereas significant induction of versican protein (3-fold) was observed after exposure to hypoxia for 5 days in PBMC-derived macrophages (Fig 5.2). These findings therefore are in line with the versican mRNA findings which suggest far higher production of versican after long term hypoxia (Fig 1.8).

Versican has been shown to form a complex with fibronectin and VEGF leading to enhanced endothelial cell adhesion, proliferation, and migration, and thus enhanced angiogenesis (Zhang et al., 1998; Zheng et al., 2004). In addition, another study has reported that versican activates macrophages through TLR2 engagement leading to the release of pro-inflammatory cytokines which promote metastatic growth (Kim et al., 2009). Versican has been shown by other studies to play important roles in cancer invasion and metastasis (Zheng et al., 2004; Ricciardelli et al., 2009). Therefore this finding of unexpectedly high up-regulation of versican in primary human macrophages exposed to long term hypoxia is biologically relevant, because macrophages entering hypoxic tissues remain there for extended periods (Lewis and Murdoch, 2005). Taken together, these findings suggest that macrophages may be a more important source of versican in hypoxic sites than previously realised.

In this project, I carried out experiments on PBMC (a mixed population of monocytes and lymphocytes) to test my hypothesis that the hypoxic up-regulation of versican mRNA is limited to macrophages. In order to compare the level of hypoxic induction of versican mRNA in macrophages and lymphocytes, monocytes (CD14⁺) were separated from lymphocytes (CD14⁻ cells) using MACS magnetic beads linked to an antibody raised to the surface antigen CD14. As figure 4.6 indicated, versican mRNA levels in CD14⁺ purified monocyte-derived macrophages were compared to adherence-purified HMDM and to lymphocytes. These data demonstrated that 18 hours of hypoxia significantly induced versican mRNA expression in adherence-purified HMDM (34-fold) and CD14⁺ monocyte-derived macrophages (25-fold) but not in lymphocytes (CD14⁻), indicating that macrophages are the principle mononuclear cells showing hypoxic upregulation of versican mRNA.

This finding was also confirmed when I analysed versican protein expression in PBMC after both 24 hours and 5 days incubation in normoxia or hypoxia using intracellular staining followed by flow cytometry. As figures 5.4, graph C and F showed, very low versican mean fluorescent intensities were observed in lymphocytes (in the R4 region) in both normoxia and hypoxia. These findings suggest that among PBMC, versican protein expression is confined to monocyte/macrophage lineage cells and is not detectable in lymphocytes, which matches well with the low versican mRNA level, and lack of hypoxic induction, which I found in CD14⁻ cells (lymphocytes) (Fig 4.6).

It was also of interest to examine whether macrophage maturation affected versican protein expression. As figure 5.4 demonstrated, increasing monocyte/macrophage cell

size (forward scatter) on the dot-scatter plot was associated with increased versican mean fluorescent intensity in both normoxia and hypoxia; the versican mean fluorescent intensity was higher in the R1 region (large mature macrophages), in both normoxic and hypoxic macrophages, in comparison with R2 and R3 (smaller, less granular, less mature monocytes/macrophages). Since increased side and forward scatter correlate with macrophage maturation (Alexis et al., 2006; Lay et al., 2009), these novel data indicate that versican protein expression increases with monocyte to macrophage differentiation / maturation.

The recent study by Asplund et al., (2010) using immunoblotting, failed to observe induction of versican protein in HMDM after 24 hours hypoxia ($0.5\% O_2$) in comparison with normoxia. However, Asplund et al., reported 3 to 7-fold hypoxia increases in versican bound GAG chains which were detected by [³⁵S]-sulphate and D-[6-³H]glucosamine hydrochloride using indirect radioactive evaluation of the levels of versican in normoxic and hypoxic HMDM. The reasons for the conflicting data from Asplund are not clear, but one possibility is a simple failure of their immunoblotting procedure, as versican is known to be quite difficult to detect using this technique because its large, variable size and variable post-translational modifications. A second possibility is donor-to-donor variability; as can be seen from figure 4.5, hypoxic induction, at least at the mRNA level, varies markedly.

With these conflicting data from Asplund et al., in mind, a key aim of my work was to clarify the facts regarding hypoxic up-regulation of versican protein in macrophages. When I analysed my flow cytometry data, the monocyte/macrophage population,

after 24 hours in either normoxia or hypoxia, showed constitutive versican protein expression, but no hypoxia-induced increases (Fig 5.3, plots B and D). However, analysis of versican protein expression in PBMC-derived macrophages after 5 days of hypoxia indicated significant hypoxic induction of versican protein (3-fold) (Fig 5.2, plot B, D and E).

My data allow an interesting observation which was not mentioned by Asplund et al., (2010): despite the high level of versican mRNA induction by hypoxia (average 140fold) in HMDM, the level of hypoxia-induced increases of versican protein remained very low (average 3-fold in my data, 3-7 fold in Asplund et al.,). This may be explained by previous findings showing that protein translation is reduced under hypoxia (Heerlein et al., 2005; Liu et al., 2006). Indeed, several studies have reported that hypoxia activates 4E-binding protein-1 (4E-BP1), a repressor of mRNA translation following hypoxic inactivation of eukaryotic elongation factor-2 (eEF2) which is an essential factor for protein synthesis (Tinton and Buc-Calderon, 1999; Arsham et al., 2003; Connolly et al., 2006). On the other hand, several other studies have investigated the correlation between the mRNA expression levels and protein abundance in many cell types suggesting significant correlations in mRNA changes and protein expression (Chen et al., 2002; Guo et al., 2008; Gry et al., 2009; Maier et al., 2009). In addition, a ground-breaking study by Pascal et al., (2008) using gene arrays and immunohistochemistry on different cell lines, demonstrated a poor relationship between mRNA and protein level for many genes. Therefore I conclude it is possible that versican could be one such gene, for which either the mRNA level does not

correspond closely with the protein expression level, or translation of versican is affected by hypoxia.

To summarise my versican mRNA and protein data, which I have discussed above, they provide good evidence in support of previous data on the biological role of versican, which suggested versican accumulation in atherosclerotic plaques may be mediated by macrophages (Seidelmann et al., 2008) and demonstrated co-localization of versican in poorly vascularised and hypoxic macrophage-rich areas such as carotid lesions (Asplund et al., 2010). As previously described, accumulation of macrophages has been reported in hypoxic and necrotic sites in breast cancers (Leek et al., 1999) and ovarian carcinomas (Negus et al., 1997), wounds (Crowther et al., 2001), atherosclerotic plaques (Hansson et al., 1988; Takahashi et al., 2002; Collot-Teixeira et al., 2007) and arthritic joints (Hansch et al. 1996; Hollander et al., 2001). Several reports have shown that versican is highly expressed in the above pathologies and highlighted the role of versican in wound healing (Theocharis et al., 2000 & 2002) and in vascular disease, especially atherosclerosis which is thought in part to be mediated by binding of versican to low density lipoproteins via its large number of GAG chains (Camejo et al., 2002; Khalil et al., 2004; Seidelmann et al., 2008). My results showing up-regulation of versican mRNA and protein by macrophages exposed to prolonged hypoxia provide further evidence for a likely important role for macrophages in versican accumulation in the blood vessel walls, which is believed to promote extracellular lipoprotein retention and uptake leading to foam cell formation, the key step in the atherosclerotic process (Wight and Merrilees, 2004).

Regarding the mechanism of versican mRNA induction by hypoxia, in this project I also investigated the decay of versican mRNA in normoxic and hypoxic adherence-purified HMDM to determine whether the observed up-regulation of versican mRNA by hypoxia is due to increased transcription or increased mRNA stability. Interesting studies by Chen and Shyu, (1995) and subsequently several other groups have identified specific RNA-binding proteins which can bind to adenylate-uridylate rich elements (AREs) such as AUUUA sequences, present in 3' untranslated regions (UTR) of many mRNAs, including growth factors and cytokines, and increase stability of the mRNA by protecting them from degradation (Chen and Shyu, 1995; Bakheet et al., 2001; Barreau et al., 2005). HuR is one such RNA-binding protein, which in addition has been found to be induced by hypoxia, and binds to the AU-rich elements of the VEGF 3' UTR, playing a critical role in the hypoxic stabilization of VEGF mRNA (Levy et al., 1998; Goldberg-Cohen et al., 2002; Leandersson et al., 2006).

My analysis of the versican 3'UTR identified several putative AU-rich elements containing the sequence AUUUA (Fig 8.1) which are predicted by computer analysis to be potentially bound by the HuR protein, which could lead to mRNA stabilization. Therefore I decided to investigate whether the observed up-regulation of versican mRNA by hypoxia is due to increased mRNA stability. Analysis of versican mRNA decay indicated very similar rates of versican mRNA decay in normoxic and hypoxic adherence-purified HMDM at different time points up to 36 hours (Fig 4.7). This finding shows that hypoxic induction of versican mRNA does not occur via changes in mRNA stability in hypoxia. This finding provided the justification for further analysis of

the versican promoter as an essential step on the way to elucidate the mechanism of hypoxic up-regulation of versican.

2188 CAUUUUGGGUUGCCGUGCUCCCAAAACAUUUUAAAUGAAAGUAUUGGCAUUCAAAAAGAC
2248 AGCAGACAAAUGAAAGAAAUGAGAGCAGAAAGUAAGCAUUUCCAGCCUAUUUAAUUUC
2308 UUUAGUUUUCUAUUUGCCUCCAGUGCAGUCCAUUUCCUAAUGUAUACCAGCCUACUGUAC
2368 UAUUUAAAAUGCUCAAUUUCAGCACCGAUGGCCAUGUAAAUAAGAUGAUUUAAUGUUGAU
2428 UUUAAUCCUGUAUAUAAAAUAAAAGUCACAAUGAGUUUGGGCAUAUUUAAUGAUGAUUA

Figure 8.1: The versican 3'-untranslated region (UTR) from base pairs 2188-2428 is shown. The underlined letters indicate putative HuR binding sites.

In this project, in order to define the DNA sequences of the versican promoter and possible regulatory elements which are responsible for the hypoxic regulation of versican, a 2.5-Kb (-2.316Kb/+184bp) versican promoter pGL3-basic construct and a series of versican promoter pGL4.10 constructs including 4.140-Kb (-3.956Kb/+184bp), 830-bp (-646bp/+184bp), 480-bp (-296bp/+184bp) and 240-bp (-56bp/+184bp) were made and transfected into HMDM. These sequences were chosen for analysis because it has previously been shown that an 876-bp (-632/+240) piece of the versican DNA sequence constitutes an active promoter. This region contains a typical TATA box and has putative binding sites for a number of transcription factors including CREB, AP-1, AP-2, SP1 and TCF/LEF (Naso et al., 1994; Rahmani et al. 2005; Domenzain et al., 2009) some of which are known to be hypoxia inducible (Xu et al., 2000; Uenoyama et al., 2006; Leonard et al., 2008). Since previous preliminary investigations showed that versican mRNA is up-regulated much more highly by long term hypoxia (5 days) than

by short term (24 hours) in PBMC and HMDM (Staples and Burke et al., unpublished), in addition to my finding which demonstrated hypoxia-induced increases of versican protein after 5 days, I decided to investigate the effect of 24 hours as well as 5 days hypoxia on HMDM transfected with the versican promoter reporter constructs.

My initial experiments to optimise the transfection in HMDM which were cotransfected with the pRL-TK Renilla luciferase plasmid to serve as an internal control for normalization showed hypoxia-induced increases of Renilla luciferase expression which was very problematic. In accordance with my findings, a very recent study by Doran et al., (2011) examined the effect of hypoxia on luciferase reporter constructs and showed that hypoxia enhanced Renilla luciferase expressed by the pRL-TK plasmid, which demonstrates that caution must be used when interpreting transfection data obtained under hypoxic conditions. Taking this problem into account, I initially had to use the "raw" (non-normalised) firefly luciferase values for analysis of transfection experiments.

In further experiments I normalised the firefly luciferase values from the versican promoter reporter constructs to the obtained protein concentrations. My preliminary investigation analysing the non-normalised luciferase expression by the 2.5-Kb versican pGL3-basic construct, which was made by Dr Burke before I started this project, showed hypoxia-induced increases (2-fold) and confirmed the promoter activity of this construct in HMDM after 24 hours hypoxia (Table 6.1). Because of this relatively low fold induction, compared to the fold inductions of the endogenous versican mRNA obtained in RT-PCR experiments (Fig 4.5), I made a longer reporter

construct, since I hypothesized that increasing the length of 2.5-Kb versican promoter from the 5' end may include transcription factor binding sites which are required for hypoxic up-regulation. As figure 6.14 showed, however this 4.140-Kb versican pGL4.10 construct was expressed in HMDM but no hypoxic induction was observed. This suggested that no extra hypoxic regulatory elements had been included in this longer construct in comparison with 2.5-Kb (Fig 6.14). Given the complete lack of induction, it is even possible that negative regulatory elements had been included in this longer construct. I therefore decided to make a series of constructs shorter than 2.5-Kb to examine the possibility that negative regulatory elements may be present in the promoter. Shorter constructs also have the advantage of containing fewer potential transcription factor binding sites to analyse. My further experiments using 830-bp, 480-bp and 240-bp pGL4.10 constructs as well as the 2.5-Kb pGL3-basic construct indicated that all the versican promoter constructs were functional under normoxia and hypoxia and some showed inducibility after short term hypoxia (Fig 6.14). I then studied the effect of longer hypoxia on HMDM transfected with 4.140-Kb pGL4.10, 2.5-Kb pGL3-basic, 830-bp pGL4.10 and 480-bp pGL4.10 constructs, which showed hypoxic induction of non-normalised luciferase activity in all the versican promoter reporter constructs after 5 days (Fig 6.15).

My data demonstrated that 5 days hypoxia markedly induced the shortest promoter reporter construct (240-bp pGL4.10; 2.7-fold), suggesting the possible existence of hypoxia regulatory elements in the 240-bp promoter fragment (Fig 6.15). Therefore I made a deletion series of the 240-bp (-56bp/+184bp) versican proximal promoter

luciferase reporter construct aiming to elucidate the DNA sequences responsible for hypoxic activity of the versican promoter in HMDM.

Another problem during the transfection experiments was that the level of luciferase expressed by the negative control reporter plasmids, pGL3-basic and pGl4.10, was very low and often undetectable. To solve this problem, and to act as a more comparable control for the very short versican promoter deletion constructs I made, a 29-bp random nucleotide sequence was cloned into pGL4.10 and used as a negative control. Unexpectedly, I observed hypoxia-induced changes in luciferase expression from this random construct, although the actual luciferase expression values were significantly lower than luciferase expression from the versican promoter constructs (Fig 6.16). However, these data indicated that great caution must be used when interpreting transfection data from short promoter reporter constructs obtained under hypoxic conditions.

As figure 6.16 showed, high levels of luciferase expression were observed in normoxic and hypoxic HMDM transfected with all versican promoter reporter constructs which contain the region from +54 to +104 whereas luciferase activity was markedly lower in constructs which do not contain this region, suggesting that +54 to +104 is important for high level expression of versican promoter constructs in normoxia and hypoxia. Further experiments on shorter versican promoter constructs showed that luciferase activity is noticeably lower in constructs beginning at -26 in comparison with -56, suggested that the region from -56 to -26 is also required for high level activity of versican promoter reporter constructs in normoxia (Fig 6.16).

Further analysis of the 240-bp (-56bp/+184bp) promoter construct, using Genomatix MatInspector and TESS software, revealed several putative binding sites for hypoxia inducible transcription factors, including CREB (Leonard et al., 2008), AP-1 (Uenoyama et al. 2006), SP-1(Xu et al., 2000; Szalad et al., 2009), E2F (O'Connor and Lu, 2000) and most importantly one site for HIF-1, the "master regulator" of hypoxic gene expression, at position +60 (Wenger et al., 2005) (Fig 6.17). This putative HIF-1 site was not identified in the recent initial report of versican hypoxia responsiveness (Asplund et al., 2010). As mentioned above, my data demonstrated reduced hypoxic activity in the -56 to +54 construct compared to constructs which contain the region +54 to +104, which includes the putative HIF binding site at +60bp (Fig 6.17). Furthermore, my data showed that promoter activity is markedly reduced in constructs which lack -56 to -26, regardless of whether they contain the putative +60 HIF site or not, suggesting that regions which do not contain a HIF-1 binding site are still important for versican promoter activity. Overall these data showed that both the -56 to -26 and +54 to +104 regions are required together in the same construct for high expression from 240-bp versican promoter (Fig 6.16).

Taking the above data into account, I propose two possible alternative mechanisms which could mediate high level transcription of the versican promoter in hypoxia. Firstly, I speculate that two or more independent hypoxia inducible transcription factors such as CREB, AP-1, E2F or HIF-1, could bind in these two regions and induce increases in versican promoter activity. Several studies have confirmed the hypoxia inducibility of CREB (Beitner-Johnson and Millhorn, 1998; Leonard et al., 2008), AP-1 (Uenoyama et al., 2006; Sermeus et al., 2008) and E2F (O'Connor and Lu, 2000) as well

as HIF-1, the main hypoxia-inducible transcription factor in many different cell lines (Semenza et al., 1994; Wenger et al., 2005). My second hypothesis proposes that transcription factors binding in these two regions may interact to facilitate the transcription of the versican promoter in hypoxia. Such regulatory elements include AP-1 (position: -34 and -4) which can interact with NF-1 (position: +94) (Shi et al., 1999), CREB (position: -34) with HIF-1 (position: +60) (Kvietikova et al., 1995) or AP-1 with HIF-1 (Yamashita et al., 2001; Salnikow et al., 2002). Further studies are required to investigate these alternative hypotheses.

In further experiments I decided to test whether the putative HIF binding site at position +60, within the +54 to +104 region of the versican promoter, is induced by HIF-1. As Figure 7.2 showed, constitutive HIF-1 α over-expression did not affect the expression of the -56bp/+184bp (240-bp) versican promoter luciferase reporter construct in normoxic HMDM, in contrast to the control PGK-1 reporter construct which is known to be regulated by HIF-1 (Carmeliet et al., 1998; Ryan et al. 1998, Ameri et al. 2002). In addition, my further analysis of the versican promoter sequence identified two other candidate HIF binding sites at positions -507 and -373 which were identical to the consensus HRE sequences proposed by Wenger et al., (2005). As figure 7.1 showed, HIF-1 α over-expression in normoxic HMDM did not affect the expression of the 2.5-Kb versican promoter luciferase reporter construct, which contains all three potential HF-1 binding sites (position -507, -373 and +60). Therefore, taking these data into account, I hypothesise that:

(I) These three potential hypoxia responsive elements which have not previously been identified, are non-functional HIF-1 binding sites; several studies have reported the existence of apparently non-functional HIF-1 binding sites in the proximal promoter of different genes which are known to be induced by hypoxia. For example, a study by Kietzmann et al., (1999) demonstrated that the promoter of the hypoxia-inducible plasminogen activator inhibitor-1 (PAI-1) gene contains two potential HIF binding sites, but further investigation using HIF-1 over expression and gel shift assays confirmed the role of only one of these potential HIF binding sites in the hypoxic induction of PAI-1 and HIF-1 binding (Kietzmann et al., 1999). In addition, a study by Garayoa et al., (2000) showed hypoxic induction of adrenomedullin (AM) via a HIF-1 dependent pathway. This study also identified several HIF binding sites on the sense and antisense strands of the AM promoter and showed that luciferase promoter constructs containing HIF binding sites close to the TATA box are not inducible in response to DFO treatment, which is known to induce HIF-1 protein in normoxia, suggesting that these potential sites are non-functional (Garayoa et al., 2000).

(II) These three HIF binding sites are not sufficient for efficient induction of the versican promoter; a study by Yamashita et al., (2001) despite showing the critical role of HIF-1 in hypoxic induction of endothelin-1 (ET1), suggested that the ET-1 HIF-1 binding site alone on the promoter sequences is not sufficient to mediate transcriptional responses to hypoxia and requires the additional DNA-binding elements such as AP-1 and GATA-2 (GATA-binding transcription factor-2) in the vicinity of the HIF-1 binding site (Yamashita et al., 2001). Therefore further investigation is required to determine whether the three potential HIF-1 binding sites in the versican

promoter are functional binding sites and whether they need to interact with other factors to direct hypoxia inducible transcription of the promoter.

In addition, to investigate the possible role of HIF-1 in inducing versican promoter reporter constructs, I also studied the role of HIF-1 in the hypoxic up-regulation of endogenous versican mRNA. Previous studies have shown the up-regulation of HIF-1 α mRNA and protein by LPS in monocyte/macrophage lineage cells and suggested the activation of hypoxia-regulated genes by LPS through a HIF-1 dependent pathway in normoxic cells (Blouin et al., 2004; Oh et al., 2008). Therefore I decided to investigate whether this mechanism also is involved in the up-regulation of versican mRNA. I treated normoxic adherence-purified HMDM with LPS, an inducer of HIF-1 (Blouin et al., 2004), and as figure 7.4 showed, LPS from two different bacterial species slightly induced versican mRNA, in contrast with significant induction by hypoxia, whereas adherence-purified HMDM treated with LPS or hypoxia indicated significant induction of the mRNA level of the known HIF-1 regulated gene VEGF (Fig 7.3). VEGF has been shown to be up-regulated by LPS via a HIF-1-dependent mechanism (Ramanathan et al., 2007). Therefore I conclude that hypoxic induction of versican mRNA is regulated differently to VEGF and that hypoxic induction of versican mRNA is regulated through a mechanism which is not activated by LPS stimulation under normoxia. Thus these data cast doubt on the role of HIF-1 in hypoxic up-regulation of versican in human primary macrophages, as proposed by Asplund et al., 2010. Therefore to assess this, I used DFO and CoCl₂ in further experiments to investigate the possibility of upregulation of versican mRNA expression by HIF-1 in HMDM.

Several studies have reported that DFO and CoCl₂, two hypoxia mimetic agents, stabilise HIF-1 α protein, resulting in increased HIF-1 protein in normoxia, and inducing the expression of HIF-1 dependent hypoxia-inducible genes such as VEGF and GLUT-1 (Wang and Semenza, 1993_b; Minchenko et al., 2002; Yuan et al., 2003; Mojsilovic-Petrovic et al., 2007). Therefore I treated HMDM with DFO and CoCl₂ to determine whether versican mRNA expression is induced by these two agents as well as by hypoxia. I first confirmed the hypoxic inducibility of versican mRNA (15.5-fold) as well as VEGF (11.7-fold) and GLUT-1 (15.5-fold) (Fig 7.5). As figures 7.6 and 7.7 showed, I observed that only DFO, but not CoCl₂, significantly induced versican mRNA (10-fold), whereas the known HIF-1 regulated genes VEGF (DFO: 11-fold, CoCl₂: 10.6-fold) and GLUT-1 (DFO: 29.5-fold, CoCl₂: 6.3-fold) were up-regulated by both. Induction of versican mRNA by DFO but not by CoCl₂ which is known to up-regulate HIF-1 α strongly, suggests that versican is regulated differently to VEGF and GLUT-1, possibly via a HIF-1 independent mechanism which can be activated by hypoxia and DFO but not by CoCl₂.

An interesting study by Vengellur et al., (2005) using microarray analysis, compared the transcriptional responses to three treatments: hypoxia, DFO and CoCl₂. Comparison between these treatments showed that some mRNAs can be up-regulated by hypoxia and DFO but not by CoCl₂, such as human monocarboxylate transporter 2 (MCT2), glyceraldehyde 3-phosphate dehydrogenase (G3PDH) and homeodomain protein (Prox 1) (Vengellur et al., 2005). This finding is in line with my speculation of a HIF-1 independent mechanism which can be activated by hypoxia and DFO but not by CoCl₂. For example, a regulatory factor such as Prox1, a transcription factor essential for the development of numerous tissues and which has previously been shown to be

induced by hypoxia (Ota et al., 2007), and as described above has been shown by Vengellur et al., (2005) to be up-regulated by hypoxia and DFO but not by CoCl₂, could be involved in the hypoxic up-regulation of versican in human MDM.

Moreover, the iron chelating activity of DFO is well known (Maxwell et al., 1999; Yuan et al., 2003). A study by Woo et al., (2006) in a mouse macrophage cell line reported that DFO increases cyclooxygenase-2 (COX-2) mRNA and protein, an important inducible enzyme in inflammation, and showed that COX-2 protein expression was reduced by addition of iron, suggesting that the iron chelating function of DFO induced increased COX-2 expression (Woo et al., 2006).

Therefore I decided to investigate whether versican mRNA induction by DFO could be blocked by addition of iron in normoxic adherence-purified HMDM, to confirm that the iron chelating activity, and not some other activity or contaminant, was responsible for the ability of DFO to induce versican mRNA. As figure 7.8 showed, DFO alone significantly induced versican mRNA and that versican mRNA expression was gradually and significantly reduced as the concentration of iron was increased and finally reached a similar level as in untreated normoxic adherence purified HMDM. This finding is presumably due to the fact that as the concentration of iron increases DFO molecules become saturated with iron, and DFO's activity is reduced and finally eliminated when each molecule becomes bound to iron. This data confirms that the iron chelating activity of DFO was responsible for versican induction in human MDM, rather than any other activity.

Given the lack of inducibility of versican by LPS and CoCl₂, two known inducers of HIF-1 protein, DFO presumably acts on the versican promoter by stimulation of a different pathway. Such a candidate pathway involves p38 mitogen-activated protein kinases (MAPK) which has been recently shown to be activated by DFO due to iron chelating in the human macrophage cell line THP-1 (Fan et al., 2010). Thus I hypothesise that the MAPK pathway, which has been additionally shown to be activated by hypoxia (Blaschke et al., 2002; Xu et al., 2004), may play a role in modulation of versican mRNA up-regulation in hypoxic human MDM. Further investigation is required to study the possible role of MAPK in hypoxic regulation of versican.

Taken together, the lack of induction of versican mRNA in response to the HIF-1 inducers LPS and CoCl₂ suggests that versican induction by hypoxia is mediated via a HIF-1 independent mechanism, in contrast to what has recently been published by Asplund et al., (2010). There are many examples of genes being up-regulated by hypoxia independently of HIF-1; for example previous studies by Dong et al., (2001) and Ameri et al., (2004) reported that inhibitor of apoptosis protein-2 (IAP-2) and activating transcription factor-4 (ATF-4), respectively, respond to severe hypoxia independently of the HIF-1 signalling pathway. In addition other studies using microarray analysis of murine HIF-1 α null (HIF-1 α -/-) fibroblasts or MCF-7 cells treated with HIF-1 α siRNA showed hypoxic up-regulation of several genes in a HIF-1 independent manner (Greijer et al., 2005; Elvidge et al., 2006). However, I conclude that my current data do not completely rule out a possible role for HIF-1 in hypoxic up-regulation of versican. Further experiments to directly target HIF-1 α , such as siRNA, should be carried out to elucidate whether hypoxia-induced increases of versican is via

a HIF-1 independent mechanism. HIF-1 α siRNA data was shown by Asplund et al., 2010, but the data was weak as it lacked appropriate controls. Using the THP-1 macrophage cell line, they showed only a very minor reduction in versican mRNA level in hypoxia (approximately 20%) compared to approximately 40% reduction for the known HIF-1 regulated gene GLUT-1. It is known that macrophages rely heavily on HIFs for energy production and a range of other cellular activities even in normoxia (Cramer et al., 2003), and therefore many mRNAs are likely to be down-regulated non-specifically by such an approach (Bosco et al., 2006; Fang et al., 2009). Therefore the specificity of the modest reductions in hypoxic versican mRNA levels observed in the Asplund et al., siRNA experiments cannot be confirmed in the absence of suitable controls, (which were not carried out) such as evidence of a HIF-1 independent gene which is unaffected by siRNA treatment. Therefore further investigation on HMDM using siRNA to target HIF-1 α and appropriate controls, which is currently taking place in the lab, needs to be carried out to completely rule out HIF-1 involvement in the hypoxic up-regulation of versican.

My data showing the possible lack of an important role for HIF-1 in hypoxic induction of versican encouraged me to further investigate alternative mechanisms which could be involved in the hypoxic regulation of versican in human MDM. A study by Rahmani et al., (2005) has demonstrated that the PI3-K / PKB pathway plays a critical role in versican mRNA expression in smooth muscle cells (SMC). This study suggested that phosphorylation (inactivation) of GSK-3 β , a downstream effector of PI3-Kinase, allowed β -catenin-TCF complex formation in the nucleus and increased versican transcription in normoxia (Rahmani et al., 2005). In addition, several studies have

reported activation of the PI3-K / PKB pathway by hypoxia and shown that this mechanism is involved in the induction of target genes such as MMP-7 (Alvarez-Tejado et al., 2001; Xu et al., 2004; Deguchi et al., 2009; Liu et al., 2010). Therefore I decided to investigate the possible role of PI3-Kinase in hypoxic up-regulation of versican mRNA. As figure 7.9 showed, the PI3-Kinase inhibitor LY290042 markedly reduced the hypoxic induction of versican mRNA. Furthermore, a similar effect was observed when wortmannin, a different inhibitor of the PI3-Kinase pathway, was used. Based on this preliminary finding, I speculate that hypoxic induction of versican mRNA may be dependent, at least in part, on the PI3-Kinase pathway in HMDM.

In addition, as previously outlined, different studies have demonstrated hypoxiainduced increases of β -catenin via the PI3-Kinase pathway (Demir et al., 2009; Du et al., 2010; Mazumdar et al., 2010) which in turn results in the nuclear accumulation of β -catenin, which can bind to the TCF transcription factor on the promoter and increase expression of target genes in different cells, including human macrophages and smooth muscle cells (Polakis, 2000; Rahmani et al., 2006; Deguchi et al., 2009; Cui et al., 2010). In preliminary immunoblotting experiments to study whether nuclear β catenin accumulation could be detected under hypoxia in my hands, I observed an increase in β -catenin in the nuclei of hypoxic human monocytic U937 cells (Fig 7.10). These findings are in line with my hypothesis that hypoxic activation of PI3-Kinase may lead to the nuclear accumulation of β -catenin, a downstream effector of the PI3-Kinase pathway and consequently cause β -catenin-TCF complex formation on the versican promoter, resulting in hypoxia-induced versican transcription in HMDM. Other studies have shown increased HIF-1 via hypoxic activation of the PI3-K / PKB

pathway in different cell lines (Chen et al., 2001; Xu et al., 2004). However, in my experiments, no decrease in the hypoxic induction of GLUT-1 mRNA, a well known HIF-1 regulated gene, was observed in HMDM treated with LY294002 or wortmannin (Fig 7.9). This finding firstly shows that possible activation of HIF-1 by the PI3-Kinase pathway in hypoxia may be confined to specific cell types, as has been previously shown by Mazure et al., (1997) and Lee et al., (2006), and secondly shows that versican is regulated differently to GLUT-1, possibly via a mechanism such as β -catenin-TCF complex formation which can be activated by PI3-Kinase pathway rather than HIF-1 in hypoxic HMDM.
Conclusion and future work

In the present study, I provided evidence that primary human macrophages are the principle peripheral blood mononuclear cell showing expression and hypoxic up-regulation of versican mRNA and protein. In addition, my data demonstrate that increases observed in versican mRNA in hypoxic adherence-purified HMDM are due to transcriptional up-regulation rather than increased mRNA stability in hypoxia. This finding led me to analyse the versican promoter by making, and transfecting primary macrophages with, a series of versican promoter deletion constructs. The data from these experiments indicated two regions which are required for high expression from the promoter in hypoxic primary human macrophages. Further experiments suggested that high-level transcription of the versican promoter in hypoxia appears to occur via a HIF-1 independent mechanism which can be activated by hypoxia and DFO not by CoCl₂. I also carried out inhibitor experiments which lead me to propose a possible role for the PI3-Kinase pathway in the hypoxic up-regulation of versican.

However, a number of further experiments are required to confirm and extend the findings presented in this thesis. As I discussed earlier, studies of versican induction in HIF-1 α knockout cells or in HMDM transfected with HIF-1 α siRNA are required to finally confirm whether HIF-1 is or is not responsible for the induction of versican in hypoxia. Likewise, further experiments are required to confirm whether MAPK or PI3-Kinase pathway activation by hypoxia and subsequent β -catenin-TCF complex formation are essential for versican transcription in HMDM.

197

In addition, site-directed mutagenesis on the regions identified in the present study using versican promoter deletion reporter constructs could be used to indicate the specific transcription factor binding-sites which are essential for high level expression from the promoter in hypoxic primary human macrophages. Electrophoretic mobility shift and supershift assays could reveal specific binding of such transcription factors to oligonucleotides corresponding to their potential binding sites in the versican promoter, and Chromatin Immunoprecipitation (ChIP) could be used to confirm the ability of these factors to bind to the promoter under hypoxic conditions in living cells.

A better understanding of how versican is regulated by hypoxia in human macrophages will hopefully be helpful for the development of future therapies for a range of different disease such as vascular disorders, including atherosclerosis, where versican accumulation plays a key role (Wight and Merrilees, 2004). As macrophages have been shown to accumulate in the areas of low oxygen tension where versican is up-regulated, the knowledge of how the versican promoter is induced by hypoxia by elucidation of the hypoxia responsive elements could be an additional advantage for future tumour gene therapy whereby a therapeutic gene could be engineered to be regulated by the hypoxia responsive promoter. Macrophages transfected with this construct could be used in the delivery of the therapeutic gene to radiotherapy and chemotherapy resistant hypoxic tumour sites where the gene would be locally induced. In addition, hypoxic up-regulation of versican by human macrophages which are recruited and retained in hypoxic and necrotic sites could be a potential prognostic factor in patients with malignants tumours.

198

References

Adams, J.C., and Watt, F.M. (1993) Regulation of development and differentiation by the extracellular matrix. Development **117**, 1183-1198.

Adams, J.M., Difazio, L.T., Rolandelli, R.H., Lujan, J.J., Hasko, G., Csoka, B., Selmeczy, Z., and Nemeth, Z.H. (2009) HIF-1: a key mediator in hypoxia. Acta Physiol Hung **96**, 19-28.

Adelsman, M.A., McCarthy, J.B., and Shimizu, Y. (1999) Stimulation of beta1-integrin function by epidermal growth factor and heregulin-beta has distinct requirements for erbB2 but a similar dependence on phosphoinositide 3-OH kinase. Mol Biol Cell **10**, 2861-2878.

Agani, F., and Semenza, G.L. (1998) Mersalyl is a novel inducer of vascular endothelial growth factor gene expression and hypoxia-inducible factor 1 activity. Mol Pharmacol **54**, 749-754.

Airley, R., Loncaster, J., Davidson, S., Bromley, M., Roberts, S., Patterson, A., Hunter, R., Stratford, I., and West, C. (2001) Glucose transporter glut-1 expression correlates with tumor hypoxia and predicts metastasis-free survival in advanced carcinoma of the cervix. Clin Cancer Res **7**, 928-934.

Akira, S., Takeda, K., and Kaisho, T. (2001) Toll-like receptors: critical proteins linking innate and acquired immunity. Nat Immunol **2**, 675-680.

Akira, S., Yamamoto, M., and Takeda, K. (2003) Role of adapters in Toll-like receptor signalling. Biochem Soc Trans **31**, 637-642.

Alam, H., Maizels, E.T., Park, Y., Ghaey, S., Feiger, Z.J., Chandel, N.S., and Hunzicker-Dunn, M. (2004) Follicle-stimulating hormone activation of hypoxia-inducible factor-1 by the phosphatidylinositol 3-kinase/AKT/Ras homolog enriched in brain (Rheb)/mammalian target of rapamycin (mTOR) pathway is necessary for induction of select protein markers of follicular differentiation. J Biol Chem **279**, 19431-19440.

Albina, J.E., Henry, W.L., Jr., Mastrofrancesco, B., Martin, B.A., and Reichner, J.S. (1995) Macrophage activation by culture in an anoxic environment. J Immunol **155**, 4391-4396.

Albina, J.E., Mastrofrancesco, B., Vessella, J.A., Louis, C.A., Henry, W.L., Jr., and Reichner, J.S. (2001) HIF-1 expression in healing wounds: HIF-1alpha induction in primary inflammatory cells by TNF-alpha. Am J Physiol Cell Physiol **281**, C1971-1977.

Alexis, N.E., Lay, J.C., Zeman, K.L., Geiser, M., Kapp, N., and Bennett, W.D. (2006) In vivo particle uptake by airway macrophages in healthy volunteers. Am J Respir Cell Mol Biol **34**, 305-313.

Alvarez-Tejado, M., Naranjo-Suarez, S., Jimenez, C., Carrera, A.C., Landazuri, M.O., and del Peso, L. (2001) Hypoxia induces the activation of the phosphatidylinositol 3-kinase/Akt cell survival pathway in PC12 cells: protective role in apoptosis. J Biol Chem **276**, 22368-22374.

Ameri, K., Burke, B., Lewis, C.E., and Harris, A.L. (2002) Regulation of a rat VL30 element in human breast cancer cells in hypoxia and anoxia: role of HIF-1. Br J Cancer **87**, 1173-1181.

Ameri, K., Lewis, C.E., Raida, M., Sowter, H., Hai, T., and Harris, A.L. (2004) Anoxic induction of ATF-4 through HIF-1-independent pathways of protein stabilization in human cancer cells. Blood **103**, 1876-1882.

Arsham, A.M., Howell, J.J., and Simon, M.C. (2003) A novel hypoxia-inducible factorindependent hypoxic response regulating mammalian target of rapamycin and its targets. J Biol Chem **278**, 29655-29660.

Arslan, F., Bosserhoff, A.K., Nickl-Jockschat, T., Doerfelt, A., Bogdahn, U., and Hau, P. (2007) The role of versican isoforms V0/V1 in glioma migration mediated by transforming growth factor-beta2. Br J Cancer **96**, 1560-1568.

Asher, R.A., Morgenstern, D.A., Shearer, M.C., Adcock, K.H., Pesheva, P., and Fawcett, J.W. (2002) Versican is upregulated in CNS injury and is a product of oligodendrocyte lineage cells. J Neurosci **22**, 2225-2236.

Asplund, A., Stillemark-Billton, P., Larsson, E., Rydberg, E.K., Moses, J., Hulten, L.M., Fagerberg, B., Camejo, G., and Bondjers, G. (2010) Hypoxic regulation of secreted proteoglycans in macrophages. Glycobiology **20**, 33-40.

Badylak, S.F. (2002) The extracellular matrix as a scaffold for tissue reconstruction. Semin Cell Dev Biol **13**, 377-383.

Bakheet, T., Frevel, M., Williams, B.R., Greer, W., and Khabar, K.S. (2001) ARED: human AU-rich element-containing mRNA database reveals an unexpectedly diverse functional repertoire of encoded proteins. Nucleic Acids Res **29**, 246-254.

Bandyopadhyay, R.S., Phelan, M., and Faller, D.V. (1995) Hypoxia induces AP-1regulated genes and AP-1 transcription factor binding in human endothelial and other cell types. Biochim Biophys Acta **1264**, 72-78.

Barreau, C., Paillard, L., and Osborne, H.B. (2005) AU-rich elements and associated factors: are there unifying principles? Nucleic Acids Res **33**, 7138-7150.

Beekhuizen, H., and van Furth, R. (1993) Monocyte adherence to human vascular endothelium. J Leukoc Biol **54**, 363-378.

Beitner-Johnson, D., and Millhorn, D.E. (1998) Hypoxia induces phosphorylation of the cyclic AMP response element-binding protein by a novel signaling mechanism. J Biol Chem **273**, 19834-19839.

Benizri, E., Ginouves, A., and Berra, E. (2008) The magic of the hypoxia-signaling cascade. Cell Mol Life Sci **65**, 1133-1149.

Benoit, M., Desnues, B., and Mege, J.L. (2008) Macrophage polarization in bacterial infections. J Immunol **181**, 3733-3739.

Berrier, A.L., and Yamada, K.M. (2007) Cell-matrix adhesion. J Cell Physiol **213**, 565-573.

Berton, G., and Lowell, C.A. (1999) Integrin signalling in neutrophils and macrophages. Cell Signal **11**, 621-635.

Bertout, J.A., Patel, S.A., and Simon, M.C. (2008) The impact of O2 availability on human cancer. Nat Rev Cancer **8**, 967-975.

Bingle, L., Brown, N.J., and Lewis, C.E. (2002) The role of tumour-associated macrophages in tumour progression: implications for new anticancer therapies. J Pathol **196**, 254-265.

Bingle, L., Lewis, C.E., Corke, K.P., Reed, M.W., and Brown, N.J. (2006) Macrophages promote angiogenesis in human breast tumour spheroids in vivo. Br J Cancer **94**, 101-107.

Bjornheden, T., Levin, M., Evaldsson, M., and Wiklund, O. (1999) Evidence of hypoxic areas within the arterial wall in vivo. Arterioscler Thromb Vasc Biol **19**, 870-876.

Blaschke, F., Stawowy, P., Goetze, S., Hintz, O., Grafe, M., Kintscher, U., Fleck, E., and Graf, K. (2002) Hypoxia activates beta(1)-integrin via ERK 1/2 and p38 MAP kinase in human vascular smooth muscle cells. Biochem Biophys Res Commun **296**, 890-896.

Blouin, C.C., Page, E.L., Soucy, G.M., and Richard, D.E. (2004) Hypoxic gene activation by lipopolysaccharide in macrophages: implication of hypoxia-inducible factor 1alpha. Blood **103**, 1124-1130.

Bode-Lesniewska, B., Dours-Zimmermann, M.T., Odermatt, B.F., Briner, J., Heitz, P.U., and Zimmermann, D.R. (1996) Distribution of the large aggregating proteoglycan versican in adult human tissues. J Histochem Cytochem **44**, 303-312.

Bosco, M.C., Puppo, M., Santangelo, C., Anfosso, L., Pfeffer, U., Fardin, P., Battaglia, F., and Varesio, L. (2006) Hypoxia modifies the transcriptome of primary human monocytes: modulation of novel immune-related genes and identification of CC-chemokine ligand 20 as a new hypoxia-inducible gene. J Immunol **177**, 1941-1955.

Bossi, G., and Sacchi, A. (2007) Restoration of wild-type p53 function in human cancer: relevance for tumor therapy. Head Neck **29**, 272-284.

Boudreau, N., and Bissell, M.J. (1998) Extracellular matrix signaling: integration of form and function in normal and malignant cells. Curr Opin Cell Biol **10**, 640-646.

Brahimi-Horn, C., and Pouyssegur, J. (2006) The role of the hypoxia-inducible factor in tumor metabolism growth and invasion. Bull Cancer **93**, E73-80.

Breuer, W., Ermers, M.J., Pootrakul, P., Abramov, A., Hershko, C., and Cabantchik, Z.I. (2001) Desferrioxamine-chelatable iron, a component of serum non-transferrin-bound iron, used for assessing chelation therapy. Blood **97**, 792-798.

Brizel, D.M., Dodge, R.K., Clough, R.W., and Dewhirst, M.W. (1999) Oxygenation of head and neck cancer: changes during radiotherapy and impact on treatment outcome. Radiother Oncol **53**, 113-117.

Burke, B., Giannoudis, A., Corke, K.P., Gill, D., Wells, M., Ziegler-Heitbrock, L., and Lewis, C.E. (2003) Hypoxia-induced gene expression in human macrophages: implications for ischemic tissues and hypoxia-regulated gene therapy. Am J Pathol **163**, 1233-1243.

Burke, B., Tang, N., Corke, K.P., Tazzyman, D., Ameri, K., Wells, M., and Lewis, C.E. (2002) Expression of HIF-1alpha by human macrophages: implications for the use of macrophages in hypoxia-regulated cancer gene therapy. J Pathol **196**, 204-212.

Cabodi, S., Moro, L., Bergatto, E., Boeri Erba, E., Di Stefano, P., Turco, E., Tarone, G., and Defilippi, P. (2004) Integrin regulation of epidermal growth factor (EGF) receptor and of EGF-dependent responses. Biochem Soc Trans **32**, 438-442.

Caldwell, C.C., Kojima, H., Lukashev, D., Armstrong, J., Farber, M., Apasov, S.G., and Sitkovsky, M.V. (2001) Differential effects of physiologically relevant hypoxic conditions on T lymphocyte development and effector functions. J Immunol **167**, 6140-6149.

Calle, Y., Burns, S., Thrasher, A.J., and Jones, G.E. (2006) The leukocyte podosome. Eur J Cell Biol **85**, 151-157.

Camejo, G., Olsson, U., Hurt-Camejo, E., Baharamian, N., and Bondjers, G. (2002) The extracellular matrix on atherogenesis and diabetes-associated vascular disease. Atheroscler Suppl **3**, 3-9.

Campbell, N.E., Kellenberger, L., Greenaway, J., Moorehead, R.A., Linnerth-Petrik, N.M., and Petrik, J. (2010) Extracellular matrix proteins and tumor angiogenesis. J Oncol **2010**, *Article ID 586905*.

Cardoso, L.E., Little, P.J., Ballinger, M.L., Chan, C.K., Braun, K.R., Potter-Perigo, S., Bornfeldt, K.E., Kinsella, M.G., and Wight, T.N. (2010) Platelet-derived growth factor differentially regulates the expression and post-translational modification of versican by arterial smooth muscle cells through distinct protein kinase C and extracellular signal-regulated kinase pathways. J Biol Chem **285**, 6987-6995.

Carmeliet, P., Dor, Y., Herbert, J.M., Fukumura, D., Brusselmans, K., Dewerchin, M., Neeman, M., Bono, F., Abramovitch, R., Maxwell, P., Koch, C.J., Ratcliffe, P., Moons, L., Jain, R.K., Collen, D., and Keshert, E. (1998) Role of HIF-1alpha in hypoxia-mediated apoptosis, cell proliferation and tumour angiogenesis. Nature **394**, 485-490.

Carulli, D., Laabs, T., Geller, H.M., and Fawcett, J.W. (2005) Chondroitin sulfate proteoglycans in neural development and regeneration. Curr Opin Neurobiol **15**, 116-120.

Cattaruzza, S., Schiappacassi, M., Ljungberg-Rose, A., Spessotto, P., Perissinotto, D., Morgelin, M., Mucignat, M.T., Colombatti, A., and Perris, R. (2002) Distribution of PG-M/versican variants in human tissues and de novo expression of isoform V3 upon endothelial cell activation, migration, and neoangiogenesis in vitro. J Biol Chem **277**, 47626-47635.

Chen, C.Y., and Shyu, A.B. (1995) AU-rich elements: characterization and importance in mRNA degradation. Trends Biochem Sci **20**, 465-470.

Chen, E.Y., Mazure, N.M., Cooper, J.A., and Giaccia, A.J. (2001) Hypoxia activates a platelet-derived growth factor receptor/phosphatidylinositol 3-kinase/Akt pathway that results in glycogen synthase kinase-3 inactivation. Cancer Res **61**, 2429-2433.

Chen, G., Gharib, T.G., Huang, C.C., Taylor, J.M., Misek, D.E., Kardia, S.L., Giordano, T.J., Iannettoni, M.D., Orringer, M.B., Hanash, S.M., and Beer, D.G. (2002) Discordant protein and mRNA expression in lung adenocarcinomas. Mol Cell Proteomics **1**, 304-313.

Clevers, H., and van de Wetering, M. (1997) TCF/LEF factor earn their wings. Trends Genet **13**, 485-489.

Cockman, M.E., Webb, J.D., and Ratcliffe, P.J. (2009) FIH-dependent asparaginyl hydroxylation of ankyrin repeat domain-containing proteins. Ann N Y Acad Sci **1177**, 9-18.

Collot-Teixeira, S., Martin, J., McDermott-Roe, C., Poston, R., and McGregor, J.L. (2007) CD36 and macrophages in atherosclerosis. Cardiovasc Res **75**, 468-477.

Connolly, E., Braunstein, S., Formenti, S., and Schneider, R.J. (2006) Hypoxia inhibits protein synthesis through a 4E-BP1 and elongation factor 2 kinase pathway controlled by mTOR and uncoupled in breast cancer cells. Mol Cell Biol **26**, 3955-3965.

Cougoule, C., Wiedemann, A., Lim, J., and Caron, E. (2004) Phagocytosis, an alternative model system for the study of cell adhesion. Semin Cell Dev Biol **15**, 679-689.

Covert, M.W., Leung, T.H., Gaston, J.E., and Baltimore, D. (2005) Achieving stability of lipopolysaccharide-induced NF-kappaB activation. Science **309**, 1854-1857.

Cramer, T., Yamanishi, Y., Clausen, B.E., Forster, I., Pawlinski, R., Mackman, N., Haase, V.H., Jaenisch, R., Corr, M., Nizet, V., Firestein, G.S., Gerber, H.P., Ferrara, N., and Johnson, R.S. (2003) HIF-1alpha is essential for myeloid cell-mediated inflammation. Cell **112**, 645-657.

Crowell, J.A., Kusserow, B.K., and Nyborg, W.L. (1977) Functional changes in white blood cells after microsonation. Ultrasound Med Biol **3**, 185-190.

Crowther, M., Brown, N.J., Bishop, E.T., and Lewis, C.E. (2001) Microenvironmental influence on macrophage regulation of angiogenesis in wounds and malignant tumors. J Leukoc Biol **70**, 478-490.

Cui, X.P., Xing, Y., Chen, J.M., Dong, S.W., Ying, D.J., and Yew, D.T. (2010) Wnt/betacatenin is involved in the proliferation of hippocampal neural stem cells induced by hypoxia. Ir J Med Sci. **180**, 387-93

Cummins, E.P., and Taylor, C.T. (2005) Hypoxia-responsive transcription factors. Pflugers Arch **450**, 363-371.

Daley, W.P., Peters, S.B., and Larsen, M. (2008) Extracellular matrix dynamics in development and regenerative medicine. J Cell Sci **121**, 255-264.

Deguchi, J.O., Yamazaki, H., Aikawa, E., and Aikawa, M. (2009) Chronic hypoxia activates the Akt and beta-catenin pathways in human macrophages. Arterioscler Thromb Vasc Biol **29**, 1664-1670.

Demir, R., Dimmler, A., Naschberger, E., Demir, I., Papadopoulos, T., Melling, N., Sturzl, M., and Hohenberger, W. (2009) Malignant progression of invasive tumour cells seen in hypoxia present an accumulation of beta-catenin in the nucleus at the tumour front. Exp Mol Pathol **87**, 109-116.

Dentan, C., Lesnik, P., Chapman, M.J., and Ninio, E. (1996) Phagocytic activation induces formation of platelet-activating factor in human monocyte-derived macrophages and in macrophage-derived foam cells. Relevance to the inflammatory reaction in atherogenesis. Eur J Biochem **236**, 48-55.

Dery, M.A., Michaud, M.D., and Richard, D.E. (2005) Hypoxia-inducible factor 1: regulation by hypoxic and non-hypoxic activators. Int J Biochem Cell Biol **37**, 535-540.

Dijkstra, C.D., Dopp, E.A., Joling, P., and Kraal, G. (1985) The heterogeneity of mononuclear phagocytes in lymphoid organs: distinct macrophage subpopulations in the rat recognized by monoclonal antibodies ED1, ED2 and ED3. Immunology **54**, 589-599.

Domenzain, C., Docampo, M.J., Serra, M., Miquel, L., and Bassols, A. (2003) Differential expression of versican isoforms is a component of the human melanoma cell differentiation process. Biochim Biophys Acta **1642**, 107-114.

Domenzain, C., Hernandez, D., Miquel-Serra, L., Docampo, M.J., Badenas, C., Fabra, A., and Bassols, A. (2009) Structure and regulation of the versican promoter: the versican promoter is regulated by AP-1 and TCF transcription factors in invasive human melanoma cells. J Biol Chem **284**, 12306-12317.

Dong, Z., Venkatachalam, M.A., Wang, J., Patel, Y., Saikumar, P., Semenza, G.L., Force, T., and Nishiyama, J. (2001) Up-regulation of apoptosis inhibitory protein IAP-2 by hypoxia. Hif-1-independent mechanisms. J Biol Chem **276**, 18702-18709.

Dongiovanni, P., Valenti, L., Ludovica Fracanzani, A., Gatti, S., Cairo, G., and Fargion, S. (2008) Iron depletion by deferoxamine up-regulates glucose uptake and insulin signaling in hepatoma cells and in rat liver. Am J Pathol **172**, 738-747.

Doran, D.M., Kulkarni-Datar, K., Cool, D.R., and Brown, T.L. (2010) Hypoxia activates constitutive luciferase reporter constructs. Biochimie **93**, 361-368.

Dours-Zimmermann, M.T., and Zimmermann, D.R. (1994) A novel glycosaminoglycan attachment domain identified in two alternative splice variants of human versican. J Biol Chem **269**, 32992-32998.

du Cros, D.L., LeBaron, R.G., and Couchman, J.R. (1995) Association of versican with dermal matrices and its potential role in hair follicle development and cycling. J Invest Dermatol **105**, 426-431.

Du, R., Huang, C., Bi, Q., Zhai, Y., Xia, L., Liu, J., Sun, S., and Fan, D. (2010) URG11 mediates hypoxia-induced epithelial-to-mesenchymal transition by modulation of E-cadherin and beta-catenin. Biochem Biophys Res Commun **391**, 135-141.

Dutt, S., Kleber, M., Matasci, M., Sommer, L., and Zimmermann, D.R. (2006) Versican V0 and V1 guide migratory neural crest cells. J Biol Chem **281**, 12123-12131.

East, L., and Isacke, C.M. (2002) The mannose receptor family. Biochim Biophys Acta **1572**, 364-386.

Edwards, J.P., Zhang, X., Frauwirth, K.A., and Mosser, D.M. (2006) Biochemical and functional characterization of three activated macrophage populations. J Leukoc Biol **80**, 1298-1307.

Egger, M., Schgoer, W., Beer, A.G., Jeschke, J., Leierer, J., Theurl, M., Frauscher, S., Tepper, O.M., Niederwanger, A., Ritsch, A., Kearney, M., Wanschitz, J., Gurtner, G.C., Fischer-Colbrie, R., Weiss, G., Piza-Katzer, H., Losordo, D.W., Patsch, J.R., Schratzberger, P., and Kirchmair, R. (2007) Hypoxia up-regulates the angiogenic cytokine secretoneurin via an HIF-1alpha- and basic FGF-dependent pathway in muscle cells. FASEB J **21**, 2906-2917.

Elbarghati, L., Murdoch, C., and Lewis, C.E. (2008) Effects of hypoxia on transcription factor expression in human monocytes and macrophages. Immunobiology **213**, 899-908.

Elvidge, G.P., Glenny, L., Appelhoff, R.J., Ratcliffe, P.J., Ragoussis, J., and Gleadle, J.M. (2006) Concordant regulation of gene expression by hypoxia and 2-oxoglutaratedependent dioxygenase inhibition: the role of HIF-1alpha, HIF-2alpha, and other pathways. J Biol Chem **281**, 15215-15226.

Ema, M., Hirota, K., Mimura, J., Abe, H., Yodoi, J., Sogawa, K., Poellinger, L., and Fujii-Kuriyama, Y. (1999) Molecular mechanisms of transcription activation by HLF and HIF1alpha in response to hypoxia: their stabilization and redox signal-induced interaction with CBP/p300. EMBO J **18**, 1905-1914.

Ema, M., Taya, S., Yokotani, N., Sogawa, K., Matsuda, Y., and Fujii-Kuriyama, Y. (1997) A novel bHLH-PAS factor with close sequence similarity to hypoxia-inducible factor 1alpha regulates the VEGF expression and is potentially involved in lung and vascular development. Proc Natl Acad Sci U S A **94**, 4273-4278.

Epstein, A.C., Gleadle, J.M., McNeill, L.A., Hewitson, K.S., O'Rourke, J., Mole, D.R., Mukherji, M., Metzen, E., Wilson, M.I., Dhanda, A., Tian, Y.M., Masson, N., Hamilton, D.L., Jaakkola, P., Barstead, R., Hodgkin, J., Maxwell, P.H., Pugh, C.W., Schofield, C.J., and Ratcliffe, P.J. (2001) C. elegans EGL-9 and mammalian homologs define a family of dioxygenases that regulate HIF by prolyl hydroxylation. Cell **107**, 43-54.

Evanko, S.P., Angello, J.C., and Wight, T.N. (1999) Formation of hyaluronan- and versican-rich pericellular matrix is required for proliferation and migration of vascular smooth muscle cells. Arterioscler Thromb Vasc Biol **19**, 1004-1013.

Evanko, S.P., Johnson, P.Y., Braun, K.R., Underhill, C.B., Dudhia, J., and Wight, T.N. (2001) Platelet-derived growth factor stimulates the formation of versican-hyaluronan aggregates and pericellular matrix expansion in arterial smooth muscle cells. Arch Biochem Biophys **394**, 29-38.

Fan, Y., Wang, J., Wei, L., He, B., Wang, C., and Wang, B. (2010) Iron deficiency activates pro-inflammatory signaling in macrophages and foam cells via the p38 MAPK-NF-kappaB pathway. Int J Cardiol. *Article in Press*

Fang, H.Y., Hughes, R., Murdoch, C., Coffelt, S.B., Biswas, S.K., Harris, A.L., Johnson, R.S., Imityaz, H.Z., Simon, M.C., Fredlund, E., Greten, F.R., Rius, J., and Lewis, C.E. (2009) Hypoxia-inducible factors 1 and 2 are important transcriptional effectors in primary macrophages experiencing hypoxia. Blood **114**, 844-859.

Fantozzi, I., Zhang, S., Platoshyn, O., Remillard, C.V., Cowling, R.T., and Yuan, J.X. (2003) Hypoxia increases AP-1 binding activity by enhancing capacitative Ca2+ entry in human pulmonary artery endothelial cells. Am J Physiol Lung Cell Mol Physiol **285**, L1233-1245.

Fidler, P.S., Schuette, K., Asher, R.A., Dobbertin, A., Thornton, S.R., Calle-Patino, Y., Muir, E., Levine, J.M., Geller, H.M., Rogers, J.H., Faissner, A., and Fawcett, J.W. (1999) Comparing astrocytic cell lines that are inhibitory or permissive for axon growth: the major axon-inhibitory proteoglycan is NG2. J Neurosci **19**, 8778-8788.

Fogg, D.K., Sibon, C., Miled, C., Jung, S., Aucouturier, P., Littman, D.R., Cumano, A., and Geissmann, F. (2006) A clonogenic bone marrow progenitor specific for macrophages and dendritic cells. Science **311**, 83-87.

Foldager, C.B., Munir, S., Ulrik-Vinther, M., Soballe, K., Bunger, C., and Lind, M. (2009) Validation of suitable house keeping genes for hypoxia-cultured human chondrocytes. BMC Mol Biol **10**, 94.

Fraser, I.P., Koziel, H., and Ezekowitz, R.A. (1998) The serum mannose-binding protein and the macrophage mannose receptor are pattern recognition molecules that link innate and adaptive immunity. Semin Immunol **10**, 363-372.

Frede, S., Stockmann, C., Freitag, P., and Fandrey, J. (2006) Bacterial lipopolysaccharide induces HIF-1 activation in human monocytes via p44/42 MAPK and NF-kappaB. Biochem J **396**, 517-527.

Freeman, B.A. (2000) Oxygen: the air-borne nutrient that both sustains and threatens life. Nutrition **16**, 478-480.

Fry, M.J. (2001) Phosphoinositide 3-kinase signalling in breast cancer: how big a role might it play? Breast Cancer Res **3**, 304-312.

Fujihara, M., Muroi, M., Tanamoto, K., Suzuki, T., Azuma, H., and Ikeda, H. (2003) Molecular mechanisms of macrophage activation and deactivation by lipopolysaccharide: roles of the receptor complex. Pharmacol Ther **100**, 171-194. Gao, J.J., Filla, M.B., Fultz, M.J., Vogel, S.N., Russell, S.W., and Murphy, W.J. (1998) Autocrine/paracrine IFN-alphabeta mediates the lipopolysaccharide-induced activation of transcription factor Stat1alpha in mouse macrophages: pivotal role of Stat1alpha in induction of the inducible nitric oxide synthase gene. J Immunol **161**, 4803-4810.

Gao, W., Gao, Y., Zhang, G., Song, L., Sun, B., and Shi, J. (2005) Hypoxia-induced expression of HIF-1alpha and its target genes in umbilical venous endothelial cells of Tibetans and immigrant Han. Comp Biochem Physiol C Toxicol Pharmacol **141**, 93-100.

Garayoa, M., Martinez, A., Lee, S., Pio, R., An, W.G., Neckers, L., Trepel, J., Montuenga, L.M., Ryan, H., Johnson, R., Gassmann, M., and Cuttitta, F. (2000) Hypoxia-inducible factor-1 (HIF-1) up-regulates adrenomedullin expression in human tumor cell lines during oxygen deprivation: a possible promotion mechanism of carcinogenesis. Mol Endocrinol **14**, 848-862.

Ghezzi, P., Dinarello, C.A., Bianchi, M., Rosandich, M.E., Repine, J.E., and White, C.W. (1991) Hypoxia increases production of interleukin-1 and tumor necrosis factor by human mononuclear cells. Cytokine **3**, 189-194.

Gleadle, J.M., and Ratcliffe, P.J. (1997) Induction of hypoxia-inducible factor-1, erythropoietin, vascular endothelial growth factor, and glucose transporter-1 by hypoxia: evidence against a regulatory role for Src kinase. Blood **89**, 503-509.

Goerdt, S., and Orfanos, C.E. (1999) Other functions, other genes: alternative activation of antigen-presenting cells. Immunity **10**, 137-142.

Goldberg-Cohen, I., Furneauxb, H., and Levy, A.P. (2002) A 40-bp RNA element that mediates stabilization of vascular endothelial growth factor mRNA by HuR. J Biol Chem **277**, 13635-13640.

Gomes, N.E., Brunialti, M.K., Mendes, M.E., Freudenberg, M., Galanos, C., and Salomao, R. (2010) Lipopolysaccharide-induced expression of cell surface receptors and cell activation of neutrophils and monocytes in whole human blood. Braz J Med Biol Res **43**, 853-858.

Gordon, S. (1998) The role of the macrophage in immune regulation. Res Immunol **149**, 685-688.

Gordon, S. (2002) Pattern recognition receptors: doubling up for the innate immune response. Cell **111**, 927-930.

Gordon, S. (2003) Alternative activation of macrophages. Nat Rev Immunol 3, 23-35.

Gordon, S., and Martinez, F.O. (2010) Alternative activation of macrophages: mechanism and functions. Immunity **32**, 593-604.127.

Gordon, S., and Taylor, P.R. (2005) Monocyte and macrophage heterogeneity. Nat Rev Immunol **5**, 953-964.

Goswami, S., Sahai, E., Wyckoff, J.B., Cammer, M., Cox, D., Pixley, F.J., Stanley, E.R., Segall, J.E., and Condeelis, J.S. (2005) Macrophages promote the invasion of breast carcinoma cells via a colony-stimulating factor-1/epidermal growth factor paracrine loop. Cancer Res **65**, 5278-5283.

Gray, L.H., Conger, A.D., Ebert, M., Hornsey, S., and Scott, O.C. (1953) The concentration of oxygen dissolved in tissues at the time of irradiation as a factor in radiotherapy. Br J Radiol **26**, 638-648.

Greijer, A.E., van der Groep, P., Kemming, D., Shvarts, A., Semenza, G.L., Meijer, G.A., van de Wiel, M.A., Belien, J.A., van Diest, P.J., and van der Wall, E. (2005) Up-regulation of gene expression by hypoxia is mediated predominantly by hypoxia-inducible factor 1 (HIF-1). J Pathol **206**, 291-304.

Grey, H.M., Colon, S.M., and Chesnut, R.W. (1982) Requirements for the processing of antigen by antigen-presenting B cells. II. Biochemical comparison of the fate of antigen in B cell tumors and macrophages. J Immunol **129**, 2389-2395.

Griffiths, L., Binley, K., Iqball, S., Kan, O., Maxwell, P., Ratcliffe, P., Lewis, C., Harris, A., Kingsman, S., and Naylor, S. (2000) The macrophage - a novel system to deliver gene therapy to pathological hypoxia. Gene Ther **7**, 255-262.

Gry, M., Rimini, R., Stromberg, S., Asplund, A., Ponten, F., Uhlen, M., and Nilsson, P. (2009) Correlations between RNA and protein expression profiles in 23 human cell lines. BMC Genomics **10**, 365.

Guo, M., Song, L.P., Jiang, Y., Liu, W., Yu, Y., and Chen, G.Q. (2006) Hypoxia-mimetic agents desferrioxamine and cobalt chloride induce leukemic cell apoptosis through different hypoxia-inducible factor-1alpha independent mechanisms. Apoptosis **11**, 67-77.

Guo, Y., Xiao, P., Lei, S., Deng, F., Xiao, G.G., Liu, Y., Chen, X., Li, L., Wu, S., Chen, Y., Jiang, H., Tan, L., Xie, J., Zhu, X., Liang, S., and Deng, H. (2008) How is mRNA expression predictive for protein expression? A correlation study on human circulating monocytes. Acta Biochim Biophys Sin (Shanghai) **40**, 426-436.

Gussow, D., Rein, R., Ginjaar, I., Hochstenbach, F., Seemann, G., Kottman, A., and Ploegh, H.L. (1987) The human beta 2-microglobulin gene. Primary structure and definition of the transcriptional unit. J Immunol **139**, 3132-3138.

Haddad, J.J. (2004) Hypoxia and the regulation of mitogen-activated protein kinases: gene transcription and the assessment of potential pharmacologic therapeutic interventions. Int Immunopharmacol **4**, 1249-1285.

Hajjar, A.M., Ernst, R.K., Tsai, J.H., Wilson, C.B., and Miller, S.I. (2002) Human Toll-like receptor 4 recognizes host-specific LPS modifications. Nat Immunol **3**, 354-359.

Hansch, A., Stiehl, P., and Geiler, G. (1996) [Quantification of macrophages and granulocytes at the joint cartilage-pannus junction in rheumatoid arthritis]. Z Rheumatol **55**, 401-409.

Hansson, G.K., Jonasson, L., Lojsthed, B., Stemme, S., Kocher, O., and Gabbiani, G. (1988) Localization of T lymphocytes and macrophages in fibrous and complicated human atherosclerotic plaques. Atherosclerosis **72**, 135-141.

Hardingham, T.E., and Fosang, A.J. (1992) Proteoglycans: many forms and many functions. FASEB J **6**, 861-870.

Harris, C.C. (1996) Structure and function of the p53 tumor suppressor gene: clues for rational cancer therapeutic strategies. J Natl Cancer Inst **88**, 1442-1455.

Harwood, A.J. (2001) Regulation of GSK-3: a cellular multiprocessor. Cell **105**, 821-824.

Hasegawa, K., Yoneda, M., Kuwabara, H., Miyaishi, O., Itano, N., Ohno, A., Zako, M., and Isogai, Z. (2007) Versican, a major hyaluronan-binding component in the dermis, loses its hyaluronan-binding ability in solar elastosis. J Invest Dermatol **127**, 1657-1663.

Hayashi, M., Sakata, M., Takeda, T., Yamamoto, T., Okamoto, Y., Sawada, K., Kimura, A., Minekawa, R., Tahara, M., Tasaka, K., and Murata, Y. (2004) Induction of glucose transporter 1 expression through hypoxia-inducible factor 1alpha under hypoxic conditions in trophoblast-derived cells. J Endocrinol **183**, 145-154.

Heerlein, K., Schulze, A., Hotz, L., Bartsch, P., and Mairbaurl, H. (2005) Hypoxia decreases cellular ATP demand and inhibits mitochondrial respiration of a549 cells. Am J Respir Cell Mol Biol **32**, 44-51.

Henderson, D.J., and Copp, A.J. (1998) Versican expression is associated with chamber specification, septation, and valvulogenesis in the developing mouse heart. Circ Res **83**, 523-532.

Hernandez, D., Miquel-Serra, L., Docampo, M.J., Marco-Ramell, A., Cabrera, J., Fabra, A., and Bassols, A. (2010) V3 versican isoform alters the behavior of human melanoma cells by interfering with CD44/ErbB-dependent signaling. J Biol Chem **286**, 1475-1485.

Hewitson, K.S., McNeill, L.A., Riordan, M.V., Tian, Y.M., Bullock, A.N., Welford, R.W., Elkins, J.M., Oldham, N.J., Bhattacharya, S., Gleadle, J.M., Ratcliffe, P.J., Pugh, C.W., and Schofield, C.J. (2002) Hypoxia-inducible factor (HIF) asparagine hydroxylase is identical to factor inhibiting HIF (FIH) and is related to the cupin structural family. J Biol Chem **277**, 26351-26355.

Hinek, A., Braun, K.R., Liu, K., Wang, Y., and Wight, T.N. (2004) Retrovirally mediated overexpression of versican v3 reverses impaired elastogenesis and heightened proliferation exhibited by fibroblasts from Costello syndrome and Hurler disease patients. Am J Pathol **164**, 119-131.

Hirose, J., Kawashima, H., Yoshie, O., Tashiro, K., and Miyasaka, M. (2001) Versican interacts with chemokines and modulates cellular responses. J Biol Chem **276**, 5228-5234.

Hirota, K. (2002) Hypoxia-inducible factor 1, a master transcription factor of cellular hypoxic gene expression. J Anesth **16**, 150-159.

Hockel, M., Schlenger, K., Aral, B., Mitze, M., Schaffer, U., and Vaupel, P. (1996) Association between tumor hypoxia and malignant progression in advanced cancer of the uterine cervix. Cancer Res **56**, 4509-4515.

Hockel, M., and Vaupel, P. (2001) Tumor hypoxia: definitions and current clinical, biologic, and molecular aspects. J Natl Cancer Inst **93**, 266-276.

Hollander, A.P., Corke, K.P., Freemont, A.J., and Lewis, C.E. (2001) Expression of hypoxia-inducible factor 1alpha by macrophages in the rheumatoid synovium: implications for targeting of therapeutic genes to the inflamed joint. Arthritis Rheum **44**, 1540-1544.

Hoshino, K., Takeuchi, O., Kawai, T., Sanjo, H., Ogawa, T., Takeda, Y., Takeda, K., and Akira, S. (1999) Cutting edge: Toll-like receptor 4 (TLR4)-deficient mice are hyporesponsive to lipopolysaccharide: evidence for TLR4 as the Lps gene product. J Immunol **162**, 3749-3752.

Hunziker, W., Whitney, J.A., and Mellman, I. (1992) Brefeldin A and the endocytic pathway. Possible implications for membrane traffic and sorting. FEBS Lett **307**, 93-96.

Husain, T., Abbott, C.R., Scott, D.J., and Gough, M.J. (1999) Macrophage accumulation within the cap of carotid atherosclerotic plaques is associated with the onset of cerebral ischemic events. J Vasc Surg **30**, 269-276.

Ikeda, E., Achen, M.G., Breier, G., and Risau, W. (1995) Hypoxia-induced transcriptional activation and increased mRNA stability of vascular endothelial growth factor in C6 glioma cells. J Biol Chem **270**, 19761-19766.

Iozzo, R.V. (1998) Matrix proteoglycans: from molecular design to cellular function. Annu Rev Biochem **67**, 609-652.

Iozzo, R.V., and Murdoch, A.D. (1996) Proteoglycans of the extracellular environment: clues from the gene and protein side offer novel perspectives in molecular diversity and function. FASEB J **10**, 598-614.

lozzo, R.V., Naso, M.F., Cannizzaro, L.A., Wasmuth, J.J., and McPherson, J.D. (1992) Mapping of the versican proteoglycan gene (CSPG2) to the long arm of human chromosome 5 (5q12-5q14). Genomics **14**, 845-851.

Issekutz, T.B., Issekutz, A.C., and Movat, H.Z. (1981) The in vivo quantitation and kinetics of monocyte migration into acute inflammatory tissue. Am J Pathol **103**, 47-55.

Ivan, M., Kondo, K., Yang, H., Kim, W., Valiando, J., Ohh, M., Salic, A., Asara, J.M., Lane, W.S., and Kaelin, W.G., Jr. (2001) HIFalpha targeted for VHL-mediated destruction by proline hydroxylation: implications for O2 sensing. Science **292**, 464-468.

Jaakkola, P., Mole, D.R., Tian, Y.M., Wilson, M.I., Gielbert, J., Gaskell, S.J., Kriegsheim, A., Hebestreit, H.F., Mukherji, M., Schofield, C.J., Maxwell, P.H., Pugh, C.W., and Ratcliffe, P.J. (2001) Targeting of HIF-alpha to the von Hippel-Lindau ubiquitylation complex by O2-regulated prolyl hydroxylation. Science **292**, 468-472.

Jacob, M.C., Favre, M., and Bensa, J.C. (1991) Membrane cell permeabilization with saponin and multiparametric analysis by flow cytometry. Cytometry **12**, 550-558.

Janeway, C.A., Jr., and Medzhitov, R. (2002) Innate immune recognition. Annu Rev Immunol **20**, 197-216.

Jewell, U.R., Kvietikova, I., Scheid, A., Bauer, C., Wenger, R.H., and Gassmann, M. (2001) Induction of HIF-1alpha in response to hypoxia is instantaneous. FASEB J **15**, 1312-1314.

Jiang, B.H., Jiang, G., Zheng, J.Z., Lu, Z., Hunter, T., and Vogt, P.K. (2001) Phosphatidylinositol 3-kinase signaling controls levels of hypoxia-inducible factor 1. Cell Growth Differ **12**, 363-369.

Jiang, B.H., Zheng, J.Z., Leung, S.W., Roe, R., and Semenza, G.L. (1997) Transactivation and inhibitory domains of hypoxia-inducible factor 1alpha. Modulation of transcriptional activity by oxygen tension. J Biol Chem **272**, 19253-19260.

Jogi, A., Ora, I., Nilsson, H., Lindeheim, A., Makino, Y., Poellinger, L., Axelson, H., and Pahlman, S. (2002) Hypoxia alters gene expression in human neuroblastoma cells toward an immature and neural crest-like phenotype. Proc Natl Acad Sci U S A **99**, 7021-7026.

Johnson, P.R. (2001) Role of human airway smooth muscle in altered extracellular matrix production in asthma. Clin Exp Pharmacol Physiol **28**, 233-236.

Jones, L.L., Sajed, D., and Tuszynski, M.H. (2003) Axonal regeneration through regions of chondroitin sulfate proteoglycan deposition after spinal cord injury: a balance of permissiveness and inhibition. J Neurosci **23**, 9276-9288.

Jurgensen, J.S., Rosenberger, C., Wiesener, M.S., Warnecke, C., Horstrup, J.H., Grafe, M., Philipp, S., Griethe, W., Maxwell, P.H., Frei, U., Bachmann, S., Willenbrock, R., and Eckardt, K.U. (2004) Persistent induction of HIF-1alpha and -2alpha in cardiomyocytes and stromal cells of ischemic myocardium. FASEB J **18**, 1415-1417.

Jutras, I., and Desjardins, M. (2005) Phagocytosis: at the crossroads of innate and adaptive immunity. Annu Rev Cell Dev Biol **21**, 511-527.

Kallio, P.J., Okamoto, K., O'Brien, S., Carrero, P., Makino, Y., Tanaka, H., and Poellinger, L. (1998) Signal transduction in hypoxic cells: inducible nuclear translocation and recruitment of the CBP/p300 coactivator by the hypoxia-inducible factor-1alpha. EMBO J **17**, 6573-6586.

Kamura, T., Sato, S., Iwai, K., Czyzyk-Krzeska, M., Conaway, R.C., and Conaway, J.W. (2000) Activation of HIF1alpha ubiquitination by a reconstituted von Hippel-Lindau (VHL) tumor suppressor complex. Proc Natl Acad Sci U S A **97**, 10430-10435.

Karin, M. (2006) Nuclear factor-kappaB in cancer development and progression. Nature **441**, 431-436.

Karin, M., and Greten, F.R. (2005) NF-kappaB: linking inflammation and immunity to cancer development and progression. Nat Rev Immunol **5**, 749-759.

Katso, R., Okkenhaug, K., Ahmadi, K., White, S., Timms, J., and Waterfield, M.D. (2001) Cellular function of phosphoinositide 3-kinases: implications for development, homeostasis, and cancer. Annu Rev Cell Dev Biol **17**, 615-675.

Kawashima, H., Atarashi, K., Hirose, M., Hirose, J., Yamada, S., Sugahara, K., and Miyasaka, M. (2002) Oversulfated chondroitin/dermatan sulfates containing GlcAbeta1/IdoAalpha1-3GalNAc(4,6-O-disulfate) interact with L- and P-selectin and chemokines. J Biol Chem **277**, 12921-12930.

Kenagy, R.D., Plaas, A.H., and Wight, T.N. (2006) Versican degradation and vascular disease. Trends Cardiovasc Med **16**, 209-215.

Kern, C.B., Twal, W.O., Mjaatvedt, C.H., Fairey, S.E., Toole, B.P., Iruela-Arispe, M.L., and Argraves, W.S. (2006) Proteolytic cleavage of versican during cardiac cushion morphogenesis. Dev Dyn **235**, 2238-2247.

Khalil, M.F., Wagner, W.D., and Goldberg, I.J. (2004) Molecular interactions leading to lipoprotein retention and the initiation of atherosclerosis. Arterioscler Thromb Vasc Biol **24**, 2211-2218.

Kietzmann, T., Roth, U., and Jungermann, K. (1999) Induction of the plasminogen activator inhibitor-1 gene expression by mild hypoxia via a hypoxia response element binding the hypoxia-inducible factor-1 in rat hepatocytes. Blood **94**, 4177-4185.

Kim, S., Takahashi, H., Lin, W.W., Descargues, P., Grivennikov, S., Kim, Y., Luo, J.L., and Karin, M. (2009) Carcinoma-produced factors activate myeloid cells through TLR2 to stimulate metastasis. Nature **457**, 102-106.

Kinne, R.W., Brauer, R., Stuhlmuller, B., Palombo-Kinne, E., and Burmester, G.R. (2000) Macrophages in rheumatoid arthritis. Arthritis Res **2**, 189-202.

Kischel, P., Waltregny, D., Dumont, B., Turtoi, A., Greffe, Y., Kirsch, S., De Pauw, E., and Castronovo, V. (2010) Versican overexpression in human breast cancer lesions: known and new isoforms for stromal tumor targeting. Int J Cancer **126**, 640-650.

Knowles, H.J., Mole, D.R., Ratcliffe, P.J., and Harris, A.L. (2006) Normoxic stabilization of hypoxia-inducible factor-1alpha by modulation of the labile iron pool in differentiating U937 macrophages: effect of natural resistance-associated macrophage protein 1. Cancer Res **66**, 2600-2607.

Kodama, J., Hasengaowa, Kusumoto, T., Seki, N., Matsuo, T., Nakamura, K., Hongo, A., and Hiramatsu, Y. (2007) Versican expression in human cervical cancer. Eur J Cancer **43**, 1460-1466.

Kresse, H., and Schonherr, E. (2001) Proteoglycans of the extracellular matrix and growth control. J Cell Physiol **189**, 266-274.

Krieger, M. (2001) Scavenger receptor class B type I is a multiligand HDL receptor that influences diverse physiologic systems. J Clin Invest **108**, 793-797.

Krutzik, P.O., and Nolan, G.P. (2003) Intracellular phospho-protein staining techniques for flow cytometry: monitoring single cell signaling events. Cytometry A 55, 61-70.

Kuziel, W.A., Morgan, S.J., Dawson, T.C., Griffin, S., Smithies, O., Ley, K., and Maeda, N. (1997) Severe reduction in leukocyte adhesion and monocyte extravasation in mice deficient in CC chemokine receptor 2. Proc Natl Acad Sci U S A **94**, 12053-12058.

Kvietikova, I., Wenger, R.H., Marti, H.H., and Gassmann, M. (1995) The transcription factors ATF-1 and CREB-1 bind constitutively to the hypoxia-inducible factor-1 (HIF-1) DNA recognition site. Nucleic Acids Res **23**, 4542-4550.

Kwiatkowska, K., and Sobota, A. (1999) Signaling pathways in phagocytosis. Bioessays **21**, 422-431.

Lando, D., Peet, D.J., Whelan, D.A., Gorman, J.J., and Whitelaw, M.L. (2002) Asparagine hydroxylation of the HIF transactivation domain a hypoxic switch. Science **295**, 858-861.

Landolt, R.M., Vaughan, L., Winterhalter, K.H., and Zimmermann, D.R. (1995) Versican is selectively expressed in embryonic tissues that act as barriers to neural crest cell migration and axon outgrowth. Development **121**, 2303-2312.

Laskin, D.L. (2009) Macrophages and inflammatory mediators in chemical toxicity: a battle of forces. Chem Res Toxicol **22**, 1376-1385.

Laughner, E., Taghavi, P., Chiles, K., Mahon, P.C., and Semenza, G.L. (2001) HER2 (neu) signaling increases the rate of hypoxia-inducible factor 1alpha (HIF-1alpha) synthesis: novel mechanism for HIF-1-mediated vascular endothelial growth factor expression. Mol Cell Biol **21**, 3995-4004.

Lay, J.C., Alexis, N.E., Zeman, K.L., Peden, D.B., and Bennett, W.D. (2009) In vivo uptake of inhaled particles by airway phagocytes is enhanced in patients with mild asthma compared with normal volunteers. Thorax **64**, 313-320.

Leandersson, K., Riesbeck, K., and Andersson, T. (2006) Wnt-5a mRNA translation is suppressed by the Elav-like protein HuR in human breast epithelial cells. Nucleic Acids Res **34**, 3988-3999.

LeBaron, R.G., Zimmermann, D.R., and Ruoslahti, E. (1992) Hyaluronate binding properties of versican. J Biol Chem **267**, 10003-10010.

Leder, L.D. (1967) The origin of blood monocytes and macrophages. A review. Blut **16**, 86-98.

Lee, E.Y., Xia, Y., Kim, W.S., Kim, M.H., Kim, T.H., Kim, K.J., Park, B.S., and Sung, J.H. (2009) Hypoxia-enhanced wound-healing function of adipose-derived stem cells: increase in stem cell proliferation and up-regulation of VEGF and bFGF. Wound Repair Regen 17, 540-547.

Lee, J.Y., and Spicer, A.P. (2000) Hyaluronan: a multifunctional, megaDalton, stealth molecule. Curr Opin Cell Biol **12**, 581-586.

Lee, M., Bikram, M., Oh, S., Bull, D.A., and Kim, S.W. (2004) Sp1-dependent regulation of the RTP801 promoter and its application to hypoxia-inducible VEGF plasmid for ischemic disease. Pharm Res 21, 736-741.

Lee, S., Jo, M., Lee, J., Koh, S.S., and Kim, S. (2007) Identification of novel universal housekeeping genes by statistical analysis of microarray data. J Biochem Mol Biol **40**, 226-231.

Lee, S.M., Lee, C.T., Kim, Y.W., Han, S.K., Shim, Y.S., and Yoo, C.G. (2006) Hypoxia confers protection against apoptosis via PI3K/Akt and ERK pathways in lung cancer cells. Cancer Lett **242**, 231-238.

Leek, R.D., Landers, R.J., Harris, A.L., and Lewis, C.E. (1999) Necrosis correlates with high vascular density and focal macrophage infiltration in invasive carcinoma of the breast. Br J Cancer **79**, 991-995.

Leek, R.D., Talks, K.L., Pezzella, F., Turley, H., Campo, L., Brown, N.S., Bicknell, R., Taylor, M., Gatter, K.C., and Harris, A.L. (2002) Relation of hypoxia-inducible factor-2 alpha (HIF-2 alpha) expression in tumor-infiltrative macrophages to tumor angiogenesis and the oxidative thymidine phosphorylase pathway in Human breast cancer. Cancer Res **62**, 1326-1329.

Lemire, J.M., Braun, K.R., Maurel, P., Kaplan, E.D., Schwartz, S.M., and Wight, T.N. (1999) Versican/PG-M isoforms in vascular smooth muscle cells. Arterioscler Thromb Vasc Biol **19**, 1630-1639.

Lemire, J.M., Merrilees, M.J., Braun, K.R., and Wight, T.N. (2002) Overexpression of the V3 variant of versican alters arterial smooth muscle cell adhesion, migration, and proliferation in vitro. J Cell Physiol **190**, 38-45.

Leonard, M.O., Howell, K., Madden, S.F., Costello, C.M., Higgins, D.G., Taylor, C.T., and McLoughlin, P. (2008) Hypoxia selectively activates the CREB family of transcription factors in the in vivo lung. Am J Respir Crit Care Med **178**, 977-983.

Levy, N.S., Chung, S., Furneaux, H., and Levy, A.P. (1998) Hypoxic stabilization of vascular endothelial growth factor mRNA by the RNA-binding protein HuR. J Biol Chem **273**, 6417-6423.

Levy, N.S., Goldberg, M.A., and Levy, A.P. (1997) Sequencing of the human vascular endothelial growth factor (VEGF) 3' untranslated region (UTR): conservation of five hypoxia-inducible RNA-protein binding sites. Biochim Biophys Acta **1352**, 167-173.

Lewis, C., and Murdoch, C. (2005) Macrophage responses to hypoxia: implications for tumor progression and anti-cancer therapies. Am J Pathol **167**, 627-635.

Lewis, C.E., and Pollard, J.W. (2006) Distinct role of macrophages in different tumor microenvironments. Cancer Res **66**, 605-612.

Lewis, J.S., Landers, R.J., Underwood, J.C., Harris, A.L., and Lewis, C.E. (2000) Expression of vascular endothelial growth factor by macrophages is up-regulated in poorly vascularized areas of breast carcinomas. J Pathol **192**, 150-158.

Lewis, J.S., Lee, J.A., Underwood, J.C., Harris, A.L., and Lewis, C.E. (1999) Macrophage responses to hypoxia: relevance to disease mechanisms. J Leukoc Biol **66**, 889-900.

Lin, S.K., Shun, C.T., Kok, S.H., Wang, C.C., Hsiao, T.Y., and Liu, C.M. (2008) Hypoxiastimulated vascular endothelial growth factor production in human nasal polyp fibroblasts: effect of epigallocatechin-3-gallate on hypoxia-inducible factor-1 alpha synthesis. Arch Otolaryngol Head Neck Surg **134**, 522-527.

Liu, L., Cash, T.P., Jones, R.G., Keith, B., Thompson, C.B., and Simon, M.C. (2006) Hypoxia-induced energy stress regulates mRNA translation and cell growth. Mol Cell **21**, 521-531. Liu, L., Zhu, X.D., Wang, W.Q., Shen, Y., Qin, Y., Ren, Z.G., Sun, H.C., and Tang, Z.Y. (2010) Activation of beta-catenin by hypoxia in hepatocellular carcinoma contributes to enhanced metastatic potential and poor prognosis. Clin Cancer Res **16**, 2740-2750.

Liu, P., Cheng, H., Roberts, T.M., and Zhao, J.J. (2009) Targeting the phosphoinositide 3-kinase pathway in cancer. Nat Rev Drug Discov **8**, 627-644.

Liu, X.H., Kirschenbaum, A., Yao, S., Stearns, M.E., Holland, J.F., Claffey, K., and Levine, A.C. (1999) Upregulation of vascular endothelial growth factor by cobalt chloridesimulated hypoxia is mediated by persistent induction of cyclooxygenase-2 in a metastatic human prostate cancer cell line. Clin Exp Metastasis **17**, 687-694.

Liu, Y., Cox, S.R., Morita, T., and Kourembanas, S. (1995) Hypoxia regulates vascular endothelial growth factor gene expression in endothelial cells. Identification of a 5' enhancer. Circ Res **77**, 638-643.

Llorca, O. (2008) Extended and bent conformations of the mannose receptor family. Cell Mol Life Sci **65**, 1302-1310.

Lu, B., Rutledge, B.J., Gu, L., Fiorillo, J., Lukacs, N.W., Kunkel, S.L., North, R., Gerard, C., and Rollins, B.J. (1998) Abnormalities in monocyte recruitment and cytokine expression in monocyte chemoattractant protein 1-deficient mice. J Exp Med **187**, 601-608.

Ma, J., Chen, T., Mandelin, J., Ceponis, A., Miller, N.E., Hukkanen, M., Ma, G.F., and Konttinen, Y.T. (2003) Regulation of macrophage activation. Cell Mol Life Sci **60**, 2334-2346.

Mace, K.A., Yu, D.H., Paydar, K.Z., Boudreau, N., and Young, D.M. (2007) Sustained expression of Hif-1alpha in the diabetic environment promotes angiogenesis and cutaneous wound repair. Wound Repair Regen 15, 636-645.

Macheda, M.L., Rogers, S., and Best, J.D. (2005) Molecular and cellular regulation of glucose transporter (GLUT) proteins in cancer. J Cell Physiol **202**, 654-662.

Mahon, P.C., Hirota, K., and Semenza, G.L. (2001) FIH-1: a novel protein that interacts with HIF-1alpha and VHL to mediate repression of HIF-1 transcriptional activity. Genes Dev **15**, 2675-2686.

Maier, T., Guell, M., and Serrano, L. (2009) Correlation of mRNA and protein in complex biological samples. FEBS Lett **583**, 3966-3973.

Mantovani, A., Schioppa, T., Porta, C., Allavena, P., and Sica, A. (2006) Role of tumorassociated macrophages in tumor progression and invasion. Cancer Metastasis Rev **25**, 315-322. Mantovani, A., Sica, A., Sozzani, S., Allavena, P., Vecchi, A., and Locati, M. (2004) The chemokine system in diverse forms of macrophage activation and polarization. Trends Immunol **25**, 677-686.

Mapp, P.I., Grootveld, M.C., and Blake, D.R. (1995) Hypoxia, oxidative stress and rheumatoid arthritis. Br Med Bull **51**, 419-436.

Maxwell, P.H., Wiesener, M.S., Chang, G.W., Clifford, S.C., Vaux, E.C., Cockman, M.E., Wykoff, C.C., Pugh, C.W., Maher, E.R., and Ratcliffe, P.J. (1999) The tumour suppressor protein VHL targets hypoxia-inducible factors for oxygen-dependent proteolysis. Nature **399**, 271-275.

Mayanil, C.S., George, D., Freilich, L., Miljan, E.J., Mania-Farnell, B., McLone, D.G., and Bremer, E.G. (2001) Microarray analysis detects novel Pax3 downstream target genes. J Biol Chem **276**, 49299-49309.

Mazumdar, J., O'Brien, W.T., Johnson, R.S., LaManna, J.C., Chavez, J.C., Klein, P.S., and Simon, M.C. (2010) O2 regulates stem cells through Wnt/beta-catenin signalling. Nat Cell Biol **12**, 1007-1013.

Mazure, N.M., Chen, E.Y., Laderoute, K.R., and Giaccia, A.J. (1997) Induction of vascular endothelial growth factor by hypoxia is modulated by a phosphatidylinositol 3-kinase/Akt signaling pathway in Ha-ras-transformed cells through a hypoxia inducible factor-1 transcriptional element. Blood **90**, 3322-3331.

McNeill, L.A., Hewitson, K.S., Claridge, T.D., Seibel, J.F., Horsfall, L.E., and Schofield, C.J. (2002) Hypoxia-inducible factor asparaginyl hydroxylase (FIH-1) catalyses hydroxylation at the beta-carbon of asparagine-803. Biochem J **367**, 571-575.

Merrilees, M.J., Hankin, E.J., Black, J.L., and Beaumont, B. (2004) Matrix proteoglycans and remodelling of interstitial lung tissue in lymphangioleiomyomatosis. J Pathol **203**, 653-660.

Minchenko, A., Leshchinsky, I., Opentanova, I., Sang, N., Srinivas, V., Armstead, V., and Caro, J. (2002) Hypoxia-inducible factor-1-mediated expression of the 6-phosphofructo-2-kinase/fructose-2,6-bisphosphatase-3 (PFKFB3) gene. Its possible role in the Warburg effect. J Biol Chem **277**, 6183-6187.

Minchenko, O., Opentanova, I., Minchenko, D., Ogura, T., and Esumi, H. (2004) Hypoxia induces transcription of 6-phosphofructo-2-kinase/fructose-2,6-biphosphatase-4 gene via hypoxia-inducible factor-1alpha activation. FEBS Lett **576**, 14-20.

Miyake, K. (2004) Innate recognition of lipopolysaccharide by Toll-like receptor 4-MD-2. Trends Microbiol **12**, 186-192.

Mojsilovic-Petrovic, J., Callaghan, D., Cui, H., Dean, C., Stanimirovic, D.B., and Zhang, W. (2007) Hypoxia-inducible factor-1 (HIF-1) is involved in the regulation of hypoxiastimulated expression of monocyte chemoattractant protein-1 (MCP-1/CCL2) and MCP-5 (Ccl12) in astrocytes. J Neuroinflammation **4**, 12.

Mole, D.R., Maxwell, P.H., Pugh, C.W., and Ratcliffe, P.J. (2001) Regulation of HIF by the von Hippel-Lindau tumour suppressor: implications for cellular oxygen sensing. IUBMB Life **52**, 43-47.

Moynagh, P.N. (2003) Toll-like receptor signalling pathways as key targets for mediating the anti-inflammatory and immunosuppressive effects of glucocorticoids. J Endocrinol **179**, 139-144.

Mukaratirwa, S., van Ederen, A.M., Gruys, E., and Nederbragt, H. (2004) Versican and hyaluronan expression in canine colonic adenomas and carcinomas: relation to malignancy and depth of tumour invasion. J Comp Pathol **131**, 259-270.

Murdoch, C., Giannoudis, A., and Lewis, C.E. (2004) Mechanisms regulating the recruitment of macrophages into hypoxic areas of tumors and other ischemic tissues. Blood **104**, 2224-2234.

Murdoch, C., Muthana, M., and Lewis, C.E. (2005) Hypoxia regulates macrophage functions in inflammation. J Immunol **175**, 6257-6263.

Mutsaers, S.E., Bishop, J.E., McGrouther, G., and Laurent, G.J. (1997) Mechanisms of tissue repair: from wound healing to fibrosis. Int J Biochem Cell Biol **29**, 5-17.

Naso, M.F., Zimmermann, D.R., and Iozzo, R.V. (1994) Characterization of the complete genomic structure of the human versican gene and functional analysis of its promoter. J Biol Chem **269**, 32999-33008.

Nathan, C.F. (1987) Secretory products of macrophages. J Clin Invest 79, 319-326.

Nebenfuhr, A., Ritzenthaler, C., and Robinson, D.G. (2002) Brefeldin A: deciphering an enigmatic inhibitor of secretion. Plant Physiol **130**, 1102-1108.

Negus, R.P., Stamp, G.W., Hadley, J., and Balkwill, F.R. (1997) Quantitative assessment of the leukocyte infiltrate in ovarian cancer and its relationship to the expression of C-C chemokines. Am J Pathol **150**, 1723-1734.

O'Connor, D.J., and Lu, X. (2000) Stress signals induce transcriptionally inactive E2F-1 independently of p53 and Rb. Oncogene **19**, 2369-2376.

Oda, T., Hirota, K., Nishi, K., Takabuchi, S., Oda, S., Yamada, H., Arai, T., Fukuda, K., Kita, T., Adachi, T., Semenza, G.L., and Nohara, R. (2006) Activation of hypoxiainducible factor 1 during macrophage differentiation. Am J Physiol Cell Physiol **291**, C104-113. Oh, Y.T., Lee, J.Y., Yoon, H., Lee, E.H., Baik, H.H., Kim, S.S., Ha, J., Yoon, K.S., Choe, W., and Kang, I. (2008) Lipopolysaccharide induces hypoxia-inducible factor-1 alpha mRNA expression and activation via NADPH oxidase and Sp1-dependent pathway in BV2 murine microglial cells. Neurosci Lett **431**, 155-160.

Olin, K.L., Potter-Perigo, S., Barrett, P.H., Wight, T.N., and Chait, A. (1999) Lipoprotein lipase enhances the binding of native and oxidized low density lipoproteins to versican and biglycan synthesized by cultured arterial smooth muscle cells. J Biol Chem **274**, 34629-34636.

Ota, H., Katsube, K., Ogawa, J., and Yanagishita, M. (2007) Hypoxia/Notch signaling in primary culture of rat lymphatic endothelial cells. FEBS Lett **581**, 5220-5226.

Overgaard, J. (2007) Hypoxic radiosensitization: adored and ignored. J Clin Oncol 25, 4066-4074.

Page, E.L., Robitaille, G.A., Pouyssegur, J., and Richard, D.E. (2002) Induction of hypoxia-inducible factor-1alpha by transcriptional and translational mechanisms. J Biol Chem **277**, 48403-48409.

Pascal, L.E., True, L.D., Campbell, D.S., Deutsch, E.W., Risk, M., Coleman, I.M., Eichner, L.J., Nelson, P.S., and Liu, A.Y. (2008) Correlation of mRNA and protein levels: cell type-specific gene expression of cluster designation antigens in the prostate. BMC Genomics **9**, 246.

Patel, K., Bhaskaran, M., Dani, D., Reddy, K., and Singhal, P.C. (2003) Role of heme oxygenase-1 in morphine-modulated apoptosis and migration of macrophages. J Infect Dis **187**, 47-54.

Paulus, W., Baur, I., Dours-Zimmermann, M.T., and Zimmermann, D.R. (1996) Differential expression of versican isoforms in brain tumors. J Neuropathol Exp Neurol **55**, 528-533.

Pearson, A.M. (1996) Scavenger receptors in innate immunity. Curr Opin Immunol 8, 20-28.

Peiser, L., Mukhopadhyay, S., and Gordon, S. (2002) Scavenger receptors in innate immunity. Curr Opin Immunol **14**, 123-128.

Pennacchietti, S., Michieli, P., Galluzzo, M., Mazzone, M., Giordano, S., and Comoglio, P.M. (2003) Hypoxia promotes invasive growth by transcriptional activation of the met protooncogene. Cancer Cell **3**, 347-361.

Perides, G., Rahemtulla, F., Lane, W.S., Asher, R.A., and Bignami, A. (1992) Isolation of a large aggregating proteoglycan from human brain. J Biol Chem **267**, 23883-23887.

Perissinotto, D., Iacopetti, P., Bellina, I., Doliana, R., Colombatti, A., Pettway, Z., Bronner-Fraser, M., Shinomura, T., Kimata, K., Morgelin, M., Lofberg, J., and Perris, R. (2000) Avian neural crest cell migration is diversely regulated by the two major hyaluronan-binding proteoglycans PG-M/versican and aggrecan. Development **127**, 2823-2842.

Perris, R., Perissinotto, D., Pettway, Z., Bronner-Fraser, M., Morgelin, M., and Kimata, K. (1996) Inhibitory effects of PG-H/aggrecan and PG-M/versican on avian neural crest cell migration. FASEB J **10**, 293-301.

Pham, I., Uchida, T., Planes, C., Ware, L.B., Kaner, R., Matthay, M.A., and Clerici, C. (2002) Hypoxia upregulates VEGF expression in alveolar epithelial cells in vitro and in vivo. Am J Physiol Lung Cell Mol Physiol **283**, L1133-1142.

Piga, A., Gaglioti, C., Fogliacco, E., and Tricta, F. (2003) Comparative effects of deferiprone and deferoxamine on survival and cardiac disease in patients with thalassemia major: a retrospective analysis. Haematologica **88**, 489-496.

Pocock, J., Gomez-Guerrero, C., Harendza, S., Ayoub, M., Hernandez-Vargas, P., Zahner, G., Stahl, R.A., and Thaiss, F. (2003) Differential activation of NF-kappa B, AP-1, and C/EBP in endotoxin-tolerant rats: mechanisms for in vivo regulation of glomerular RANTES/CCL5 expression. J Immunol **170**, 6280-6291.

Polakis, P. (2000) Wnt signaling and cancer. Genes Dev 14, 1837-1851.

Pouyssegur, J., Dayan, F., and Mazure, N.M. (2006) Hypoxia signalling in cancer and approaches to enforce tumour regression. Nature **441**, 437-443.

Pugh, C.W., O'Rourke, J.F., Nagao, M., Gleadle, J.M., and Ratcliffe, P.J. (1997) Activation of hypoxia-inducible factor-1; definition of regulatory domains within the alpha subunit. J Biol Chem **272**, 11205-11214.

Pugh, C.W., and Ratcliffe, P.J. (2003) The von Hippel-Lindau tumor suppressor, hypoxia-inducible factor-1 (HIF-1) degradation, and cancer pathogenesis. Semin Cancer Biol **13**, 83-89.

Rabinovitch, M. (1995) Professional and non-professional phagocytes: an introduction. Trends Cell Biol **5**, 85-87.

Rahmani, M., Read, J.T., Carthy, J.M., McDonald, P.C., Wong, B.W., Esfandiarei, M., Si, X., Luo, Z., Luo, H., Rennie, P.S., and McManus, B.M. (2005) Regulation of the versican promoter by the beta-catenin-T-cell factor complex in vascular smooth muscle cells. J Biol Chem **280**, 13019-13028.

Rahmani, M., Wong, B.W., Ang, L., Cheung, C.C., Carthy, J.M., Walinski, H., and McManus, B.M. (2006) Versican: signaling to transcriptional control pathways. Can J Physiol Pharmacol **84**, 77-92.

Ramanathan, M., Pinhal-Enfield, G., Hao, I., and Leibovich, S.J. (2007) Synergistic upregulation of vascular endothelial growth factor (VEGF) expression in macrophages by adenosine A2A receptor agonists and endotoxin involves transcriptional regulation via the hypoxia response element in the VEGF promoter. Mol Biol Cell **18**, 14-23.

Rameh, L.E., and Cantley, L.C. (1999) The role of phosphoinositide 3-kinase lipid products in cell function. J Biol Chem **274**, 8347-8350.

Ratcliffe, P.J., O'Rourke, J.F., Maxwell, P.H., and Pugh, C.W. (1998) Oxygen sensing, hypoxia-inducible factor-1 and the regulation of mammalian gene expression. J Exp Biol **201**, 1153-1162.

Ricardo, S.D., van Goor, H., and Eddy, A.A. (2008) Macrophage diversity in renal injury and repair. J Clin Invest **118**, 3522-3530.

Ricciardelli, C., Mayne, K., Sykes, P.J., Raymond, W.A., McCaul, K., Marshall, V.R., and Horsfall, D.J. (1998) Elevated levels of versican but not decorin predict disease progression in early-stage prostate cancer. Clin Cancer Res **4**, 963-971.

Ricciardelli, C., Sakko, A.J., Ween, M.P., Russell, D.L., and Horsfall, D.J. (2009) The biological role and regulation of versican levels in cancer. Cancer Metastasis Rev 28, 233-245.

Richard, D.E., Berra, E., and Pouyssegur, J. (2000) Nonhypoxic pathway mediates the induction of hypoxia-inducible factor 1alpha in vascular smooth muscle cells. J Biol Chem **275**, 26765-26771.

Rupec, R.A., and Baeuerle, P.A. (1995) The genomic response of tumor cells to hypoxia and reoxygenation. Differential activation of transcription factors AP-1 and NF-kappa B. Eur J Biochem **234**, 632-640.

Russell, D.L., Ochsner, S.A., Hsieh, M., Mulders, S., and Richards, J.S. (2003) Hormoneregulated expression and localization of versican in the rodent ovary. Endocrinology **144**, 1020-1031.

Ryan, H.E., Lo, J., and Johnson, R.S. (1998) HIF-1 alpha is required for solid tumor formation and embryonic vascularization. EMBO J **17**, 3005-3015.

Sakko, A.J., Ricciardelli, C., Mayne, K., Suwiwat, S., LeBaron, R.G., Marshall, V.R., Tilley, W.D., and Horsfall, D.J. (2003) Modulation of prostate cancer cell attachment to matrix by versican. Cancer Res **63**, 4786-4791.

Salnikow, K., Kluz, T., Costa, M., Piquemal, D., Demidenko, Z.N., Xie, K., and Blagosklonny, M.V. (2002) The regulation of hypoxic genes by calcium involves c-Jun/AP-1, which cooperates with hypoxia-inducible factor 1 in response to hypoxia. Mol Cell Biol **22**, 1734-1741.

Sanchez-Elsner, T., Ramirez, J.R., Sanz-Rodriguez, F., Varela, E., Bernabeu, C., and Botella, L.M. (2004) A cross-talk between hypoxia and TGF-beta orchestrates erythropoietin gene regulation through SP1 and Smads. J Mol Biol **336**, 9-24.

Scannell, G., Waxman, K., Kaml, G.J., Ioli, G., Gatanaga, T., Yamamoto, R., and Granger, G.A. (1993) Hypoxia induces a human macrophage cell line to release tumor necrosis factor-alpha and its soluble receptors in vitro. J Surg Res **54**, 281-285.

Schmid-Schonbein, G.W. (2006) Analysis of inflammation. Annu Rev Biomed Eng 8, 93-131.

Schonherr, E., Jarvelainen, H.T., Sandell, L.J., and Wight, T.N. (1991) Effects of plateletderived growth factor and transforming growth factor-beta 1 on the synthesis of a large versican-like chondroitin sulfate proteoglycan by arterial smooth muscle cells. J Biol Chem **266**, 17640-17647.

Schonherr, E., Kinsella, M.G., and Wight, T.N. (1997) Genistein selectively inhibits platelet-derived growth factor-stimulated versican biosynthesis in monkey arterial smooth muscle cells. Arch Biochem Biophys **339**, 353-361.

Schwartz, N.B., and Domowicz, M. (2004) Proteoglycans in brain development. Glycoconj J **21**, 329-341.

Schreiber, E., Matthias, P., Muller, M.M., and Schaffner, W. (1989) Rapid detection of octamer binding proteins with 'mini-extracts', prepared from a small number of cells. Nucleic Acids Res **17**, 6419.

Sciaky, N., Presley, J., Smith, C., Zaal, K.J., Cole, N., Moreira, J.E., Terasaki, M., Siggia, E., and Lippincott-Schwartz, J. (1997) Golgi tubule traffic and the effects of brefeldin A visualized in living cells. J Cell Biol **139**, 1137-1155.

Seidelmann, S.B., Kuo, C., Pleskac, N., Molina, J., Sayers, S., Li, R., Zhou, J., Johnson, P., Braun, K., Chan, C., Teupser, D., Breslow, J.L., Wight, T.N., Tall, A.R., and Welch, C.L. (2008) Athsq1 is an atherosclerosis modifier locus with dramatic effects on lesion area and prominent accumulation of versican. Arterioscler Thromb Vasc Biol **28**, 2180-2186.

Semenza, G.L. (2010) Oxygen homeostasis. Wiley Interdiscip Rev Syst Biol Med 2, 336-361.

Semenza, G.L. (1999) Regulation of mammalian O2 homeostasis by hypoxia-inducible factor 1. Annu Rev Cell Dev Biol **15**, 551-578.

Semenza, G.L. (2000_a) Expression of hypoxia-inducible factor 1: mechanisms and consequences. Biochem Pharmacol **59**, 47-53.

Semenza, G.L. (2000_b) HIF-1: mediator of physiological and pathophysiological responses to hypoxia. J Appl Physiol **88**, 1474-1480.

Semenza, G.L. (2001_a) HIF-1 and mechanisms of hypoxia sensing. Curr Opin Cell Biol **13**, 167-171.

Semenza, G.L. (2001_b) Hypoxia-inducible factor 1: oxygen homeostasis and disease pathophysiology. Trends Mol Med **7**, 345-350.

Semenza, G.L. (2003) Targeting HIF-1 for cancer therapy. Nat Rev Cancer 3, 721-732.

Semenza, G.L. (2004) Hydroxylation of HIF-1: oxygen sensing at the molecular level. Physiology (Bethesda) **19**, 176-182.

Semenza, G.L. (2009) Regulation of oxygen homeostasis by hypoxia-inducible factor 1. Physiology (Bethesda) **24**, 97-106.

Semenza, G.L., Jiang, B.H., Leung, S.W., Passantino, R., Concordet, J.P., Maire, P., and Giallongo, A. (1996) Hypoxia response elements in the aldolase A, enolase 1, and lactate dehydrogenase A gene promoters contain essential binding sites for hypoxia-inducible factor 1. J Biol Chem **271**, 32529-32537.

Semenza, G.L., Roth, P.H., Fang, H.M., and Wang, G.L. (1994) Transcriptional regulation of genes encoding glycolytic enzymes by hypoxia-inducible factor 1. J Biol Chem **269**, 23757-23763.

Serbina, N.V., Jia, T., Hohl, T.M., and Pamer, E.G. (2008) Monocyte-mediated defense against microbial pathogens. Annu Rev Immunol **26**, 421-452.

Sermeus, A., Cosse, J.P., Crespin, M., Mainfroid, V., de Longueville, F., Ninane, N., Raes, M., Remacle, J., and Michiels, C. (2008) Hypoxia induces protection against etoposideinduced apoptosis: molecular profiling of changes in gene expression and transcription factor activity. Mol Cancer **7**, 27.

Serra, M., Miquel, L., Domenzain, C., Docampo, M.J., Fabra, A., Wight, T.N., and Bassols, A. (2005) V3 versican isoform expression alters the phenotype of melanoma cells and their

tumorigenic potential. Int J Cancer **114**, 879-886.

Sheng, W., Wang, G., Wang, Y., Liang, J., Wen, J., Zheng, P.S., Wu, Y., Lee, V., Slingerland, J., Dumont, D., and Yang, B.B. (2005) The roles of versican V1 and V2 isoforms in cell proliferation and apoptosis. Mol Biol Cell **16**, 1330-1340.

Shi, Q., Le, X., Abbruzzese, J.L., Wang, B., Mujaida, N., Matsushima, K., Huang, S., Xiong, Q., and Xie, K. (1999) Cooperation between transcription factor AP-1 and NF-kappaB in the induction of interleukin-8 in human pancreatic adenocarcinoma cells by hypoxia. J Interferon Cytokine Res **19**, 1363-1371.

Simanonok, J.P. (1996) Non-ischemic hypoxia of the arterial wall is a primary cause of atherosclerosis. Med Hypotheses **46**, 155-161.

Simon, M.C., and Keith, B. (2008) The role of oxygen availability in embryonic development and stem cell function. Nat Rev Mol Cell Biol **9**, 285-296.

Smith, P.D., Ochsenbauer-Jambor, C., and Smythies, L.E. (2005) Intestinal macrophages: unique effector cells of the innate immune system. Immunol Rev **206**, 149-159.

Smyth, C.M., Logan, G., Boadle, R., Rowe, P.B., Smythe, J.A., and Alexander, I.E. (2005) Differential subcellular localization of CD86 in human PBMC-derived macrophages and DCs, and ultrastructural characterization by immuno-electron microscopy. Int Immunol **17**, 123-132.

Smythies, L.E., Sellers, M., Clements, R.H., Mosteller-Barnum, M., Meng, G., Benjamin, W.H., Orenstein, J.M., and Smith, P.D. (2005) Human intestinal macrophages display profound inflammatory anergy despite avid phagocytic and bacteriocidal activity. J Clin Invest **115**, 66-75.

Sobell, H.M. (1985) Actinomycin and DNA transcription. Proc Natl Acad Sci U S A 82, 5328-5331.

Staples, K.J., Bergmann, M., Barnes, P.J., and Newton, R. (2000) Stimulus-specific inhibition of IL-5 by cAMP-elevating agents and IL-10 reveals differential mechanisms of action. Biochem Biophys Res Commun **273**, 811-815.

Staples, K.J., Smallie, T., Williams, L.M., Foey, A., Burke, B., Foxwell, B.M., and Ziegler-Heitbrock, L. (2007) IL-10 induces IL-10 in primary human monocyte-derived macrophages via the transcription factor Stat3. J Immunol **178**, 4779-4785.

Staples, K.J., Sotoodehnejadnematalahi, F., Pearson, H., Frankenberger, M., Francescut, L., Ziegler-Heitbrock, L., and Burke, B. (2011) Monocyte-derived macrophages matured under prolonged hypoxia transcriptionally up-regulate HIF-1alpha mRNA. Immunobiology **216**, 832-839

Stein, R.C. (2001) Prospects for phosphoinositide 3-kinase inhibition as a cancer treatment. Endocr Relat Cancer **8**, 237-248.

Stevens, C.R., Williams, R.B., Farrell, A.J., and Blake, D.R. (1991) Hypoxia and inflammatory synovitis: observations and speculation. Ann Rheum Dis **50**, 124-132.

Stock, W., and Hoffman, R. (2000) White blood cells 1: non-malignant disorders. Lancet **355**, 1351-1357.

Streuli, C. (1999) Extracellular matrix remodelling and cellular differentiation. Curr Opin Cell Biol **11**, 634-640.

Stuart, L.M., and Ezekowitz, R.A. (2005) Phagocytosis: elegant complexity. Immunity **22**, 539-550

Subarsky, P., and Hill, R.P. (2003) The hypoxic tumour microenvironment and metastatic progression. Clin Exp Metastasis **20**, 237-250.

Sugiura, N., Sakurai, K., Hori, Y., Karasawa, K., Suzuki, S., and Kimata, K. (1993) Preparation of lipid-derivatized glycosaminoglycans to probe a regulatory function of the carbohydrate moieties of proteoglycans in cell-matrix interaction. J Biol Chem **268**, 15779-15787.

Sutherland, R.M. (1998) Tumor hypoxia and gene expression--implications for malignant progression and therapy. Acta Oncol **37**, 567-574.

Sutter, C.H., Laughner, E., and Semenza, G.L. (2000) Hypoxia-inducible factor 1alpha protein expression is controlled by oxygen-regulated ubiquitination that is disrupted by deletions and missense mutations. Proc Natl Acad Sci U S A **97**, 4748-4753.

Suwiwat, S., Ricciardelli, C., Tammi, R., Tammi, M., Auvinen, P., Kosma, V.M., LeBaron, R.G., Raymond, W.A., Tilley, W.D., and Horsfall, D.J. (2004) Expression of extracellular matrix components versican, chondroitin sulfate, tenascin, and hyaluronan, and their association with disease outcome in node-negative breast cancer. Clin Cancer Res **10**, 2491-2498.

Syrokou, A., Tzanakakis, G.N., Hjerpe, A., and Karamanos, N.K. (1999) Proteoglycans in human malignant mesothelioma. Stimulation of their synthesis induced by epidermal, insulin and platelet-derived growth factors involves receptors with tyrosine kinase activity. Biochimie **81**, 733-744.

Szalad, A., Katakowski, M., Zheng, X., Jiang, F., and Chopp, M. (2009) Transcription factor Sp1 induces ADAM17 and contributes to tumor cell invasiveness under hypoxia. J Exp Clin Cancer Res **28**, 129.

Takahashi, K., Takeya, M., and Sakashita, N. (2002) Multifunctional roles of macrophages in the development and progression of atherosclerosis in humans and experimental animals. Med Electron Microsc **35**, 179-203.

Takeda, K., and Akira, S. (2005) Toll-like receptors in innate immunity. Int Immunol **17**, 1-14.

Talks, K.L., Turley, H., Gatter, K.C., Maxwell, P.H., Pugh, C.W., Ratcliffe, P.J., and Harris, A.L. (2000) The expression and distribution of the hypoxia-inducible factors HIF-1alpha and HIF-2alpha in normal human tissues, cancers, and tumor-associated macrophages. Am J Pathol **157**, 411-421.

Talusan, P., Bedri, S., Yang, S., Kattapuram, T., Silva, N., Roughley, P.J., and Stone, J.R. (2005) Analysis of intimal proteoglycans in atherosclerosis-prone and atherosclerosis-resistant human arteries by mass spectrometry. Mol Cell Proteomics **4**, 1350-1357.

Tang, X., Davies, J.E., and Davies, S.J. (2003) Changes in distribution, cell associations, and protein expression levels of NG2, neurocan, phosphacan, brevican, versican V2, and tenascin-C during acute to chronic maturation of spinal cord scar tissue. J Neurosci Res **71**, 427-444.

Tanimoto, K., Makino, Y., Pereira, T., and Poellinger, L. (2000) Mechanism of regulation of the hypoxia-inducible factor-1 alpha by the von Hippel-Lindau tumor suppressor protein. EMBO J **19**, 4298-4309.

Taylor, P.R., Martinez-Pomares, L., Stacey, M., Lin, H.H., Brown, G.D., and Gordon, S. (2005) Macrophage receptors and immune recognition. Annu Rev Immunol **23**, 901-944.

Theocharis, A.D. (2002) Human colon adenocarcinoma is associated with specific posttranslational modifications of versican and decorin. Biochim Biophys Acta **1588**, 165-172.

Theocharis, A.D., Tsara, M.E., Papageorgacopoulou, N., Karavias, D.D., and Theocharis, D.A. (2000) Pancreatic carcinoma is characterized by elevated content of hyaluronan and chondroitin sulfate with altered disaccharide composition. Biochim Biophys Acta **1502**, 201-206.

Tinton, S.A., and Buc-Calderon, P.M. (1999) Hypoxia increases the association of 4Ebinding protein 1 with the initiation factor 4E in isolated rat hepatocytes. FEBS Lett **446**, 55-59.

Touab, M., Villena, J., Barranco, C., Arumi-Uria, M., and Bassols, A. (2002) Versican is differentially expressed in human melanoma and may play a role in tumor development. Am J Pathol **160**, 549-557.

Treins, C., Giorgetti-Peraldi, S., Murdaca, J., Semenza, G.L., and Van Obberghen, E. (2002) Insulin stimulates hypoxia-inducible factor 1 through a phosphatidylinositol 3-kinase/target of rapamycin-dependent signaling pathway. J Biol Chem **277**, 27975-27981.

Triantafilou, M., and Triantafilou, K. (2002) Lipopolysaccharide recognition: CD14, TLRs and the LPS-activation cluster. Trends Immunol **23**, 301-304.

Tripathi, A., and Sodhi, A. (2008) Prolactin-induced production of cytokines in macrophages in vitro involves JAK/STAT and JNK MAPK pathways. Int Immunol **20**, 327-336.

Uchimura, E., Kodaira, T., Kurosaka, K., Yang, D., Watanabe, N., and Kobayashi, Y. (1997) Interaction of phagocytes with apoptotic cells leads to production of proinflammatory cytokines. Biochem Biophys Res Commun **239**, 799-803. Uenoyama, Y., Seno, H., Fukuda, A., Sekikawa, A., Nanakin, A., Sawabu, T., Kawada, M., Kanda, N., Suzuki, K., Yada, N., Fukui, H., and Chiba, T. (2006) Hypoxia induced by benign intestinal epithelial cells is associated with cyclooxygenase-2 expression in stromal cells through AP-1-dependent pathway. Oncogene **25**, 3277-3285.

Unanue, E.R. (1984) Antigen-presenting function of the macrophage. Annu Rev Immunol **2**, 395-428.

Underhill, D.M., Ozinsky, A., Hajjar, A.M., Stevens, A., Wilson, C.B., Bassetti, M., and Aderem, A. (1999) The Toll-like receptor 2 is recruited to macrophage phagosomes and discriminates between pathogens. Nature **401**, 811-815.

van Furth, R., and Cohn, Z.A. (1968) The origin and kinetics of mononuclear phagocytes. J Exp Med **128**, 415-435.

Van Ginderachter, J.A., Movahedi, K., Hassanzadeh Ghassabeh, G., Meerschaut, S., Beschin, A., Raes, G., and De Baetselier, P. (2006) Classical and alternative activation of mononuclear phagocytes: picking the best of both worlds for tumor promotion. Immunobiology **211**, 487-501.

van Horssen, J., Dijkstra, C.D., and de Vries, H.E. (2007) The extracellular matrix in multiple sclerosis pathology. J Neurochem **103**, 1293-1301.

van Rooijen, N., and Sanders, A. (1997) Elimination, blocking, and activation of macrophages: three of a kind? J Leukoc Biol **62**, 702-709.

Vaupel, P., Kallinowski, F., and Okunieff, P. (1989) Blood flow, oxygen and nutrient supply, and metabolic microenvironment of human tumors: a review. Cancer Res **49**, 6449-6465.

Vaupel, P., and Mayer, A. (2007) Hypoxia in cancer: significance and impact on clinical outcome. Cancer Metastasis Rev **26**, 225-239.

Vaupel, P., Mayer, A., and Hockel, M. (2004) Tumor hypoxia and malignant progression. Methods Enzymol **381**, 335-354.

Vaupel, P., Thews, O., and Hoeckel, M. (2001) Treatment resistance of solid tumors: role of hypoxia and anemia. Med Oncol **18**, 243-259.

Vengellur, A., Phillips, J.M., Hogenesch, J.B., and LaPres, J.J. (2005) Gene expression profiling of hypoxia signaling in human hepatocellular carcinoma cells. Physiol Genomics **22**, 308-318.

Voutilainen, K., Anttila, M., Sillanpaa, S., Tammi, R., Tammi, M., Saarikoski, S., and Kosma, V.M. (2003) Versican in epithelial ovarian cancer: relation to hyaluronan, clinicopathologic factors and prognosis. Int J Cancer **107**, 359-364.

Wang, G.L., and Semenza, G.L. (1993_a) General involvement of hypoxia-inducible factor 1 in transcriptional response to hypoxia. Proc Natl Acad Sci U S A **90**, 4304-4308.

Wang, G.L., and Semenza, G.L. (1993_b) Desferrioxamine induces erythropoietin gene expression and hypoxia-inducible factor 1 DNA-binding activity: implications for models of hypoxia signal transduction. Blood **82**, 3610-3615.

Wang, G.L., and Semenza, G.L. (1995) Purification and characterization of hypoxiainducible factor 1. J Biol Chem **270**, 1230-1237.

Wang, J., Dodd, C., Shankowsky, H.A., Scott, P.G., and Tredget, E.E. (2008) Deep dermal fibroblasts contribute to hypertrophic scarring. Lab Invest **88**, 1278-1290.

Watson, J.A., Watson, C.J., McCrohan, A.M., Woodfine, K., Tosetto, M., McDaid, J., Gallagher, E., Betts, D., Baugh, J., O'Sullivan, J., Murrell, A., Watson, R.W., and McCann, A. (2009) Generation of an epigenetic signature by chronic hypoxia in prostate cells. Hum Mol Genet **18**, 3594-3604.

Weinmann, M., Belka, C., Guner, D., Goecke, B., Muller, I., Bamberg, M., and Jendrossek, V. (2005) Array-based comparative gene expression analysis of tumor cells with increased apoptosis resistance after hypoxic selection. Oncogene **24**, 5914-5922.

Wenger, R.H. (2002) Cellular adaptation to hypoxia: O2-sensing protein hydroxylases, hypoxia-inducible transcription factors, and O2-regulated gene expression. FASEB J **16**, 1151-1162.

Wenger, R.H., Stiehl, D.P., and Camenisch, G. (2005) Integration of oxygen signaling at the consensus HRE. Sci STKE **2005**, re12.

Werb, Z. (1997) ECM and cell surface proteolysis: regulating cellular ecology. Cell **91**, 439-442.

Westergren-Thorsson, G., Norman, M., Bjornsson, S., Endresen, U., Stjernholm, Y., Ekman, G., and Malmstrom, A. (1998) Differential expressions of mRNA for proteoglycans, collagens and transforming growth factor-beta in the human cervix during pregnancy and involution. Biochim Biophys Acta **1406**, 203-213.

White, J.R., Harris, R.A., Lee, S.R., Craigon, M.H., Binley, K., Price, T., Beard, G.L., Mundy, C.R., and Naylor, S. (2004) Genetic amplification of the transcriptional response to hypoxia as a novel means of identifying regulators of angiogenesis. Genomics **83**, 1-8.

Wight, T.N. (2002) Versican: a versatile extracellular matrix proteoglycan in cell biology. Curr Opin Cell Biol **14**, 617-623.

Wight, T.N., and Merrilees, M.J. (2004) Proteoglycans in atherosclerosis and restenosis: key roles for versican. Circ Res **94**, 1158-1167.

Wijelath, E.S., Murray, J., Rahman, S., Patel, Y., Ishida, A., Strand, K., Aziz, S., Cardona, C., Hammond, W.P., Savidge, G.F., Rafii, S., and Sobel, M. (2002) Novel vascular endothelial growth factor binding domains of fibronectin enhance vascular endothelial growth factor biological activity. Circ Res **91**, 25-31.

Woo, K.J., Lee, T.J., Park, J.W., and Kwon, T.K. (2006) Desferrioxamine, an iron chelator, enhances HIF-1alpha accumulation via cyclooxygenase-2 signaling pathway. Biochem Biophys Res Commun **343**, 8-14.

Wu, Y., Chen, L., Cao, L., Sheng, W., and Yang, B.B. (2004) Overexpression of the C-terminal PG-M/versican domain impairs growth of tumor cells by intervening in the interaction between epidermal growth factor receptor and beta1-integrin. J Cell Sci **117**, 2227-2237.

Wu, Y., Chen, L., Zheng, P.S., and Yang, B.B. (2002) beta 1-Integrin-mediated glioma cell adhesion and free radical-induced apoptosis are regulated by binding to a C-terminal domain of PG-M/versican. J Biol Chem **277**, 12294-12301.

Wu, Y., Zhang, Y., Cao, L., Chen, L., Lee, V., Zheng, P.S., Kiani, C., Adams, M.E., Ang, L.C., Paiwand, F., and Yang, B.B. (2001) Identification of the motif in versican G3 domain that plays a dominant-negative effect on astrocytoma cell proliferation through inhibiting versican secretion and binding. J Biol Chem **276**, 14178-14186.

Wu, Y.J., La Pierre, D.P., Wu, J., Yee, A.J., and Yang, B.B. (2005) The interaction of versican with its binding partners. Cell Res **15**, 483-494.

Xu, L., Pathak, P.S., and Fukumura, D. (2004) Hypoxia-induced activation of p38 mitogen-activated protein kinase and phosphatidylinositol 3'-kinase signaling pathways contributes to expression of interleukin 8 in human ovarian carcinoma cells. Clin Cancer Res **10**, 701-707.

Xu, Q., Ji, Y.S., and Schmedtje, J.F., Jr. (2000) Sp1 increases expression of cyclooxygenase-2 in hypoxic vascular endothelium. Implications for the mechanisms of aortic aneurysm and heart failure. J Biol Chem **275**, 24583-24589.

Yamada, K.M., and Even-Ram, S. (2002) Integrin regulation of growth factor receptors. Nat Cell Biol **4**, E75-76.

Yamagata, M., and Kimata, K. (1994) Repression of a malignant cell-substratum adhesion phenotype by inhibiting the production of the anti-adhesive proteoglycan, PG-M/versican. J Cell Sci **107** (Pt 9), 2581-2590.

Yamagata, M., Saga, S., Kato, M., Bernfield, M., and Kimata, K. (1993) Selective distributions of proteoglycans and their ligands in pericellular matrix of cultured fibroblasts. Implications for their roles in cell-substratum adhesion. J Cell Sci **106** (Pt 1), 55-65.

Yamagata, M., Suzuki, S., Akiyama, S.K., Yamada, K.M., and Kimata, K. (1989) Regulation of cell-substrate adhesion by proteoglycans immobilized on extracellular substrates. J Biol Chem **264**, 8012-8018.

Yamagata, M., Yamada, K.M., Yoneda, M., Suzuki, S., and Kimata, K. (1986) Chondroitin sulfate proteoglycan (PG-M-like proteoglycan) is involved in the binding of hyaluronic acid to cellular fibronectin. J Biol Chem **261**, 13526-13535.

Yamashita, K., Discher, D.J., Hu, J., Bishopric, N.H., and Webster, K.A. (2001) Molecular regulation of the endothelin-1 gene by hypoxia. Contributions of hypoxia-inducible factor-1, activator protein-1, GATA-2, AND p300/CBP. J Biol Chem **276**, 12645-12653.

Yang, B.L., Cao, L., Kiani, C., Lee, V., Zhang, Y., Adams, M.E., and Yang, B.B. (2000) Tandem repeats are involved in G1 domain inhibition of versican expression and secretion and the G3 domain enhances glycosaminoglycan modification and product secretion via the complement-binding protein-like motif. J Biol Chem **275**, 21255-21261.

Yee, A.J., Akens, M., Yang, B.L., Finkelstein, J., Zheng, P.S., Deng, Z., and Yang, B. (2007) The effect of versican G3 domain on local breast cancer invasiveness and bony metastasis. Breast Cancer Res **9**, R47.

Yoon, H., Liyanarachchi, S., Wright, F.A., Davuluri, R., Lockman, J.C., de la Chapelle, A., and Pellegata, N.S. (2002) Gene expression profiling of isogenic cells with different TP53 gene dosage reveals numerous genes that are affected by TP53 dosage and identifies CSPG2 as a direct target of p53. Proc Natl Acad Sci U S A **99**, 15632-15637.

Yu, Q., Toole, B.P., and Stamenkovic, I. (1997) Induction of apoptosis of metastatic mammary carcinoma cells in vivo by disruption of tumor cell surface CD44 function. J Exp Med **186**, 1985-1996.

Yuan, J., and Glazer, P.M. (1998) Mutagenesis induced by the tumor microenvironment. Mutat Res **400**, 439-446.

Yuan, Y., Hilliard, G., Ferguson, T., and Millhorn, D.E. (2003) Cobalt inhibits the interaction between hypoxia-inducible factor-alpha and von Hippel-Lindau protein by direct binding to hypoxia-inducible factor-alpha. J Biol Chem **278**, 15911-15916.

Zako, M., Shinomura, T., Ujita, M., Ito, K., and Kimata, K. (1995) Expression of PG-M(V3), an alternatively spliced form of PG-M without a chondroitin sulfate attachment in region in mouse and human tissues. J Biol Chem **270**, 3914-3918.

Zhang, X., Gaspard, J.P., and Chung, D.C. (2001) Regulation of vascular endothelial growth factor by the Wnt and K-ras pathways in colonic neoplasia. Cancer Res **61**, 6050-6054.

Zhang, Y., Cao, L., Kiani, C., Yang, B.L., Hu, W., and Yang, B.B. (1999) Promotion of chondrocyte proliferation by versican mediated by G1 domain and EGF-like motifs. J Cell Biochem **73**, 445-457.

Zhang, Y., Cao, L., Yang, B.L., and Yang, B.B. (1998) The G3 domain of versican enhances cell proliferation via epidermial growth factor-like motifs. J Biol Chem **273**, 21342-21351.

Zheng, P.S., Reis, M., Sparling, C., Lee, D.Y., La Pierre, D.P., Wong, C.K., Deng, Z., Kahai, S., Wen, J., and Yang, B.B. (2006) Versican G3 domain promotes blood coagulation through

suppressing the activity of tissue factor pathway inhibitor-1. J Biol Chem **281**, 8175-8182.

Zheng, P.S., Wen, J., Ang, L.C., Sheng, W., Viloria-Petit, A., Wang, Y., Wu, Y., Kerbel, R.S., and Yang, B.B. (2004) Versican/PG-M G3 domain promotes tumor growth and angiogenesis. FASEB J **18**, 754-756.

Zhong, H., Chiles, K., Feldser, D., Laughner, E., Hanrahan, C., Georgescu, M.M., Simons, J.W., and Semenza, G.L. (2000) Modulation of hypoxia-inducible factor 1alpha expression by the epidermal growth factor/phosphatidylinositol 3-kinase/PTEN/AKT/FRAP pathway in human prostate cancer cells: implications for tumor angiogenesis and therapeutics. Cancer Res **60**, 1541-1545.

Zimmermann, D.R., Dours-Zimmermann, M.T., Schubert, M., and Bruckner-Tuderman, L. (1994) Versican is expressed in the proliferating zone in the epidermis and in association with the elastic network of the dermis. J Cell Biol **124**, 817-825.

Zimmermann, D.R., and Ruoslahti, E. (1989) Multiple domains of the large fibroblast proteoglycan, versican. EMBO J **8**, 2975-2981.

Zvelebil, M.J., MacDougall, L., Leevers, S., Volinia, S., Vanhaesebroeck, B., Gout, I., Panayotou, G., Domin, J., Stein, R., Pages, F., and et al. (1996) Structural and functional diversity of phosphoinositide 3-kinases. Philos Trans R Soc Lond B Biol Sci **351**, 217-223.
Appendix

Publication

 Staples, K.J., Sotoodehnejadnematalahi, F., Pearson, H., Frankenberger, M., Francescut, L., Ziegler-Heitbrock, L., and Burke, B. (2011) Monocyte-derived macrophages matured under prolonged hypoxia transcriptionally up-regulate HIF-1alpha mRNA. Immunobiology **216**, 832-839